

# OŚRODEK DS. IMO BIULETYN INFORMACYJNY



# Zestawienie postanowień IMO - I półrocze 2021 r.

Polski Rejestr Statków S.A., Ośrodek ds. IMO, tel.: +48 58 75 11 226, email: imo@prs.pl, www.prs.pl

#### ZESTAWIENIE WYMAGAŃ IMO W I PÓŁROCZU 2021 R.

#### REZOLUCJE 103 SESJI KOMITETU BEZPIECZEŃSTWA NA MORZU (MSC 103) (5-14.05.2021 r.)

### Rezolucja MSC.482(103) Poprawki do Międzynarodowej konwencji o bezpieczeństwie życia na morzu SOLAS 1974 wraz z późniejszymi zmianami

Rezolucja wprowadza następujące zmiany:

- rozdział II-1 w cz. B-4 (Zarządzanie statecznością) wprowadzenie nowego prawidła 25-1 dotyczącego detektorów poziomu wody na statkach towarowych z wieloma ładowniami, innych niż masowce;
- rozdział III, prawidło 33 cz. B (Urządzenia do wsiadania do jednostek ratunkowych i do ich wodowania) zastąpienie pkt. 33.2 nowym wymaganiem, zgodnie z którym na statkach o pojemności brutto 20 000 lub większej powinno być możliwe wodowanie łodzi ratunkowych, wodowanych za pomocą żurawików, z wykorzystaniem falenia, jeśli jest to konieczne, ze statku przy ruchu naprzód z prędkością 5 węzłów i na spokojnej wodzie.

Przedmiotowe zmiany wejdą w życie <u>1 stycznia 2024 roku.</u>

### Rezolucja MSC.483(103) Poprawki do Międzynarodowego kodeksu wdrażania programu rozszerzonych przeglądów na masowcach i zbiornikowcach olejowych, 2011 (Kodeks ESP, 2011)

Rezolucja wprowadza zmiany do kolumny *Renewal Survey No.1* w tabeli minimalnych wymagań dla pomiarów grubości podczas przeglądów odnowieniowych zbiornikowców olejowych o podwójnym kadłubie w Załączniku 2, części A Kodeksu.

Przedmiotowe zmiany wejdą w życie <u>1 stycznia 2023 r</u>.

### Rezolucja MSC.484(103) Poprawki do Międzynarodowego kodeksu systemów bezpieczeństwa pożarowego (Kodeks FSS)

Rezolucja wprowadza zmiany do rozdziału 9 Kodeksu (Stałe systemy wykrywania i sygnalizacji pożaru), polegające na dodaniu nowego punktu 2.1.8., o braku wymogu instalowania modułów izolujących przy każdej czujce przeciwpożarowej, przy zachowaniu konkretnych warunków, na statkach towarowych i na balkonach kabin statków pasażerskich.

Przedmiotowe zmiany wejdą w życie 1 stycznia 2024 roku.

### Rezolucja MSC.485(103) Poprawki do Międzynarodowego kodeksu środków ratunkowych (Kodeks LSA)

Rezolucja wprowadza zmiany do pkt. 4.4.1.3.2 rozdziału 4 (Jednostki ratunkowe), polegające na wyłączeniu łodzi ratunkowych swobodnego spadku z wymagania, aby mogły być wodowane i holowane podczas ruchu statku naprzód przy prędkości do 5 węzłów na spokojnej wodzie.

Przedmiotowe zmiany wejdą w życie 1 stycznia 2024 roku.

### Rezolucja MSC.486(103) Poprawki do Międzynarodowej konwencji o wymaganiach w zakresie wyszkolenia marynarzy, wydawania świadectw oraz pełnienia wacht, 1978 (STCW 1978)

Rezolucja wprowadza zmiany do prawidła I/1.1, polegające na dodaniu nowej definicji wysokiego napięcia.

Przedmiotowe zmiany wejdą w życie <u>1 stycznia 2023 r</u>.

### Rezolucja MSC.487(103) Poprawki do Kodeksu wyszkolenia marynarzy, wydawania świadectw oraz pełnienia wacht (Kodeks STCW)

Rezolucja wprowadza zmiany do część A-I/1 Kodeksu, polegające na zmianie definicji poziomu operacyjnego. Przedmiotowe zmiany wejdą w życie <u>1 stycznia 2023 r</u>.

### Rezolucja MSC.488(103) Poprawki do Znowelizowanych zaleceń w zakresie przeprowadzania prób środków ratunkowych (rez. MSC.81(70))

Rezolucja wprowadza zmiany, które wynikają z projektu poprawek do rozdz. III SOLAS oraz Kodeksu LSA, z perspektywą przyjęcia wraz ze zmianami do prawidła III/33 oraz rozdz. IV Kodeksu LSA.

Dotyczą one:

- Dodatkowych testów materiałowych wyłącznie nadmuchiwanych tratw ratunkowych,
- Testów technicznych zwalniaków hydrostatycznych,
- Wyłączenia łodzi ratunkowych swobodnego spadku z prób wodowania w pełni wyposażonych łodzi ratunkowych oraz łodzi ratowniczych, na statkach towarowych o pojemności brutto 20 000 lub większej, przy statku poruszającym się naprzód z prędkością nie mniejszą niż 5 węzłów, na spokojnej wodzie i w stanie równowagi.

Przedmiotowe zmiany wejdą w życie <u>1 stycznia 2024 r</u>.

### Rezolucja MSC.489(103) Działania rekomendowane w zakresie przeciwdziałania piractwu i zbrojnym napadom na statki w Zatoce Gwinejskiej

Rezolucja przypomina o wysiłkach IMO w zakresie przeciwdziałania piractwu i zbrojnym napadom na statki oraz w zakresie promowania bezpiecznej żeglugi na wodach Zatoki Gwinejskiej. Jednocześnie zwraca się do państw członkowskich, podmiotów krajowych oraz Organizacji Narodów Zjednoczonych o podjęcie działań związanych ze wzmocnieniem egzekwowania prawa i ujednoliceniem kar w państwach nadbrzeżnych; lepszym zarządzaniem dozwolonymi rozwiązaniami związanymi z ochroną; oraz propagowaniem uczestnictwa w projektach realizowanych w celu poprawy bezpieczeństwa w tym regionie.

Rezolucja prosi również o rozważenie finansowej kontrybucji na rzecz Funduszu Powierniczego Ochrony Żeglugi IMO WCA, który wspiera programy adresowane do państw nadbrzeżnych w rejonie Zatoki Gwinejskiej oraz ośrodki regionalne usytuowane w państwach, które przyjęły Porozumienie z Jaunde.

### Rezolucja MSC.490(103) Zalecane działania mające na celu nadanie priorytetu szczepieniom marynarzy przeciwko COVID-19

Rezolucja zwraca się do państw członkowskich oraz właściwych władz krajowych o podjęcie następujących działań:

- 1. priorytetowego traktowania marynarzy jako pracowników o kluczowym znaczeniu, w ramach krajowych programów szczepień zgodnie z zaleceniami Mapy drogowej WHO SAGE 4;
- 2. wyłączenie tej grupy zawodowej z konieczności przedstawienia dowodu szczepień przy przekraczaniu granic;
- 3. opracowanie odpowiednich planów i infrastruktury wspomagającej szczepienie marynarzy.

Rezolucja wzywa ponadto państwa członkowskie, organizacje międzynarodowe, związki armatorskie i inne podmioty do prowadzenia akcji informacyjnej nt. bezpieczeństwa i korzyści wynikających z zaszczepienia przeciwko COVID-19.

#### OKÓLNIKI KOMITETU BEZPIECZEŃSTWA NA MORZU (MSC)

### Okólnik MSC.1/Circ.797/Rev.35 Lista osób nominowanych przez Państwa członkowskie jako osoby kompetentne zgodnie z sekcją A-1/7 Kodeksu STCW

Okólnik zawiera aktualną listę nominowanych osób kompetentnych, prowadzoną przez Sekretarza Generalnego zgodnie z wymaganiami sekcji A-I/7 Kodeksu STCW.

#### Okólnik MSC.1/Circ.1164/Rev.23 Rozpowszechnienie informacji o państwach stronach Konwencji STCW, 1978 wraz z późn. zm., które wdrożyły jej postanowienia i złożyły odpowiednie raporty oraz sprawozdania z niezależnej oceny zgodnie z wymaganiami Konwencji

Okólnik zawiera kolejną uaktualnioną listę państw stron Konwencji STCW, które wypełniły ww. obowiązki konwencyjne wg stanu do 103 sesji Komitetu MSC.

### Okólnik MSC.1/Circ.1318/Rev.1 Zmienione wytyczne dotyczące konserwacji i przeglądów stałych instalacji gaśniczych na dwutlenek węgla

Niniejszy okólnik zawiera Zrewidowane wytyczne dotyczące konserwacji i przeglądów stałych instalacji gaśniczych na dwutlenek węgla, które dostosowują je do Zmienionych wytycznych dotyczących konserwacji i przeglądów systemów i urządzeń przeciwpożarowych (MSC.1/Circ.1432) w zakresie badania wszystkich zaworów w butlach na CO<sub>2</sub> podczas ich serwisowania co pięć lat oraz wprowadzenia stosownych poprawek do przykładowych kart serwisowych.

Niniejszy okólnik zastępuje dokument MSC.1/Circ.1318.

### Okólnik MSC.1/Circ.1601/Rev.1 Znowelizowany poradnik przemysłu w zakresie zapobiegania i powstrzymywania piractwa

Okólnik zawiera Znowelizowany poradnik przemysłu w zakresie zapobiegania i powstrzymywania piractwa, w którym zastąpiono załącznik 3 okólnika Poradnikiem BMP West Africa "Najlepsze praktyki powstrzymywania piractwa i zwiększenia bezpieczeństwa morskiego u wybrzeży Afryki Zachodniej, w tym Zatoki Gwinejskiej".

Niniejszy okólnik i wszystkie jego rewizje zastępują dokument MSC.1/Circ.1339.

### Okólnik MSC.1/Circ.1636/Rev.1 Zalecane przez przemysł ramy protokołów, które mają na celu zapewnienie bezpiecznej wymiany załóg i podróży podczas pandemii koronawirusa (COVID-19)

Okólnik zawiera poprawione Zalecane przez przemysł ramy protokołów, które mają na celu zapewnienie bezpiecznej wymiany załóg i podróży podczas pandemii koronawirusa (COVID-19). W rewizji uwzględniono kwestie związane z globalnym wprowadzeniem szczepionek oraz kluczową rolę, jaką one odgrywają w ułatwianiu bezpiecznej wymiany załóg i światowym handlu, o których mowa m.in. w liście okólnym nr 4204/Add.38 z dnia 25 marca 2021 r. oraz rezolucji MSC.473(ES.2) (Zalecane działania, mające na celu ułatwienie wymiany załóg, dostępu do opieki medycznej oraz podróżowania marynarzy w dobie pandemii COVID-19). Wraz z rozwojem sytuacji pandemicznej, przedmiotowe protokoły podlegają ciągłej rewizji międzynarodowych stowarzyszeń branżowych i jako takie będą na bieżąco publikowane jako kolejne rewizje okólnika MSC.1/Circ.1636.

### Okólnik MSC.1/Circ.1638 Wyniki przeglądu regulacji prawnych w zakresie korzystania z morskich nawodnych jednostek autonomicznych MASS

Okólnik zawiera wyniki przeglądu regulacji prawnych w zakresie korzystania z morskich nawodnych jednostek autonomicznych MASS.

Jego publikacja ma na celu uporządkowanie wyników dotychczasowej pracy, udostępnienie ich szerszemu kręgowi odbiorców oraz ułatwienie przyszłych prac nad opracowaniem ram prawnych dla jednostek MASS.

### Okólnik MSC.1/Circ.1639 czwarta rewizja Wytycznych dot. ochrony cybernetycznej na pokładach statków

Niniejszy okólnik informuje o aktualizacji opracowanych przez przemysł *Wytycznych dot. ochrony cybernetycznej na pokładach statków*. Wytyczne mają dobrowolny charakter i są dostępne pod adresem: <u>https://www.imo.org/en/OurWork/Security/Pages/Cyber-security.aspx</u>

Wytyczne są adresowane do właścicieli, zarządców i armatorów statków, którzy podejmują kwestie ryzyka cybernetycznego na statkach, zgodnie z celami I wymaganiami funkcjonalnymi Kodeksu ISM oraz rezolucji MSC.428(98) (*Zarządzanie Bezpieczeństwem Cybernetycznym dla Systemów Zarządzania Bezpieczeństwem*).

#### Okólnik MSC.1/Circ.1640 Poprawki do Poradnika IAMSAR

Niniejszy okólnik zawiera poprawki do Poradnika IAMSAR, które zastały sfinalizowane na 27 spotkaniu Wspólnej Grupy Roboczej ICAO/IMO ds. harmonizacji lotniczych i morskich procedur poszukiwania i ratownictwa, w październiku 2020 r.

Przedmiotowe poprawki będą miały zastosowanie od 1 czerwca 2022 roku.

### Okólnik MSC.1/Circ.1641 Wytyczne dla jednostek rybackich o długości 24m i większej, uprawiających żeglugę na wodach polarnych

Niniejsze *Wytyczne dla jednostek rybackich o długości 24m i większej, uprawiających żeglugę na wodach polarnych* zostały opracowane jako uzupełnienie istniejących instrumentów IMO w celu poprawy bezpieczeństwa tych statków oraz osób na ich pokładzie, jak również w celu złagodzenia wpływu środowiska na ludzi na odległych, podatnych i potencjalnie surowych obszarach wód polarnych. Wytyczne wpisują się w kontekst Porozumienia z Cape Town, 2012, które oczekuje na wejście w życie oraz Międzynarodowego kodeksu statków uprawiających żeglugę na wodach polarnych (Kodeks polarny).

Wytyczne mają charakter rekomendacji; sformułowania w nich użyte są wskazówkami, a nie obowiązkowymi instrukcjami, które naruszałyby krajowe systemy kontroli żeglugi.

### Okólnik MSC.1/Circ.1642 Wytyczne dla jachtów rekreacyjnych o pojemności 300 GT i większej, uprawiających żeglugę na wodach polarnych

Niniejsze Wytyczne dla jachtów rekreacyjnych o pojemności 300 GT i większej, uprawiających żeglugę na wodach polarnych zostały opracowane jako uzupełnienie istniejących norm przemysłu i standardów międzynarodowych. Wytyczne zawierają dodatkowe wskazówki mające na celu zwiększenie bezpieczeństwa jachtów i osób na ich pokładzie, jak również złagodzenie ryzyka, które jest związane z warunkami klimatycznymi i innymi czynnikami obecnymi na wodach polarnym.

Wytyczne mają charakter rekomendacji; sformułowania w nich użyte są wskazówkami, a nie obowiązkowymi instrukcjami, które naruszałyby krajowe systemy kontroli żeglugi.

#### Okólnik MSC.8/Circ.2 w/s Wcześniejszego dobrowolnego wdrożenia projektu poprawek do rozdziału III SOLAS oraz Kodeksu LSA

Okólnik dotyczy wcześniejszego dobrowolnego wdrożenia poprawek do prawidła III/33.2 SOLAS oraz pkt. 4.4.1.3.2 Kodeksu LSA dot. wymagań testowych łodzi ratunkowych swobodnego spadku, które zostały zawarte odpowiednio w rezolucjach MSC.482(103) oraz MSC.485(103). Data wejścia w życie przedmiotowych poprawek przypada na 1 stycznia 2024 r.

#### Okólnik MSC-FAL.1/Circ.3/Rev.1 Wytyczne ds. zarządzania morskim ryzykiem cybernetycznym

Wytyczne zawierają ogólne zalecenia na temat zarządzania morskim ryzykiem cybernetycznym, które mają na celu ochronę żeglugi przed aktualnymi i pojawiającymi się zagrożeniami cybernetycznymi oraz elementy funkcjonalne, które wspomagają efektywne zarządzanie ryzykiem cybernetycznym.

Wcześniejsze wytyczne zostały uzupełnione o Skonsolidowane rekomendacje w zakresie cyberbezpieczeństwa opracowane przez IACS (Rec. 166).

Niniejszy okólnik i wszystkie jego rewizje, zastępuje tymczasowe wytyczne zawarte w dokumencie MSC.1/Circ.1526.

#### REZOLUCJE 76 sesji KOMITETU Ochrony Środowiska Morskiego (MEPC 76) (10-17.06.2021)

### Rezolucja MEPC.328(76) Poprawki do Załącznika VI do Konwencji MARPOL (zmieniony Załącznik VI 2021)

Rezolucja zawiera znowelizowany i skonsolidowany Załącznik VI do Konwencji MARPOL. Na ten moment, brak ostatecznego tekstu tej rezolucji.

Rezolucja MEPC.329(76) Poprawki do Załącznika do Międzynarodowej Konwencji o zapobieganiu zanieczyszczaniu morza przez statki MARPOL 1973, zmienionej Protokołem z 1978

Poprawki do Załącznika I - Zakaz stosowania i przewozu jako paliwa, ciężkiego paliwa olejowego przez statki na wodach Arktyki.

Rezolucja wprowadza:

- nowe brzmienie tytułu rozdziału 9 tj. "Rozdział 9 Specjalne wymagania dotyczące stosowania lub przewozu olejów/paliw/paliw olejowych na wodach polarnych" (*Chapter 9 – Special requirements* for the use or carriage of oils in polar waters);
- 2. nowe prawidło 43A w rozdziale 9 pt. Specjalne wymagania dotyczące stosowania i przewozu olejów jako paliwa na wodach arktycznych (*Special requirements for the use and carriage of oils as fuel in Arctic waters*), informujące, iż co do zasady, od 1 lipca 2024 r. stosowanie i przewóz olejów wymienionych w prawidle 43.1.2 niniejszego załącznika, jako paliwa przez statki, jest zabronione na wodach arktycznych; natomiast dla statków do których ma zastosowanie prawidło 12A niniejszego załącznika lub prawidło 1.2.1 rozdziału 1 części II-A Kodeksu polarnego, datą wejścia wymagań 1 lipca 2029 r.

Poprawki wejdą w życie <u>1 listopada 2022 r.</u>

### Rezolucja MEPC.330(76) Poprawki do Załącznika do Międzynarodowej Konwencji o zapobieganiu zanieczyszczaniu morza przez statki MARPOL 1973, zmienionej Protokołem z 1978

Poprawki do Załączników I i IV - Zwolnienie bezzałogowych barek bez własnego napędu z niektórych wymagań dotyczących przeglądów i certyfikacji

#### W zakresie Załącznika I poprawki objęły:

- prawidło 1 (Definicje), w którym dodano nowy punkt 40 zawierający definicję bezzałogowej barki bez własnego napędu;
- prawidło 3 (Zwolnienia i odstępstwa), w którym zmieniono brzmienie paragrafu 2 oraz dodano nowy paragraf 7 informujący, iż zwolnienie następuje na podstawie Międzynarodowego świadectwa o wyłączeniu bezzałogowych barek bez własnego napędu (UNSP) z obowiązku zapobiegania zanieczyszczaniu olejami, na okres nie przekraczający 5 lat;
- prawidło 8 (Wydawanie lub potwierdzanie Świadectwa przez inny Rząd), w których zmieniono brzmienie paragrafu 4 dotyczącego nie wydawania ww. Świadectwa dla statku noszącego banderę państwa nie będącego stroną Konwencji MARPOL;
- prawidło 9 (Forma Świadectwa), w którym dodano nowy paragraf 2 zawierający informacje o Świadectwie;
- Uzupełnienie IV Formularz Międzynarodowego świadectwa o wyłączeniu bezzałogowych barek bez własnego napędu (UNSP) z obowiązku zapobiegania zanieczyszczaniu olejami.

#### W zakresie Załącznika IV poprawki objęły:

- prawidło 1 (Definicje), w którym dodano nowy paragraf 16 zawierający definicję bezzałogowej barki bez własnego napędu;
- prawidło 3 (Wyjątki), w którym zmieniono tytuł prawidła na "Wyłączenia i zwolnienia" (*Exceptions* and exemptions) oraz dodano nowy paragraf 2 informujący, iż zwolnienie następuje na podstawie

Międzynarodowego świadectwa o wyłączeniu bezzałogowych barek bez własnego napędu (UNSP) z obowiązku zapobiegania zanieczyszczaniu ściekami fekalnymi, na okres nie przekraczający 5 lat;

Nr 5/2021

- prawidło 6 (Wydawanie lub potwierdzanie Świadectwa przez inny Rząd) w którym zmieniono treść paragrafu 4 dotyczącego nie wydawania ww. Świadectwa dla statku noszącego banderę państwa nie będącego stroną Konwencji MARPOL;
- prawidło 7 (Forma Świadectwa) w którym umieszczono odniesienie do Uzupełnienia 1 oraz dodano nowy paragraf 2 zawierający odniesienie do nowego Uzupełnienia II zawierającego Formularz Międzynarodowego świadectwa o wyłączeniu bezzałogowych barek bez własnego napędu (UNSP) z obowiązku zapobiegania zanieczyszczaniu ściekami fekalnymi.

#### Poprawki wejdą w życie <u>1 listopada 2022 r.</u>

### Rezolucja MEPC.331(76) Poprawki do Międzynarodowej konwencji w sprawie kontroli nad systemami przeciwporostowymi stosowanymi na statkach, 2001

Kontrola cybutryny oraz formularz Międzynarodowego świadectwa systemu przeciwporostowego

Zaproponowane poprawki objęły:

- Załącznik I (Kontrola systemów przeciwporostowych) zmiany w tabeli
- Załącznik IV (Przeglądy i certyfikacja systemów przeciwporostowych) zmiana brzmienia prawidła 2(3)
- Uzupełnienie I do Załącznika IV (Formularz Międzynarodowego świadectwa systemu przeciwporostowego) zmiany w tabeli.

Poprawki wejdą w życie <u>1 stycznia 2023 r.</u>

# Rezolucja MEPC.332(76) Poprawki do Wytycznych 2018 dotyczących metody obliczania osiągniętego współczynnika projektowego efektywności energetycznej (EEDI) dla nowych statków (rezolucja MEPC.308(73) zmieniona rezolucją MEPC.322(74)).

Rezolucja wprowadza nową sekcję 3, dotyczącą obowiązkowego raportowania uzyskanych wartości EEDI i związanych z nimi informacji, oraz nowy Dodatek 5 zawierający standardowy formularz składania informacji o EEDI, w celu umieszczenia ich w bazie danych EEDI.

### Rezolucja MEPC.333(76) Wytyczne 2021 dotyczące metody obliczania osiągniętego wskaźnika efektywności energetycznej dla statku istniejącego (EEXI)

Wytyczne zawierają m.in. informacje na temat definicji, wzoru na obliczenie wskaźnika EEXI oraz takich parametrów koniecznych do jego obliczenia, jak: moc silników głównych ( $P_{ME(i)}$ ), moc silników pomocniczych ( $P_{AE(i)}$ ), prędkość statku ( $V_{ref}$ ), certyfikowane jednostkowe zużycie paliwa (SFC), współczynnik konwersji pomiędzy zużyciem paliwa, a emisją CO<sub>2</sub> ( $C_F$ ), współczynnik korygujący dla statków towarowych ro-ro i pasażerskich ro-ro ( $f_{jRORO}$ ), współczynnik korygujący pojemność sześcienną dla statków towarowych ro-ro (do przewozu pojazdów) ( $f_{cVEHICLE}$ ), oraz Dodatek zawierający parametry do obliczania przybliżonej wartości prędkości odniesienia  $V_{ref,app}$ .

### Rezolucja MEPC.334(76) Wytyczne 2021 dotyczące przeglądu i certyfikacji osiągniętego wskaźnika efektywności energetycznej dla statku istniejącego (EEXI)

Wytyczne zawierają m.in. informacje na temat definicji, zastosowania wytycznych (wytyczne powinny być stosowane w odniesieniu do statków, dla których został przedłożony weryfikatorowi wniosek o przeprowadzenie przeglądu w celu weryfikacji EEXI statku, określonego w prawidle 5 Załącznika VI do Konwencji MARPOL), oraz określają procedury dla przeglądu i certyfikacji (w tym weryfikację osiągniętego EEXI w przypadku znacznej przebudowy). W Dodatku do niniejszych wytycznych znajduje się przykładowa Kartoteka techniczna EEXI.

### Rezolucja MEPC.335(76) Wytyczne 2021 dotyczące układu ograniczenia mocy na wale/silnika w celu spełnienia wymagań EEXI i wykorzystania rezerwy mocy

Wytyczne zawierają m.in. informacje na temat definicji, wymagań technicznych dla systemu SHaPoLi/EPL (*Overridable Shaft Power Limitation*/ *Overridable Engine Power Limitation*), wykorzystania rezerwy mocy bez ograniczenia mocy na wale/silnika, Pokładowego Podręcznika Postępowania/Użytkowania (*Onboard Management Manual*) dla systemu SHaPoLi/EPL, oraz sposobu wykazania zgodności systemu SHaPoLi/EPL z prawidłem 5.4 (Przeglądy) Załącznika VI do Konwencji MARPOL.

### Rezolucja MEPC.336(76) Wytyczne 2021 dotyczące operacyjnych wskaźników intensywności emisji węgla oraz metod ich obliczania (*CII GUIDELINES*, *G*1)

Wytyczne zawierają m.in. informacje na temat definicji, zastosowania wytycznych (co do zasady dla wszystkich typów statków, do których ma zastosowanie prawidło 28 Załącznika VI do Konwencji MARPOL), podają wzór na obliczenie osiągniętego rocznego operacyjnego wskaźnika CII dla poszczególnych statków (stosunek całkowitej masy wyemitowanego CO<sub>2</sub> (M) do całkowitej pracy transportowej (W) wykonanej w danym roku kalendarzowym) oraz sposób obliczania CII dla poszczególnych statków, do celów próbnych.

### Rezolucja MEPC.337(76) Wytyczne 2021 dotyczące linii odniesienia do stosowania z eksploatacyjnymi wskaźnikami intensywności emisji węgla (*CII REFERENCE LINES GUIDELINES, G2*)

Zawarte w rezolucji wytyczne określają metody obliczania linii odniesienia do stosowania z eksploatacyjnymi wskaźnikami intensywności emisji węgla oraz linii odniesienia intensywności emisji węgla dla poszczególnych typów statków, o których mowa w prawidle 28 Załącznika VI do Konwencji MARPOL.

Wytyczne zawierają m.in. informacje na temat definicji oraz metody opracowania linii odniesienia dla CII.

# Rezolucja MEPC.338(76) Wytyczne 2021 dotyczące eksploatacyjnych współczynników redukcji intensywności emisji węgla w stosunku do linii odniesienia (*CII REDUCTION FACTORS GUIDELINES, G3*)

Wytyczne określają metody ustalania rocznych współczynników redukcji intensywności emisji węgla oraz ich konkretnych wartości w latach 2023-2030, o których mowa w prawidle 28 Załącznika VI do Konwencji MARPOL.

Wytyczne zawierają m.in. informacje na temat definicji, metod określania rocznego współczynnika redukcji dla różnych typów statków, w tym wzory na obliczenie eksploatacyjnej, osiągniętej intensywności emisji węgla z żeglugi międzynarodowej. Punkt 5 niniejszych wytycznych zawiera podstawowe informacje na temat zakresów współczynników redukcji dla różnych typów statków w roku 2030, przy zastosowaniu pomiaru opartego na popycie i na podaży.

### Rezolucja MEPC.339(76) Wytyczne 2021 dotyczące oceny eksploatacyjnej intensywności emisji węgla przez statki (CII RATING GUIDELINES, G4)

Wytyczne określają metody przypisywania statkom wskaźników eksploatacyjnej efektywności energetycznej, o których mowa w prawidle 28 Załącznika VI do Konwencji MARPOL. Na tej podstawie określono również granice wyznaczania rocznych eksploatacyjnych wskaźników intensywności emisji węgla dla statków w latach 2023-2030. Wytyczne zawierają m.in. informacje na temat definicji, ram dla oceny eksploatacyjnej efektywności energetycznej oraz metod określania granic tej oceny.

#### Okólniki Komitetu Ochrony Środowiska Morskiego (MEPC)

### MEPC.1/Circ.888/Rev.2 Raporty podsumowujące i analizy obowiązkowych sprawozdań zgodnie z Konwencją MARPOL (lata 2013-2018)

Okólnik zawiera informację dotyczącą wypełniania obowiązku raportowania do IMO w sprawach dotyczących ochrony środowiska, obejmującą lata 2013-2018, zgodnie z wymaganiami Konwencji MARPOL oraz formatem ustalonym w okólniku MEPC/Circ.318.

Strony są zobowiązane do składania rocznych sprawozdań, za pomocą wzorów raportów określonych w okólniku MEPC/Circ.318, do Organizacji do dnia 31 grudnia każdego roku. Okólnik ten, decyzją MEPC 67, wydawany jest corocznie.

### MEPC.1/Circ.891 Raporty podsumowujące i analizy obowiązkowych sprawozdań zgodnie z Konwencją MARPOL (lata 2014-2019)

Okólnik zawiera informację dotyczącą wypełniania obowiązku raportowania do IMO w sprawach dotyczących ochrony środowiska, obejmującą lata 2014-2019, zgodnie z wymaganiami Konwencji MARPOL oraz formatem ustalonym w okólniku MEPC/Circ.318.

Strony Konwencji są zobowiązane do składania do IMO do dnia 31 grudnia każdego roku rocznych sprawozdań, z wykorzystaniem wzorów raportów określonych w okólniku MEPC/Circ.318.

Okólnik ten, decyzją MEPC 67, wydawany jest corocznie.

### MEPC.1/Circ.892 Wytyczne dotyczące zwolnienia bezzałogowych barek bez własnego napędu (UNSP) z wymogów przeglądów i certyfikacji na zgodność z Konwencją MARPOL

Wytyczne mają zastosowanie do zwolnienia barek UNSP z niektórych wymagań określonych w prawidle 3.7 Załącznika I do Konwencji MARPOL, prawidle 3.2 Załącznika IV do Konwencji MARPOL oraz prawidle 3.4 Załącznika VI do Konwencji MARPOL.

Informują o wymaganiach technicznych i eksploatacyjnych, opisują procedury przyznawania i utrzymania zwolnień, oraz stwierdzają, iż podczas operacji holowania lub pchania, świadectwa zwolnienia dla każdej barki UNSP powinny być dostępne dla statku holującego lub pchającego. Niniejsze wytyczne powiązane są z rezolucjami MEPC.330(76) oraz MEPC.328(76).

### MEPC.1/Circ.893 Zapewnienie odpowiednich urządzeń w portach i terminalach do odbioru odpadów z tworzyw sztucznych ze statków

Okólnik przypomina o tym iż Komitet MEPC przyjął Plan działań IMO na rzecz przeciwdziałania zanieczyszczeniom morza odpadami z tworzyw sztucznych ze statków (rezolucja MEPC.310(73)) oraz przypomina Stronom Załącznika V do Konwencji MARPOL o ich obowiązku wynikającym z prawidła 8 tego Załącznika, polegającym na zapewnieniu odpowiednich urządzeń w portach i terminalach do odbioru śmieci, w tym odpadów z tworzyw sztucznych pochodzących ze statków i narzędzi połowowych, bez powodowania zbędnej zwłoki dla statków i zgodnie z ich potrzebami.

Okólnik przypomina również publikacje wydane przez IMO mając na celu ułatwienie skutecznego i ekologicznie odpowiedzialnego usuwania takich odpadów.

Okólnik przypomina ponadto Stronom Konwencji o ich obowiązku powiadamiania IMO (za pośrednictwem modułu PRF w systemie GISIS), o wszystkich przypadkach, w których urządzenia odbiorcze w portach, które powinny być zapewnione na mocy prawidła 8 Załącznika V do Konwencji MARPOL, były nieodpowiednie.

## MEPC.1/Circ.894 Dzielenie się informacjami na temat wyników badań nad odpadami morskimi i zachęcanie do ich kontynuowania w celu lepszego zrozumienia zagadnienia odpadów z mikroplastiku pochodzących ze statków

Okólnik przypomina o tym, iż Komitet MEPC przyjął Plan działań IMO na rzecz przeciwdziałania zanieczyszczeniom morza odpadami z tworzyw sztucznych ze statków (rezolucja MEPC.310(73)). Aby osiągnąć postępy w realizacji ww. Planu, okólnik zachęca państwa członkowskie i organizacje międzynarodowe do:

- 1. dostarczania IMO wyników wszelkich badań prowadzonych nad tym zagadnieniem, w tym wszelkich informacji o obszarach zanieczyszczonych odpadami morskimi pochodzącymi ze statków; oraz
- 2. podejmowania badań w celu lepszego zrozumienia mikro plastiku pochodzącego ze statków i przekazywania wyników takich badań Organizacji.

### MEPC.1/Circ.895 Ujednolicone interpretacje do Kodeksu Technicznego NOx z 2008 r., z późniejszymi zmianami

Okólnik zawiera uaktualniony tekst wszystkich istniejących ujednoliconych interpretacji do Kodeksu Technicznego NOx z 2008 r., z późniejszymi zmianami, również tych zawartych w okólniku MEPC.1/Circ.865. W okólniku mowa jest o paragrafach: 2.2.4.1, 4.4.6.1, 4.4.6.2 oraz 5.10.1.

### MEPC.1/Circ.884/Rev.1 Wytyczne dotyczące najlepszych praktyk dla państw członkowskich/państw nadbrzeżnych

Zawarte w okólniku najlepsze praktyki mają pomóc państwom członkowskim w wypełnianiu ich obowiązków wynikających z Załącznika VI do Konwencji MARPOL. Odzwierciedlają one zbiór celów, do których należy dążyć, aby zapewnić, że paliwo używane na statkach spełnia wymagania ustawowe, m.in. w ramach krajowych rozwiązań regulacyjnych, należy dążyć do rozwiązania problemu wiarygodności lokalnych dostawców paliwa, podlegających jurysdykcji państwa członkowskiego/państwa nadbrzeżnego. Okólnik zawiera również propozycję formularza licencji dostawcy paliwa.

### MEPC.1/Circ.850/Rev.3 Wytyczne dla określania minimalnych wartości napędu dla utrzymania manewrowości w warunkach niekorzystnych

Celem niniejszych wytycznych jest pomoc administracjom i uznanym organizacjom w weryfikacji, czy statki spełniające wymagania w zakresie projektowego współczynnika efektywności energetycznej (EEDI), określone w przepisach dotyczących efektywności energetycznej statków, posiadają wystarczającą zainstalowaną moc napędową do utrzymania zdolności manewrowej w niekorzystnych warunkach, zgodnie z prawidłem 21.5 rozdziału 4 Załącznika VI do Konwencji MARPOL. Wytyczne stosuje się do statków, które muszą spełnić wymagania przepisów dotyczących efektywności energetycznej statków zgodnie z prawidłem 24 Załącznika VI do Konwencji MARPOL (tj. do tych typów statków o nośności 20.000 DWT i powyżej, które wymieniono w tabeli 1 Dodatku 1.

#### MEPC.1/Circ.795/Rev.5 Ujednolicone interpretacje do Załącznika VI Konwencji MARPOL

Okólnik zawiera skonsolidowany tekst wszystkich istniejących ujednoliconych interpretacji do Załącznika VI Konwencji MARPOL dotyczących takich zagadnień jak: definicji nowego statku, znacznej przebudowy, statków przeznaczonych do przewozu soków owocowych w chłodniach ładunkowych, terminów w których istniejące statki muszą posiadać na pokładzie plan SEEMP, sekcji 2.3 Dodatku do certyfikatu IAPP, identycznych silników zastępczych, czasu wymiany silnika, wymagań dotyczących rejestracji przy zmianie/wyłączeniu silnika z eksploatacji w rejonie NECA NOx (*Engine changeover/on-off recording requirements*), stosowania limitu siarki do urządzeń awaryjnych

Nr 5/2021

(Application of sulphur limit to emergency equipment), Planu VOC (Volatile Organic Compounds Management Plan), spalarek okrętowych z ciągłym podawaniem wsadu, stosowania wymogów dotyczących dokumentu dostawy paliwa (kwitu bunkrowego), potwierdzenia zgodności z wymaganiami dla SEEMP dla nowych statków, uwzględniania w raportowaniu DCS gazu odparowanego i użytego następnie do spalenia oraz dostępu do danych zebranych zgodnie z prawidłem 22A, w sprawie gromadzenia i zgłaszania danych dotyczących zużycia paliwa olejowego przez statki.

#### Inne okólniki dotyczące ochrony środowiska morskiego

### BWM.2/Circ.74 Porozumienie z WMU dotyczące fazy budowania wiedzy i doświadczenia związanej z Konwencją BWM

Okólnik informuje o zawarciu umowy ze Światowym Uniwersytetem Morskim (WMU) na gromadzenie danych, ich analizę oraz opracowanie raportu z tej analizy, który zostanie przedłożony Komitetowi MEPC w celu wsparcia i uzupełnienia EBP (*Experience-building phase*) związanej z Konwencją BWM.

#### PPR.1/Circ.11 Ocena zagrożenia substancjami przewożonymi przez statki

Okólnik zawiera raport z 58 sesji Grupy Roboczej GESAMP/EHS dotyczący oceny zagrożenia szkodliwymi substancjami przewożonymi przez statki. Raport zawiera m. in. definicje, listę omawianych substancji wraz z ich charakterystyką oraz mechanizmy finansowania.

### LC-LP.1/Circ.95 Konwencja o zapobieganiu zanieczyszczaniu mórz przez zatapianie odpadów i innych substancji, 1972 i jej Protokół z 1996

Okólnik zawiera Raport końcowy dotyczący przypadków zatapiania odpadów w 2017 r. Dodatkowo informuje, iż w 2017 r. Konwencję ratyfikowało 87 państw, a 50 państw posiadało status Umawiających się Stron Protokołu. Raporty dotyczące ich aktywności w zakresie zatapiania odpadów przedstawiło 36. państw.

### LC-LP.1/Circ.96 Konwencja o zapobieganiu zanieczyszczaniu mórz przez zatapianie odpadów i innych substancji, 1972 i jej Protokół z 1996

Okólnik zawiera wspólny program prac na lata 2021-2023.

### LC-LP.1/Circ.97 Konwencja o zapobieganiu zanieczyszczaniu mórz przez zatapianie odpadów i innych substancji, 1972 i jej Protokół z 1996

Okólnik stanowi zaproszenie do przesyłania informacji na temat działań związanych z usuwaniem odpadów i innych substancji do morza, przeprowadzanych w ramach jurysdykcji państw za rok 2020. Zaproszenie jest również skierowane do umawiających się stron Konwencji Londyńskiej i Protokołu Londyńskiego. Termin nadsyłania informacji to <u>1 listopada 2021 r.</u>

### LC-LP.1/Circ.98 Konwencja o zapobieganiu zanieczyszczaniu mórz przez zatapianie odpadów i innych substancji, 1972 i jej Protokół z 1996

Plan strategiczny LP/LC, który został przyjęty w 2016 roku, ma na celu ułatwienie wdrażania Protokołu Londyńskiego i Konwencji Londyńskiej, aby przyczynić się do zapobiegania zanieczyszczeniu mórz oraz do postępu w realizacji Agendy 2030 na rzecz zrównoważonego rozwoju. W planie strategicznym określono strategiczne kierunki i cele, które strony Protokołu Londyńskiego i Konwencji Londyńskiej starają się osiągnąć do 2030 r. Aby ułatwić przegląd planu strategicznego w 2022 r. oraz zgodnie ze zobowiązaniami wynikającymi z Konwencji i Protokołu, wzywa się umawiające się strony do przedłożenia Sekretariatowi IMO informacji, o których mowa w niniejszym okólniku.

Poniżej zamieszczono:

- Rezolucje uchwalone podczas 103 sesji MSC zawarte w załączniku do Raportu posesyjnego MSC 103/21/Add.1 oraz
- Rezolucje uchwalone podczas 76 sesji MEPC zawarte w załącznikach do Raportu posesyjnego MEPC 76/15/Add.1 i MEPC 76/15/Add.2



MARITIME SAFETY COMMITTEE 103rd session Agenda item 21 MSC 103/21/Add.1 14 June 2021 Original: ENGLISH

#### REPORT OF THE MARITIME SAFETY COMMITTEE ON ITS 103RD SESSION

Attached are the annexes to the report of the Maritime Safety Committee on its 103rd session (MSC 103/21).

#### LIST OF ANNEXES

- ANNEX 1 RESOLUTION MSC.482(103) AMENDMENTS TO THE INTERNATIONAL CONVENTION FOR THE SAFETY OF LIFE AT SEA, 1974 (SOLAS 1974)
- ANNEX 2 RESOLUTION MSC.483(103) AMENDMENTS TO THE INTERNATIONAL CODE ON THE ENHANCED PROGRAMME OF INSPECTIONS DURING SURVEYS OF BULK CARRIERS AND OIL TANKERS, 2011 (2011 ESP CODE)
- ANNEX 3 RESOLUTION MSC.484(103) AMENDMENTS TO THE INTERNATIONAL CODE FOR FIRE SAFETY SYSTEMS (FSS CODE)
- ANNEX 4 RESOLUTION MSC.485(103) AMENDMENTS TO THE INTERNATIONAL LIFE-SAVING APPLIANCE CODE (LSA CODE)
- ANNEX 5 RESOLUTION MSC.486(103) AMENDMENTS TO THE INTERNATIONAL CONVENTION ON STANDARDS OF TRAINING, CERTIFICATION AND WATCHKEEPING FOR SEAFARERS, 1978 (STCW 1978)
- ANNEX 6 RESOLUTION MSC.487(103) AMENDMENTS TO PART A OF THE SEAFARERS TRAINING, CERTIFICATION AND WATCHKEEPING CODE (STCW CODE)
- ANNEX 7 RESOLUTION MSC.488(103) AMENDMENTS TO THE REVISED RECOMMENDATION ON TESTING OF LIFE-SAVING APPLIANCES (RESOLUTION MSC.81(70))
- ANNEX 8 OUTCOME OF THE REGULATORY SCOPING EXERCISE FOR THE USE OF MARITIME AUTONOMOUS SURFACE SHIPS (MASS)
- ANNEX 9 RESOLUTION MSC.489(103) RECOMMENDED ACTION TO ADDRESS PIRACY AND ARMED ROBBERY IN THE GULF OF GUINEA
- ANNEX 10 BIENNIAL STATUS REPORT OF THE SDC AND SSE SUB-COMMITTEES
- ANNEX 11 PROVISIONAL AGENDAS OF THE SDC AND SSE SUB-COMMITTEES
- ANNEX 12 BIENNIAL STATUS REPORT OF THE MARITIME SAFETY COMMITTEE
- ANNEX 13 POST-BIENNIAL AGENDA OF THE MARITIME SAFETY COMMITTEE
- ANNEX 14 SUBSTANTIVE ITEMS FOR INCLUSION IN THE AGENDAS FOR MSC 104 AND MSC 105
- ANNEX 15 RESOLUTION MSC.490(103) RECOMMENDED ACTION TO PRIORITIZE COVID-19 VACCINATION OF SEAFARERS
- ANNEX 16 STATEMENTS BY DELEGATIONS AND OBSERVERS

#### RESOLUTION MSC.482(103)) (adopted on 13 May 2021)

#### AMENDMENTS TO THE INTERNATIONAL CONVENTION FOR THE SAFETY OF LIFE AT SEA, 1974 (SOLAS 1974)

#### THE MARITIME SAFETY COMMITTEE,

RECALLING Article 28(b) of the Convention on the International Maritime Organization concerning the functions of the Committee,

RECALLING ALSO article VIII(b) of the International Convention for the Safety of Life at Sea, 1974 ("the Convention"), concerning the amendment procedure applicable to the annex to the Convention, other than to the provisions of chapter I,

HAVING CONSIDERED, at its 103rd session, amendments to the Convention proposed and circulated in accordance with article VIII(b)(i) of the Convention,

1 ADOPTS, in accordance with article VIII(b)(iv) of the Convention, amendments to the Convention, the text of which is set out in the annex to the present resolution;

2 DETERMINES, in accordance with article VIII(b)(vi)(2)(aa) of the Convention, that the said amendments shall be deemed to have been accepted on 1 July 2023, unless, prior to that date, more than one third of the Contracting Governments to the Convention, or Contracting Governments the combined merchant fleets of which constitute not less than 50% of the gross tonnage of the world's merchant fleet, have notified the Secretary-General of their objections to the amendments;

3 INVITES Contracting Governments to the Convention to note that, in accordance with article VIII(b)(vii)(2) of the Convention, the amendments shall enter into force on 1 January 2024, upon their acceptance, in accordance with paragraph 2 above;

4 REQUESTS the Secretary-General, for the purposes of article VIII(b)(v) of the Convention, to transmit certified copies of the present resolution and the text of the amendments contained in the annex to all Contracting Governments to the Convention;

5 REQUESTS ALSO the Secretary-General to transmit copies of this resolution and its annex to Members of the Organization which are not Contracting Governments to the Convention.

#### AMENDMENTS TO THE INTERNATIONAL CONVENTION FOR THE SAFETY OF LIFE AT SEA, 1974 (SOLAS 1974)

#### CHAPTER II-1 CONSTRUCTION – STRUCTURE, SUBDIVISION AND STABILITY, MACHINERY AND ELECTRICAL INSTALLATIONS

#### Part B-4 Stability management

1 The following new regulation 25-1 is added after existing regulation 25 with the associated footnotes:

#### "Regulation 25-1

### Water level detectors on multiple hold cargo ships other than bulk carriers and tankers

- 1 Multiple hold cargo ships other than bulk carriers and tankers constructed on or after 1 January 2024 shall be fitted with water level detectors\* in each cargo hold intended for dry cargoes. Water level detectors are not required for cargo holds located entirely above the freeboard deck.
- <sup>\*</sup> Refer to the *Performance standards for water level detectors on bulk carriers and single hold cargo ships other than bulk carriers* (resolution MSC.188(79)), as may be amended.
- 2 The water level detectors required by paragraph 1 shall:
  - .1 give audible and visual alarms at the navigation bridge, one when the water level above the bottom of the cargo hold reaches a height of not less than 0.3 m, and another at a height not less than 15% of the depth of the cargo hold but not more than 2 m; and
  - .2 be fitted at the aft end of the cargo holds. For cargo holds which are occasionally used for water ballast, an alarm overriding device may be installed. The visual alarms shall clearly discriminate between the two different water levels detected in each hold.
- 3 As an alternative to the water level detector at a height of not less than 0.3 m as per sub-paragraph 2.1, a bilge level sensor\* serving the bilge pumping arrangements required by regulation 35-1 and installed in the cargo hold bilge wells or other suitable location is considered acceptable, subject to:
  - .1 the fitting of the bilge level sensor at a height of not less than 0.3 m at the aft end of the cargo hold; and
  - .2 the bilge level sensor giving audible and visual alarm at the navigation bridge which is clearly distinctive from the alarm given by the other water level detector fitted in the cargo hold.

\* Refer to the *Performance standards for water level detectors on bulk carriers and single hold cargo ships other than bulk carriers* (resolution MSC.188(79)), as may be amended."

#### CHAPTER III LIFE-SAVING APPLIANCES AND ARRANGEMENTS

#### Part B Requirements for ships and life-saving appliances

#### Regulation 33 – Survival craft embarkation and launching arrangements

1 Paragraph 33.2 is replaced by the following:

"2 On cargo ships of 20,000 gross tonnage and upwards, davit-launched lifeboats shall be capable of being launched, utilizing painters where necessary, with the ship making headway at speeds up to 5 knots in calm water."

#### RESOLUTION MSC.483(103) (adopted on 13 May 2021)

#### AMENDMENTS TO THE INTERNATIONAL CODE ON THE ENHANCED PROGRAMME OF INSPECTIONS DURING SURVEYS OF BULK CARRIERS AND OIL TANKERS, 2011 (2011 ESP CODE)

THE MARITIME SAFETY COMMITTEE,

RECALLING Article 28(b) of the Convention on the International Maritime Organization concerning the functions of the Committee,

RECALLING ALSO resolution A.1049(27), by which the Assembly adopted the International Code on the Enhanced Programme of Inspections during Surveys of Bulk Carriers and Oil Tankers, 2011 ("2011 ESP Code"), which has become mandatory under chapter XI-1 of the International Convention for the Safety of Life at Sea, 1974 ("the Convention"),

RECALLING FURTHER article VIII(b) and regulation XI-1/2 of the Convention concerning the procedure for amending the 2011 ESP Code,

HAVING CONSIDERED, at its 103rd session, amendments to the 2011 ESP Code proposed and circulated in accordance with article VIII(b)(i) of the Convention,

1 ADOPTS, in accordance with article VIII(b)(iv) of the Convention, amendments to the 2011 ESP Code, the text of which is set out in the annex to the present resolution;

2 DETERMINES, in accordance with article VIII(b)(vi)(2)(bb) of the Convention, that the said amendments shall be deemed to have been accepted on 1 July 2022 unless, prior to that date, more than one third of the Contracting Governments to the Convention, or Contracting Governments the combined merchant fleets of which constitute not less than 50% of the gross tonnage of the world's merchant fleet, have notified the Secretary-General of their objections to the amendments;

3 INVITES Contracting Governments to the Convention to note that, in accordance with article VIII(b)(vii)(2) of the Convention, the amendments shall enter into force on 1 January 2023 upon their acceptance in accordance with paragraph 2 above;

4 REQUESTS the Secretary-General, for the purposes of article VIII(b)(v) of the Convention, to transmit certified copies of the present resolution and the text of the amendments contained in the annex to all Contracting Governments to the Convention;

5 REQUESTS ALSO the Secretary-General to transmit copies of this resolution and its annex to Members of the Organization which are not Contracting Governments to the Convention.

#### AMENDMENTS TO THE INTERNATIONAL CODE ON THE ENHANCED PROGRAMME OF INSPECTIONS DURING SURVEYS OF BULK CARRIERS AND OIL TANKERS, 2011 (2011 ESP CODE)

#### ANNEX B

#### CODE ON THE ENHANCED PROGRAMME OF INSPECTIONS DURING SURVEYS OF OIL TANKERS

#### Part A

#### CODE ON THE ENHANCED PROGRAMME OF INSPECTIONS DURING SURVEYS OF DOUBLE-HULL OIL TANKERS

#### **ANNEX 2**

#### MINIMUM REQUIREMENTS FOR THICKNESS MEASUREMENTS AT RENEWAL SURVEYS OF DOUBLE-HULL OIL TANKERS

1 In the table for "Minimum requirements for thickness measurements at renewal surveys of double-hull oil tankers", the column for "Renewal Survey No.1" is replaced by the following:

"1 Suspect areas"

#### RESOLUTION MSC.484(103) (adopted on 13 May 2021)

#### AMENDMENTS TO THE INTERNATIONAL CODE FOR FIRE SAFETY SYSTEMS (FSS CODE)

THE MARITIME SAFETY COMMITTEE,

RECALLING Article 28(b) of the Convention on the International Maritime Organization concerning the functions of the Committee,

RECALLING ALSO resolution MSC.98(73), by which it adopted the International Code for Fire Safety Systems ("the FSS Code"), which has become mandatory under chapter II-2 of the International Convention for the Safety of Life at Sea, 1974 ("the Convention"),

RECALLING FURTHER article VIII(b) and regulation II-2/3.22 of the Convention concerning the procedure for amending the FSS Code,

HAVING CONSIDERED, at its 103rd session, amendments to the FSS Code, proposed and circulated in accordance with article VIII(b)(i) of the Convention,

1 ADOPTS, in accordance with article VIII(b)(iv) of the Convention, amendments to the FSS Code, the text of which is set out in the annex to the present resolution;

2 DETERMINES, in accordance with article VIII(b)(vi)(2)(aa) of the Convention, that the amendments shall be deemed to have been accepted on 1 July 2023 unless, prior to that date, more than one third of the Contracting Governments to the Convention, or Contracting Governments the combined merchant fleets of which constitute not less than 50% of the gross tonnage of the world's merchant fleet, have notified their objections to the amendments;

3 INVITES Contracting Governments to note that, in accordance with article VIII(b)(vii)(2) of the Convention, the amendments shall enter into force on 1 January 2024 upon their acceptance in accordance with paragraph 2 above;

4 REQUESTS the Secretary-General, in conformity with article VIII(b)(v) of the Convention, to transmit certified copies of the present resolution and the text of the amendments contained in the annex to all Contracting Governments to the Convention;

5 REQUESTS ALSO the Secretary-General to transmit copies of this resolution and its annex to Members of the Organization which are not Contracting Governments to the Convention.

### AMENDMENTS TO THE INTERNATIONAL CODE FOR FIRE SAFETY SYSTEMS (FSS CODE)

#### CHAPTER 9 FIXED FIRE DETECTION AND FIRE ALARM SYSTEMS

#### 2 Engineering specifications

#### 2.1 General requirements

1 The following new paragraph 2.1.8 is inserted after existing paragraph 2.1.7:

"2.1.8 In cargo ships and on passenger ship cabin balconies, where an individually identifiable system is fitted, notwithstanding the provisions in paragraph 2.1.6.1, isolator modules need not be provided at each fire detector if the system is arranged in such a way that the number and location of individually identifiable fire detectors rendered ineffective due to a fault would not be larger than an equivalent section in a section identifiable system, arranged in accordance with paragraph 2.4.1."

#### RESOLUTION MSC.485(103) (adopted on 13 May 2021)

#### AMENDMENTS TO THE INTERNATIONAL LIFE-SAVING APPLIANCE CODE (LSA CODE)

THE MARITIME SAFETY COMMITTEE,

RECALLING Article 28(b) of the Convention on the International Maritime Organization concerning the functions of the Committee,

RECALLING ALSO resolution MSC.48(66), by which it adopted the International Life-Saving Appliance (LSA) Code ("the LSA Code"), which has become mandatory under chapter III of the International Convention for the Safety of Life at Sea, 1974 ("the Convention"),

RECALLING FURTHER article VIII(b) and regulation III/3.10 of the Convention concerning the procedure for amending the LSA Code,

HAVING CONSIDERED, at its 103rd session, amendments to the LSA Code proposed and circulated in accordance with article VIII(b)(i) of the Convention,

1 ADOPTS, in accordance with article VIII(b)(iv) of the Convention, amendments to the LSA Code, the text of which is set out in the annex to the present resolution;

2 DETERMINES, in accordance with article VIII(b)(vi)(2)(aa) of the Convention, that the amendments shall be deemed to have been accepted on 1 July 2023 unless, prior to that date, more than one third of the Contracting Governments to the Convention, or Contracting Governments the combined merchant fleets of which constitute not less than 50% of the gross tonnage of the world's merchant fleet, have notified their objections to the amendments;

3 INVITES Contracting Governments to note that, in accordance with article VIII(b)(vii)(2) of the Convention, the amendments shall enter into force on 1 January 2024 upon their acceptance in accordance with paragraph 2 above;

4 REQUESTS the Secretary-General, in conformity with article VIII(b)(v) of the Convention, to transmit certified copies of the present resolution and the text of the amendments contained in the annex to all Contracting Governments to the Convention;

5

REQUESTS ALSO the Secretary-General to transmit copies of this resolution and its annex to Members of the Organization which are not Contracting Governments to the Convention.

#### AMENDMENTS TO THE INTERNATIONAL LIFE-SAVING APPLIANCES CODE (LSA CODE)

#### CHAPTER IV SURVIVAL CRAFT

#### 4.4 General requirements for lifeboats

- 1 Paragraph 4.4.1.3.2 is replaced by the following:
  - ".2 except for free-fall lifeboats, be capable of being launched and towed when the ship is making headway at speeds up to 5 knots in calm water."

#### RESOLUTION MSC.486(103) (adopted on 13 May 2021)

#### AMENDMENTS TO THE INTERNATIONAL CONVENTION ON STANDARDS OF TRAINING, CERTIFICATION AND WATCHKEEPING FOR SEAFARERS, 1978 (STCW 1978)

THE MARITIME SAFETY COMMITTEE,

RECALLING Article 28(b) of the Convention on the International Maritime Organization concerning the functions of the Committee,

RECALLING ALSO article XII of the International Convention on Standards of Training, Certification and Watchkeeping for Seafarers, 1978 ("1978 STCW Convention"), concerning the procedures for amending the 1978 STCW Convention,

NOTING the repeated references to "high-voltage" in the Seafarers' Training, Certification and Watchkeeping Code ("STCW Code"), without a specific definition for this term,

HAVING CONSIDERED, at its 103rd session, amendments to the 1978 STCW Convention proposed and circulated in accordance with article XII(1)(a)(i) thereof,

1 ADOPTS, in accordance with article XII(1)(a)(iv) of the 1978 STCW Convention, amendments to the said Convention, the text of which is set out in the annex to the present resolution;

2 DETERMINES, in accordance with article XII(1)(a)(vii)(2) of the 1978 STCW Convention, that the said amendments shall be deemed to have been accepted on 1 July 2022, unless, prior to that date, more than one third of Parties, or Parties the combined merchant fleets of which constitute not less than 50% of the gross tonnage of the world's merchant shipping of ships of 100 gross register tons or more, have notified the Secretary-General of the Organization that they object to the amendments;

3 INVITES Parties to note that, in accordance with article XII(1)(a)(ix) of the 1978 STCW Convention, the amendments annexed hereto shall enter into force on 1 January 2023 upon their acceptance, in accordance with paragraph 2 above;

4 URGES Parties to implement the amendments to regulation I/1.1 at an early stage;

5 REQUESTS the Secretary-General, for the purposes of article XII(1)(a)(v) of the 1978 STCW Convention, to transmit certified copies of the present resolution and the text of the amendments contained in the annex to all Parties to the 1978 STCW Convention;

6 REQUESTS ALSO the Secretary-General to transmit copies of this resolution and its annex to Members of the Organization which are not Parties to the 1978 STCW Convention.

#### AMENDMENTS TO THE INTERNATIONAL CONVENTION ON STANDARDS OF TRAINING, CERTIFICATION AND WATCHKEEPING FOR SEAFARERS, 1978 (STCW 1978)

#### CHAPTER I General provisions

- 1 In regulation I/1.1, the following new definition is added:
  - ".44 *High-voltage* means an alternating current (AC) or direct current (DC) voltage in excess of 1,000 volts."

#### RESOLUTION MSC.487(103) (adopted on 13 May 2021)

#### AMENDMENTS TO PART A OF THE SEAFARERS' TRAINING, CERTIFICATION AND WATCHKEEPING CODE (STCW CODE)

THE MARITIME SAFETY COMMITTEE,

RECALLING Article 28(b) of the Convention on the International Maritime Organization concerning the functions of the Committee,

RECALLING ALSO article XII and regulation I/1.2.3 of the International Convention on Standards of Training, Certification and Watchkeeping for Seafarers, 1978 ("1978 STCW Convention"), concerning the procedures for amending part A of the Seafarers' Training, Certification and Watchkeeping Code ("STCW Code"),

NOTING that all the functions for the capacity "electro-technical officer", introduced as part of the 2010 Amendments (the Manila Amendments), are provided at the "operational level",

HAVING CONSIDERED, at its 103rd session, amendments to part A of the STCW Code, proposed and circulated in accordance with article XII(1)(a)(i) of the 1978 STCW Convention,

1 ADOPTS, in accordance with article XII(1)(a)(iv) of the 1978 STCW Convention, amendments to the STCW Code, the text of which is set out in the annex to the present resolution;

2 DETERMINES, in accordance with article XII(1)(a)(vii)(2) of the 1978 STCW Convention, that the said amendments to the STCW Code shall be deemed to have been accepted on 1 July 2022, unless, prior to that date, more than one third of Parties, or Parties the combined merchant fleets of which constitute not less than 50% of the gross tonnage of the world's merchant shipping of ships of 100 gross register tons or more, have notified the Secretary-General of the Organization that they object to the amendments;

3 INVITES Parties to note that, in accordance with article XII(1)(a)(ix) of the 1978 STCW Convention, the amendments to the STCW Code annexed hereto shall enter into force on 1 January 2023 upon their acceptance, in accordance with paragraph 2 above;

4 URGES Parties to implement the amendments to section A-I/1 of the STCW Code at an early stage;

5 REQUESTS the Secretary-General, for the purposes of article XII(1)(a)(v) of the 1978 STCW Convention, to transmit certified copies of the present resolution and the text of the amendments contained in the annex to all Parties to the 1978 STCW Convention;

6 REQUESTS ALSO the Secretary-General to transmit copies of this resolution and its annex to Members of the Organization which are not Parties to the 1978 STCW Convention.

#### AMENDMENTS TO PART A OF THE SEAFARERS' TRAINING, CERTIFICATION AND WATCHKEEPING CODE (STCW CODE)

#### CHAPTER I Standards regarding general provisions

1 In section A-I/1, sub-paragraph .3.1 under the definition for "*Operational level*" is replaced by the following:

".3.1 serving as officer in charge of a navigational or engineering watch or as designated duty engineer for periodically unmanned machinery spaces or as electro-technical officer or as radio operator on board a seagoing ship, and"

#### RESOLUTION MSC.488(103) (adopted on 13 May 2021)

#### AMENDMENTS TO THE REVISED RECOMMENDATION ON TESTING OF LIFE-SAVING APPLIANCES (RESOLUTION MSC.81(70))

#### THE MARITIME SAFETY COMMITTEE,

RECALLING Article 28(b) of the Convention on the International Maritime Organization concerning the functions of the Committee,

RECALLING ALSO that the Assembly, when adopting resolution A.689(17) on *Testing of life-saving appliances*, authorized the Committee to keep the annexed Recommendation on testing of life-saving appliances under review and to adopt, when appropriate, amendments thereto,

RECALLING FURTHER that, since the adoption of resolution A.689(17), the Committee has amended the Recommendation annexed thereto by resolutions MSC.54(66) and MSC.81(70), and by circulars MSC/Circ.596, MSC/Circ.615 and MSC/Circ.809,

RECOGNIZING the need to ensure that the references in the *Revised recommendation on testing of life-saving appliances* (resolution MSC.81(70)) are kept up to date,

1 ADOPTS the Amendments to the Revised recommendation on testing of life-saving appliances (resolution MSC.81(70)), set out in the annex to the present resolution;

2 INVITES Contracting Governments to the SOLAS Convention to bring the above amendments to the attention of all parties concerned.

#### AMENDMENTS TO THE REVISED RECOMMENDATION ON TESTING OF LIFE-SAVING APPLIANCES (RESOLUTION MSC.81(70))

#### PART 1– PROTOTYPE TESTS FOR LIFE-SAVING APPLIANCES

#### 5 LIFERAFTS – RIGID AND INFLATABLE

#### 5.17 Additional tests applicable to inflatable liferafts only

#### Material tests

- 1 Paragraph 5.17.13.2.2.7.1 is replaced by the following:
  - ".1 Test for porosity

A specimen of the fabric should be prepared and tested in accordance with ISO 15372:2000, paragraph 6.2.9.2."

- 2 Amend paragraph 5.17.13.2.2.8, as follows:
  - ".2.2.8 Oil resistance
    - .1 When tested by the method prescribed below, after exposing the outer surface to oil IRM 901, for 2 h at 20± 2°C, there should be no separation of coating from textile and no residual tackiness when two exposed faces are pressed together. The coating should not smear when rubbed with a single pass of the finger.
    - .2 The test should be carried out not less than 16 h after vulcanization or curing.
    - .3 The apparatus, preparation of specimens and test procedure should be in accordance with ISO 15372:2000/Amd 1:2021, paragraph 6.2.5. Each coated face should be tested."

#### 11 HYDROSTATIC RELEASE UNITS

#### 11.2 Technical tests

- 3 Paragraph 11.2.5.5.3 is replaced by the following:
  - ".5.3 Test for surface resistance to oil

Number of specimens2 membranesTemperature+18°C to +20°CType of oilA mineral oil meeting the following<br/>requirements:<br/>Aniline point: 120 ± 5°C<br/>Flashpoint: minimum 240°C

Viscosity: 10-25 cSt at 99.0°C

The following oils may be used:

IRM 901 IRM 905 ISO Oil No. 1

Testing period: Requirements: 3 h on each side The material should show no deterioration."

#### PART 2 – PRODUCTION AND INSTALLATION TESTS

#### 5 SURVIVAL CRAFT

#### 5.4 Launch test

4 Paragraph 5.4 is replaced by the following:

"Except in the case of a free-fall lifeboat, it should be demonstrated that the fully equipped lifeboat on cargo ships of 20,000 gross tonnage and upwards and rescue boat can be launched from a ship proceeding ahead at a speed of not less than 5 knots in calm water and on an even keel. There should be no damage to the lifeboat or the rescue boat or their equipment as a result of this test."

#### RESOLUTION MSC.489(103) (adopted on 14 May 2021)

#### RECOMMENDED ACTION TO ADDRESS PIRACY AND ARMED ROBBERY IN THE GULF OF GUINEA

THE MARITIME SAFETY COMMITTEE,

RECALLING the role of the International Maritime Organization (the Organization) in ensuring the safety and security of international shipping,

RECALLING ALSO Article 28(b) of the Convention on the International Maritime Organization concerning the functions of the Committee,

RECALLING FURTHER Assembly resolution A.1069(28) on *Prevention and suppression of piracy, armed robbery against ships and illicit maritime activity in the Gulf of Guinea* urging Governments to cooperate with and assist States in the Gulf of Guinea region to enhance their national and regional capabilities to improve maritime governance in waters under their jurisdiction, while reiterating full respect for the sovereignty, sovereign rights, jurisdiction and territorial integrity of all States and the relevant provisions of international law, in particular the United Nations Convention on the Law of the Sea (UNCLOS),

RECOGNIZING the ongoing work of the United Nations and other relevant organizations and stakeholders in support of the countries of the region in their efforts to prevent and combat piracy and armed robbery against vessels,

BEING DEEPLY CONCERNED about the escalation in the number and severity of attacks in the Gulf of Guinea region which threaten the lives and well-being of seafarers and the safety of shipping in the region,

NOTING WITH APPRECIATION the continuous efforts made by the region to curb piracy and armed robbery against ships in the Gulf of Guinea, including drafting anti-piracy laws, the Deep Blue Project, and the establishment of the GoG Maritime Collaboration Forum (GoG-MCF/SHADE GoG) working with law enforcement and the Interregional Coordination Centre (ICC Yaoundé) and the ongoing establishment of the Yaoundé Architecture Regional Integration System (YARIS), and utilizing useful platforms working in the region, such as the G7++ Friends of the Gulf of Guinea (FOGG) in support of the framework,

NOTING ALSO that IMO and industry are taking actions to address threats posed by piracy and armed robbery attacks against vessels and the kidnapping of seafarers and/or passengers in the Gulf of Guinea, including providing technical assistance to Member States in the region regarding the implementation of maritime security measures, supporting regional initiatives such as the Interregional Coordination Centre (ICC) to assist with the implementation of the Yaoundé Code of Conduct (YCC), and providing Best Management Practices (BMP) West Africa (WA) to assist companies and seafarers to assess the risks associated with voyages through the Gulf of Guinea and mitigate any potential threats to their safety and security,

BEARING IN MIND that greater collaboration with all critical stakeholders on activities is needed, due to a number of challenges encountered by regional countries, including information-sharing on maritime criminality and illegality, maritime domain awareness such as MDAT-GoG (Maritime Domain Awareness for Trade for the Gulf of Guinea) and surface and/or

air patrol capabilities, functional legal frameworks in line with international best practice, capacity-building such as skill development of maritime law enforcement agents, integration of national inter-agency efforts and youth empowerment programmes,

REITERATING the Organization's determination to continue to address piracy and armed robbery against vessels and to promote safe and secure navigation in the waters of the Gulf of Guinea,

1 CALLS ON Member States, national authorities, the United Nations and other relevant organizations to:

- .1 consider strengthening law enforcement, and harmonization of criminal penalties across coastal States to arrest and prosecute pirates in relevant jurisdictions in accordance with international law and national legal frameworks;
- .2 consider options and international best practices for improved governance of available protection solutions such as security escort vessels for assisting other vessels, in accordance with international law, and with due respect for the sovereignty, sovereign rights and territorial integrity of coastal States; and
- .3 support and encourage wider participation in the international framework GoG-MCF/SHADE GoG as well as other platforms, such as G7++FOGG, with a view to improving maritime security and safety in the region and facilitating the strengthening of cooperation mechanisms for regional maritime patrol and protection;

#### 2 REQUESTS the Secretary-General to:

- .1 make full use of its funds to sustain technical cooperation on addressing piracy and armed robbery for the regional countries, such as providing training workshops/webinars and developing courses in relationship with existing initiatives and programmes;
- .2 encourage the creation, subject to available resources, of a common platform to better facilitate the timely and reliable communication of standardized information on piracy or armed robbery between existing mechanisms such as MDAT-GoG, the NIMASA C4i-Centre, Regional Reporting Centres, the ICC IMB Piracy Reporting Centre and relevant responding law enforcement, as well as the subsequent analysis of the reported information; and
- .3 sustain its efforts in addressing piracy and armed robbery in the Gulf of Guinea in coordination with Member States, the United Nations and the industry.

3 CALLS UPON Member States, international organizations and relevant stakeholders to consider making financial contributions to the IMO West and Central Africa Maritime Security Trust Fund, which remains open to support the Organization's maritime security capacity-building programme to assist Gulf of Guinea coastal States as well as regional centres under the YCC architecture; and 4 INVITES the Secretary-General, Member States, international organizations and relevant stakeholders to bring this resolution to the attention of all parties concerned.

#### RESOLUTION MSC.490(103) (adopted 14 May 2021)

#### **RECOMMENDED ACTION TO PRIORITIZE COVID-19 VACCINATION OF SEAFARERS**

THE MARITIME SAFETY COMMITTEE,

RECALLING Article 28(b) of the Convention on the International Maritime Organization concerning the functions of the Committee,

RECALLING ALSO that the Council, at its thirty-first extraordinary session, urged flag and port States to ensure the welfare of seafarers, in particular that their rights to wages, shore leave, sick leave, access to medical care, food supplies and repatriation were preserved,

RECALLING FURTHER the statement adopted by Council, at its thirty-second extraordinary session, on recognition of the importance of the flow of trade by sea during the COVID-19 pandemic,

RECALLING that, at its second extraordinary session, it adopted resolution MSC.473(ES.2) on *Recommended action to facilitate ship crew change, access to medical care and seafarer travel during the COVID-19 pandemic,* 

RECALLING ALSO that, at its 102nd session, it approved a circular containing the *Industry recommended framework of protocols for ensuring safe ship crew changes and travel during the coronavirus (COVID-19) pandemic* (MSC.1/Circ.1636), recognizing the importance of the protocols and inviting Member States and relevant stakeholders to consider them and take them into account,

NOTING resolution A/RES/75/17 on International cooperation to address challenges faced by seafarers as a result of the COVID-19 pandemic to support global supply chains, adopted by the UN General Assembly on 1 December 2020,

NOTING ALSO resolution GB.340/Resolution (Rev.2) – *Resolution concerning maritime labour issues and the COVID-19 pandemic*, adopted by the Governing Body of the International Labour Organization on 8 December 2020,

NOTING FURTHER the Resolution concerning the implementation and practical application of the MLC, 2006, during the COVID-19 pandemic, and the Resolution concerning COVID-19 vaccination for seafarers, adopted by the ILO Special Tripartite Committee of the Maritime Labour Convention, 2006, as amended (MLC, 2006) on 23 April 2021,<sup>1</sup>

REAFFIRMING the unique and essential work of seafarers for international shipping and for the world, contributing to the uninterrupted transportation of vital medical supplies, food and other basic necessities during the COVID-19 pandemic,

NOTING the Joint Statement of 25 March 2021 issued by ICAO, ILO, IMO, WHO and IOM (Circular Letter No.4204/Add.38), urging all United Nations Member States to prioritize seafarers and aircrew in their national COVID-19 vaccination programmes, to protect them through vaccination as soon as possible and to facilitate their safe movement across borders,

<sup>&</sup>lt;sup>1</sup> https://www.ilo.org/global/standards/maritime-labour-convention/events/WCMS\_679152/lang--en/index.htm

NOTING ALSO the Statement on the seventh meeting of the International Health Regulations (2005) Emergency Committee regarding the coronavirus disease (COVID-19) pandemic,<sup>2</sup> recommending IHR States Parties to prioritize vaccination for seafarers and aircrews in line with the aforementioned Joint Statement,

NOTING FURTHER that, given the limited period of time ships are in port, single dose COVID-19 vaccines for seafarers would be preferable,

BEING CONSCIOUS of the limited and highly uneven access to COVID-19 vaccines around the world,

BEING ALSO CONSCIOUS of the need to protect seafarers through vaccination, as soon as possible, to facilitate their safe movement across borders,

BEING CONSCIOUS of the importance of cooperation among Member States,

NOTING that the UN Crisis Management Team for COVID-19, under the leadership of WHO, recognized that all countries should consider seafarers for essential allocation of vaccines,

BEING AWARE OF the possibility that countries may introduce requirements for proof of COVID-19 vaccination for international travel of seafarers as a condition of entry, notwithstanding WHO's recommendations,<sup>3</sup>

- 1 RECOMMENDS that Member States and relevant national authorities:
  - .1 prioritize their seafarers, as far as practicable, in their national COVID-19 vaccination programmes, noting the advice of the WHO SAGE Roadmap<sup>4</sup> for prioritizing the use of COVID-19 vaccines in the context of limited supply published in November 2020. Proper consideration of extending COVID-19 vaccines to seafarers of other nationalities is also recommended, taking into account their national vaccines supply;
  - .2 consider exempting seafarers from any national policy requiring proof of COVID-19 vaccination as a condition for entry, taking into account that seafarers should be designated as "key workers", as they travel across borders frequently;
  - .3 develop appropriate plans, where feasible, to provide necessary infrastructure and facilities to support COVID-19 vaccination of seafarers;

2 INVITES Member States, international organizations, shipping companies and other stakeholders to inform seafarers about the safety and possible benefits of the COVID-19 vaccination, bearing in mind that taking the vaccination is a decision made on an individual basis;

<sup>&</sup>lt;sup>2</sup> https://www.who.int/news/item/19-04-2021-statement-on-the-seventh-meeting-of-the-international-health-regulations-(2005)-emergency-committee-regarding-the-coronavirus-disease-(covid-19)-pandemic

<sup>&</sup>lt;sup>3</sup> Statement on the sixth meeting of the International Health Regulations (2005) Emergency Committee regarding the coronavirus disease (COVID-19) pandemic: https://www.who.int/news/item/15-01-2021-statement-on-the-sixth-meeting-of-the-international-health-regulations-(2005)-emergency-committee-regarding-the-coronavirus-disease-(covid-19)-pandemic

<sup>&</sup>lt;sup>4</sup> WHO SAGE Roadmap For Prioritizing Uses Of COVID-19 Vaccines In The Context Of Limited Supply: https://www.who.int/publications/m/item/who-sage-roadmap-for-prioritizing-uses-of-covid-19-vaccines-inthe-context-of-limited-supply
3 REQUESTS Member States and international organizations to bring this resolution to the attention of all parties concerned.

\*\*\*



MARINE ENVIRONMENT PROTECTION COMMITTEE 76th session Agenda item 15 MEPC 76/15/Add.1 24 August 2021 Original: ENGLISH

Ε

# REPORT OF THE MARINE ENVIRONMENT PROTECTION COMMITTEE ON ITS SEVENTY-SIXTH SESSION

Attached is annex 1 to the report of the Marine Environment Protection Committee on its seventy-sixth session (MEPC 76/15).



# LIST OF ANNEXES

ANNEX 1 RESOLUTION MEPC.328(76) - AMENDMENTS TO MARPOL ANNEX VI (2021 REVISED MARPOL ANNEX VI)

(See document MEPC 76/15/Add.2 for annexes 2 to 20)

\*\*\*

# ANNEX 1

### **RESOLUTION MEPC.328(76)**

#### AMENDMENTS TO THE ANNEX OF THE PROTOCOL OF 1997 TO AMEND THE INTERNATIONAL CONVENTION FOR THE PREVENTION OF POLLUTION FROM SHIPS, 1973, AS MODIFIED BY THE PROTOCOL OF 1978 RELATING THERETO

#### 2021 Revised MARPOL Annex VI

THE MARINE ENVIRONMENT PROTECTION COMMITTEE,

RECALLING Article 38(a) of the Convention on the International Maritime Organization concerning the functions of the Marine Environment Protection Committee (the Committee) conferred upon it by international conventions for the prevention and control of marine pollution from ships,

RECALLING ALSO article 16 of the International Convention for the Prevention of Pollution from Ships, 1973, as modified by the Protocols of 1978 and 1997 relating thereto (MARPOL), which specifies the amendment procedure and confers upon the appropriate body of the Organization the function of considering amendments thereto for adoption by the Parties,

RECALLING FURTHER that the Committee, at its seventy-second session, adopted resolution MEPC.304(72) on the *Initial IMO Strategy on reduction of GHG emissions from ships*,

HAVING CONSIDERED, at its seventy-sixth session, proposed amendments to MARPOL Annex VI concerning mandatory goal-based technical and operational measures to reduce carbon intensity of international shipping and exemption of unmanned non-self-propelled (UNSP) barges from certain survey and certification requirements, which were circulated in accordance with article 16(2)(a) of MARPOL,

HAVING ALSO CONSIDERED, at its seventy-sixth session, the comprehensive assessment of the impacts of the proposed amendments to MARPOL Annex VI on States, including on developing countries, especially on least developed countries (LDCs) and small island developing States (SIDS),\*

1 ADOPTS, in accordance with article 16(2)(d) of MARPOL, the amendments to MARPOL Annex VI, the text of which is set out in the annex to the present resolution;

2 DETERMINES, in accordance with article 16(2)(f)(iii) of MARPOL, that the amendments to MARPOL Annex VI shall be deemed to have been accepted on 1 May 2022 unless prior to that date not less than one third of the Parties or Parties the combined merchant fleets of which constitute not less than 50% of the gross tonnage of the world's merchant fleet have communicated to the Organization their objection to the amendments;

3 INVITES the Parties to note that, in accordance with article 16(2)(g)(ii) of MARPOL, the amendments to MARPOL Annex VI shall enter into force on 1 November 2022 upon its acceptance in accordance with paragraph 2 above;

4 INVITES ALSO the Parties to consider and initiate as soon as possible the development of a Carbon Intensity Code;

<sup>\*</sup> As set out in documents MEPC 76/7/13, MEPC 76/INF.68, and MEPC 76/INF.68/Add.1, Add.2 and Add.3.

5 INVITES the Organization, mindful of the review clauses provided for in regulations 25.3 and 28.11 of the amendments to MARPOL Annex VI, to initiate the respective reviews as early as possible;

6 INVITES ALSO the Organization to keep under review the impacts on States of the aforesaid amendments to MARPOL Annex VI, paying particular attention to the needs of developing countries, especially LDCs and SIDS, so that any necessary adjustments can be made;

7 AGREES to undertake a lessons-learned exercise from the comprehensive impact assessment of the amendments to MARPOL Annex VI, with a view to improving the procedure for conducting future impact assessments taking into account the *Procedure for assessing impacts on States of candidate measures* (MEPC.1/Circ.885) and the terms of reference for the impact assessment of the short-term measure;<sup>†</sup>

8 ENCOURAGES the Parties to consider early application of the aforesaid amendments;

9 REQUESTS the Secretary-General, for the purposes of article 16(2)(e) of MARPOL, to transmit certified copies of the present resolution and the text of the amendments to MARPOL Annex VI contained in the annex to all Parties to MARPOL;

10 REQUESTS ALSO the Secretary-General to transmit copies of the present resolution and its annex to Members of the Organization which are not Parties to MARPOL.

<sup>&</sup>lt;sup>†</sup> As set out in the Terms of reference and arrangements for the conduct of a comprehensive impact assessment of the short-term measure before MEPC 76 (MEPC 75/18, annex 6).

# ANNEX

# MARPOL ANNEX VI

# **REGULATIONS FOR THE PREVENTION OF AIR POLLUTION FROM SHIPS**

### Chapter 1 – General

# Regulation 1

Application

The provisions of this Annex shall apply to all ships, except where expressly provided otherwise.

# **Regulation 2**

Definitions

- 1 For the purpose of this Annex:
  - .1 Annex means Annex VI to the International Convention for the Prevention of Pollution from Ships, 1973 (MARPOL), as modified by the Protocol of 1978 relating thereto, and as modified by the Protocol of 1997, as amended by the Organization, provided that such amendments are adopted and brought into force in accordance with the provisions of article 16 of the present Convention.
  - .2 A similar stage of construction means the stage at which:
    - .1 construction identifiable with a specific ship begins; and
    - .2 assembly of that ship has commenced comprising at least 50 tonnes or one per cent of the estimated mass of all structural material, whichever is less.
  - .3 Anniversary date means the day and the month of each year that will correspond to the date of expiry of the International Air Pollution Prevention Certificate.
  - .4 *Audit* means a systematic, independent and documented process for obtaining audit evidence and evaluating it objectively to determine the extent to which audit criteria are fulfilled.
  - .5 *Audit Scheme* means the IMO Member State Audit Scheme established by the Organization and taking into account the guidelines developed by the Organization.<sup>1</sup>
  - .6 *Audit Standard* means the Code for Implementation.
  - .7 Auxiliary control device means a system, function or control strategy installed on a marine diesel engine that is used to protect the engine and/or its ancillary equipment against operating conditions that could result in damage or failure, or that is used to facilitate the starting of the engine. An auxiliary control device

<sup>&</sup>lt;sup>1</sup> Refer to the Framework and Procedures for the IMO Member State Audit Scheme (resolution A.1067(28)).

may also be a strategy or measure that has been satisfactorily demonstrated not to be a defeat device.

- .8 Code for Implementation means the IMO Instruments Implementation Code (III Code) adopted by the Organization by resolution A.1070(28).
- .9 *Continuous feeding* is defined as the process whereby waste is fed into a combustion chamber without human assistance while the incinerator is in normal operating conditions with the combustion chamber operative temperature between 850°C and 1,200°C.
- .10 Defeat device means a device that measures, senses or responds to operating variables (e.g. engine speed, temperature, intake pressure or any other parameter) for the purpose of activating, modulating, delaying or deactivating the operation of any component or the function of the emission control system such that the effectiveness of the emission control system is reduced under conditions encountered during normal operation, unless the use of such a device is substantially included in the applied emission certification test procedures.
- .11 *Electronic Record Book* means a device or system, approved by the Administration, used to electronically record the required entries for discharges, transfers and other operations as required under this Annex in lieu of a hard copy record book.<sup>2</sup>
- .12 *Emission* means any release of substances, subject to control by this Annex, from ships into the atmosphere or sea.
- .13 *Emission control area* means an area where the adoption of special mandatory measures for emissions from ships is required to prevent, reduce and control air pollution from  $NO_X$  or  $SO_X$  and particulate matter or all three types of emissions and their attendant adverse impacts on human health and the environment. Emission control areas shall include those listed in, or designated under, regulations 13 and 14 of this Annex.
- .14 *Fuel oil* means any fuel delivered to and intended for combustion purposes for propulsion or operation on board a ship, including gas, distillate and residual fuels.
- .15 *Gross tonnage* means the gross tonnage calculated in accordance with the tonnage measurement regulations contained in Annex I to the International Convention on Tonnage Measurements of Ships, 1969, or any successor Convention.
- .16 *In-use sample* means a sample of fuel oil in use on a ship.
- .17 *Installations* in relation to regulation 12 of this Annex means the installation of systems, equipment, including portable fire-extinguishing units, insulation, or other material on a ship, but excludes the repair or recharge of previously installed systems, equipment, insulation or other material, or the recharge of portable fire-extinguishing units.

<sup>&</sup>lt;sup>2</sup> Refer to the *Guidelines for the use of electronic record books under MARPOL* (resolution MEPC.312(74)).

- .18 *Installed* means a marine diesel engine that is or is intended to be fitted on a ship, including a portable auxiliary marine diesel engine, only if its fuelling, cooling or exhaust system is an integral part of the ship. A fuelling system is considered integral to the ship only if it is permanently affixed to the ship. This definition includes a marine diesel engine that is used to supplement or augment the installed power capacity of the ship and is intended to be an integral part of the ship.
- .19 *Irrational emission control strategy* means any strategy or measure that, when the ship is operated under normal conditions of use, reduces the effectiveness of an emission control system to a level below that expected on the applicable emission test procedures.
- .20 *Low-flashpoint fuel* means gaseous or liquid fuel oil having a flashpoint lower than otherwise permitted under paragraph 2.1.1 of regulation 4 of chapter II-2 of the International Convention for the Safety of Life at Sea (SOLAS), 1974, as amended.
- .21 *Marine diesel engine* means any reciprocating internal combustion engine operating on liquid or dual fuel, to which regulation 13 of this Annex applies, including booster/compound systems if applied. In addition, a gas-fuelled engine installed on a ship constructed on or after 1 March 2016 or a gas-fuelled additional or non-identical replacement engine installed on or after that date is also considered as a marine diesel engine.
- .22 *MARPOL delivered sample* means the sample of fuel oil delivered in accordance with regulation 18.8.1 of this Annex.
- .23 NO<sub>X</sub> Technical Code means the Technical Code on Control of Emission of Nitrogen Oxides from Marine Diesel Engines adopted by resolution 2 of the 1997 MARPOL Conference, as amended by the Organization, provided that such amendments are adopted and brought into force in accordance with the provisions of article 16 of the present Convention.
- .24 *Onboard sample* means a sample of fuel oil intended to be used or carried for use on board that ship.
- .25 *Ozone-depleting substances* means controlled substances defined in paragraph (4) of article 1 of the Montreal Protocol on Substances that Deplete the Ozone Layer, 1987, listed in Annexes A, B, C or E to the said Protocol in force at the time of application or interpretation of this Annex.

Ozone-depleting substances that may be found on board ship include, but are not limited to:

Halon 1211	Bromochlorodifluoromethane
Halon 1301	Bromotrifluoromethane
Halon 2402	1,2-Dibromo-1,1,2,2-tetraflouroethane (also known as Halon 114B2)
CFC-11	Trichlorofluoromethane
CFC-12	Dichlorodifluoromethane
CFC-113	1,1,2-Trichloro-1,2,2-trifluoroethane

CFC-114	1,2-Dichloro-1,1,2,2-tetrafluoroethane
CFC-115	Chloropentafluoroethane

- .26 *Shipboard incineration* means the incineration of wastes or other matter on board a ship, if such wastes or other matter were generated during the normal operation of that ship.
- .27 *Shipboard incinerator* means a shipboard facility designed for the primary purpose of incineration.
- .28 *Ships constructed* means ships the keels of which are laid or that are at a similar stage of construction.
- .29 *Sludge oil* means sludge from the fuel oil or lubricating oil separators, waste lubricating oil from main or auxiliary machinery, or waste oil from bilge water separators, oil filtering equipment or drip trays.
- .30 *Sulphur content of fuel oil* means the concentration of sulphur in a fuel oil, measured in % m/m as tested in accordance with a standard acceptable to the Organization.<sup>3</sup>
- .31 *Tanker* in relation to regulation 15 of this Annex means an oil tanker as defined in regulation 1 of Annex I of the present Convention or a chemical tanker as defined in regulation 1 of Annex II of the present Convention.
- .32 Unmanned non-self-propelled (UNSP) barge means a barge that:
  - .1 is not propelled by mechanical means;
  - .2 has no system, equipment and/or machinery fitted that may generate emissions regulated by this Annex; and
  - .3 has neither persons nor living animals on board.
- 2 For the purpose of chapter 4:
  - .1 A ship delivered on or after 1 September 2019 means a ship:
    - .1 for which the building contract is placed on or after 1 September 2015; or
    - .2 in the absence of a building contract, the keel of which is laid, or which is at a similar stage of construction, on or after 1 March 2016; or
    - .3 the delivery of which is on or after 1 September 2019.
  - .2 Attained annual operational CII is the operational carbon intensity indicator value achieved by an individual ship in accordance with regulations 26 and 28 of this Annex.

<sup>&</sup>lt;sup>3</sup> Refer to ISO 8754:2003 Petroleum products – Determination of sulphur content – Energy-dispersive X-ray fluorescence spectrometry.

- .3 *Attained EEDI* is the EEDI value achieved by an individual ship in accordance with regulation 22 of this Annex.
- .4 *Attained EEXI* is the EEXI value achieved by an individual ship in accordance with regulation 23 of this Annex.
- .5 *Bulk carrier* means a ship which is intended primarily to carry dry cargo in bulk, including such types as ore carriers as defined in regulation 1 of chapter XII of the International Convention for the Safety of Life at Sea (SOLAS), 1974, (as amended) but excluding combination carriers.
- .6 *Calendar year* means the period from 1 January until 31 December inclusive.
- .7 *Combination carrier* means a ship designed to load 100% deadweight with both liquid and dry cargo in bulk.
- .8 *Company* means the owner of the ship or any other organization or person such as the manager, or the bareboat charterer, who has assumed the responsibility for operation of the ship from the owner of the ship and who on assuming such responsibility has agreed to take over all the duties and responsibilities imposed by the International Management Code for the Safe Operation of Ships and for Pollution Prevention, as amended.
- .9 *Containership* means a ship designed exclusively for the carriage of containers in holds and on deck.
- .10 *Conventional propulsion* in relation to chapter 4 means a method of propulsion where a main reciprocating internal combustion engine(s) is the prime mover and coupled to a propulsion shaft either directly or through a gear box.
- .11 *Cruise passenger ship* in relation to chapter 4 means a passenger ship not having a cargo deck, designed exclusively for commercial transportation of passengers in overnight accommodations on a sea voyage.
- .12 *Distance travelled* means distance travelled over ground.
- .13 *Existing ship* means a ship which is not a new ship.
- .14 *Gas carrier* in relation to chapter 4 means a cargo ship, other than an LNG carrier as defined in paragraph 2.16 of this regulation, constructed or adapted and used for the carriage in bulk of any liquefied gas.
- .15 *General cargo ship* means a ship with a multi-deck or single deck hull designed primarily for the carriage of general cargo. This definition excludes specialized dry cargo ships, which are not included in the calculation of reference lines for general cargo ships, namely livestock carrier, barge carrier, heavy load carrier, yacht carrier, nuclear fuel carrier.
- .16 *LNG carrier* in relation to chapter 4 of this Annex means a cargo ship constructed or adapted and used for the carriage in bulk of liquefied natural gas (LNG).
- .17 *Major conversion* means in relation to chapter 4 of this Annex a conversion of a ship:

- .1 which substantially alters the dimensions, carrying capacity or engine power of the ship; or
- .2 which changes the type of the ship; or
- .3 the intent of which in the opinion of the Administration is substantially to prolong the life of the ship; or
- .4 which otherwise so alters the ship that, if it were a new ship, it would become subject to relevant provisions of the present Convention not applicable to it as an existing ship; or
- .5 which substantially alters the energy efficiency of the ship and includes any modifications that could cause the ship to exceed the applicable required EEDI as set out in regulation 24 of this Annex or the applicable required EEXI as set out in regulation 25 of this Annex.
- .18 *New ship* means a ship:
  - .1 for which the building contract is placed on or after 1 January 2013; or
  - .2 in the absence of a building contract, the keel of which is laid or which is at a similar stage of construction on or after 1 July 2013; or
  - .3 the delivery of which is on or after 1 July 2015.
- .19 *Non-conventional propulsion* in relation to chapter 4 of this Annex means a method of propulsion, other than conventional propulsion, including diesel-electric propulsion, turbine propulsion, and hybrid propulsion systems.
- .20 *Passenger ship* means a ship which carries more than 12 passengers.
- .21 *Polar Code* means the International Code for Ships Operating in Polar Waters, consisting of an introduction, parts I-A and II-A and parts I-B and II-B, adopted by resolutions MSC.385(94) and MEPC.264(68), as may be amended, provided that:
  - .1 amendments to the environment-related provisions of the introduction and chapter 1 of part II-A of the Polar Code are adopted, brought into force and take effect in accordance with the provisions of article 16 of the present Convention concerning the amendment procedures applicable to an appendix to an annex; and
  - .2 amendments to part II-B of the Polar Code are adopted by the Marine Environment Protection Committee in accordance with its Rules of Procedure.
- .22 *Refrigerated cargo carrier* means a ship designed exclusively for the carriage of refrigerated cargoes in holds.

- .23 *Required annual operational CII* is the target value of attained annual operational CII in accordance with regulations 26 and 28 of this Annex for the specific ship type and size.
- .24 *Required EEDI* is the maximum value of attained EEDI that is allowed by regulation 24 of this Annex for the specific ship type and size.
- .25 *Required EEXI* is the maximum value of attained EEXI that is allowed by regulation 25 of this Annex for the specific ship type and size.
- .26 *Ro-ro cargo ship* means a ship designed for the carriage of roll-on-roll-off cargo transportation units.
- .27 *Ro-ro cargo ship (vehicle carrier)* means a multi-deck roll-on-roll-off cargo ship designed for the carriage of empty cars and trucks.
- .28 *Ro-ro passenger ship* means a passenger ship with roll-on-roll-off cargo spaces.
- .29 *Tanker* means an oil tanker as defined in regulation 1 of Annex I of the present Convention or a chemical tanker or an NLS tanker as defined in regulation 1 of Annex II of the present Convention.

# **Regulation 3**

Exceptions and exemptions

### General

- 1 Regulations of this Annex shall not apply to:
  - .1 any emission necessary for the purpose of securing the safety of a ship or saving life at sea; or
  - .2 any emission resulting from damage to a ship or its equipment:
    - .2.1 provided that all reasonable precautions have been taken after the occurrence of the damage or discovery of the emission for the purpose of preventing or minimizing the emission; and
    - .2.2 except if the owner or the master acted either with intent to cause damage, or recklessly and with knowledge that damage would probably result.

### Trials for ship emission reduction and control technology research

2 The Administration of a Party may, in cooperation with other Administrations as appropriate, issue an exemption from specific provisions of this Annex for a ship to conduct trials for the development of ship emission reduction and control technologies and engine design programmes. Such an exemption shall only be provided if the applications of specific provisions of the Annex or the revised NO<sub>X</sub> Technical Code 2008 could impede research into the development of such technologies or programmes. A permit issued under this regulation shall not exempt a ship from the reporting requirement under regulation 27 and shall not alter the type and scope of data required to be reported under regulation 27. A permit for such an exemption

shall only be provided to the minimum number of ships necessary and be subject to the following provisions:

- .1 for marine diesel engines with a per cylinder displacement up to 30 L, the duration of the sea trial shall not exceed 18 months. If additional time is required, a permitting Administration or Administrations may permit a renewal for one additional 18-month period; or
- .2 for marine diesel engines with a per cylinder displacement at or above 30 L, the duration of the ship trial shall not exceed five years and shall require a progress review by the permitting Administration or Administrations at each intermediate survey. A permit may be withdrawn based on this review if the testing has not adhered to the conditions of the permit or if it is determined that the technology or programme is not likely to produce effective results in the reduction and control of ship emissions. If the reviewing Administration or Administrations determine that additional time is required to conduct a test of a particular technology or programme, a permit may be renewed for an additional time period not to exceed five years.

# Emissions from seabed mineral activities

3.1 Emissions directly arising from the exploration, exploitation and associated offshore processing of seabed mineral resources are, consistent with article 2(3)(b)(ii) of the present Convention, exempt from the provisions of this Annex. Such emissions include the following:

- .1 emissions resulting from the incineration of substances that are solely and directly the result of exploration, exploitation and associated offshore processing of seabed mineral resources, including but not limited to the flaring of hydrocarbons and the burning of cuttings, muds, and/or stimulation fluids during well completion and testing operations, and flaring arising from upset conditions;
- .2 the release of gases and volatile compounds entrained in drilling fluids and cuttings;
- .3 emissions associated solely and directly with the treatment, handling or storage of seabed minerals; and
- .4 emissions from marine diesel engines that are solely dedicated to the exploration, exploitation and associated offshore processing of seabed mineral resources.

3.2 The requirements of regulation 18 of this Annex shall not apply to the use of hydrocarbons that are produced and subsequently used on site as fuel, when approved by the Administration.

### Unmanned non-self-propelled barges

4 The Administration may exempt an unmanned non-self-propelled (UNSP) barge<sup>4</sup> from the requirements of regulations 5.1 and 6.1 of this Annex by means of an International Air Pollution Prevention Exemption Certificate for Unmanned Non-self-propelled (UNSP) Barges,

<sup>&</sup>lt;sup>4</sup> Refer to the *Guidelines for exemption of unmanned non-self-propelled* (UNSP) *barges from the survey and certification requirements under the MARPOL Convention* (MEPC.1/Circ.892).

for a period not exceeding five years provided that the barge has undergone a survey to confirm that conditions referred to in regulations 2.1.32.1 to 2.1.32.3 of this Annex are met.

# **Regulation 4**

Equivalents

1 The Administration of a Party may allow any fitting, material, appliance or apparatus to be fitted in a ship or other procedures, alternative fuel oils, or compliance methods used as an alternative to those required by this Annex if such fitting, material, appliance or apparatus or other procedures, alternative fuel oils, or compliance methods are at least as effective in terms of emissions reductions as those required by this Annex, including any of the standards set forth in regulations 13 and 14.

2 The Administration of a Party that allows a fitting, material, appliance or apparatus or other procedures, alternative fuel oils, or compliance methods used as an alternative to those required by this Annex shall communicate to the Organization for circulation to the Parties particulars thereof, for their information and appropriate action, if any.

3 The Administration of a Party should take into account any relevant guidelines developed by the Organization<sup>5</sup> pertaining to the equivalents provided for in this regulation.

4 The Administration of a Party that allows the use of an equivalent as set forth in paragraph 1 of this regulation shall endeavour not to impair or damage its environment, human health, property or resources or those of other States.

# Chapter 2 – Survey, certification and means of control

# **Regulation 5**

Surveys

1 Every ship of 400 gross tonnage and above and every fixed and floating drilling rig or other platform shall, to ensure compliance with the requirements of chapter 3 of this Annex, be subject to the surveys specified below:

- .1 An initial survey before the ship is put into service or before the certificate required under regulation 6 of this Annex is issued for the first time. This survey shall be such as to ensure that the equipment, systems, fittings, arrangements and material fully comply with the applicable requirements of chapter 3 of this Annex;
- .2 A renewal survey at intervals specified by the Administration, but not exceeding five years, except where regulation 9.2, 9.5, 9.6 or 9.7 of this Annex is applicable. The renewal survey shall be such as to ensure that the equipment, systems, fittings, arrangements and material fully comply with applicable requirements of chapter 3 of this Annex;
- .3 An intermediate survey within three months before or after the second anniversary date or within three months before or after the third anniversary date of the certificate which shall take the place of one of the annual surveys specified in paragraph 1.4 of this regulation. The intermediate survey shall be such as to ensure that the equipment and arrangements fully comply with the applicable requirements of chapter 3 of this Annex and are in good working

<sup>&</sup>lt;sup>5</sup> Refer to 2015 Guidelines for exhaust gas cleaning systems (resolution MEPC.259(68)).

order. Such intermediate surveys shall be endorsed on the IAPP Certificate issued under regulation 6 or 7 of this Annex;

- .4 An annual survey within three months before or after each anniversary date of the certificate, including a general inspection of the equipment, systems, fittings, arrangements and material referred to in paragraph 1.1 of this regulation to ensure that they have been maintained in accordance with paragraph 5 of this regulation and that they remain satisfactory for the service for which the ship is intended. Such annual surveys shall be endorsed on the IAPP Certificate issued under regulation 6 or 7 of this Annex; and
- .5 An additional survey either general or partial, according to the circumstances, shall be made whenever any important repairs or renewals are made as prescribed in paragraph 5 of this regulation or after a repair resulting from investigations prescribed in paragraph 6 of this regulation. The survey shall be such as to ensure that the necessary repairs or renewals have been effectively made, that the material and workmanship of such repairs or renewals are in all respects satisfactory and that the ship complies in all respects with the requirements of chapter 3 of this Annex.

2 In the case of ships of less than 400 gross tonnage, the Administration may establish appropriate measures in order to ensure that the applicable provisions of chapter 3 of this Annex are complied with.

3 Surveys of ships as regards the enforcement of the provisions of this Annex shall be carried out by officers of the Administration.

- .1 The Administration may, however, entrust the surveys either to surveyors nominated for the purpose or to organizations recognized by it. Such organizations shall comply with the guidelines adopted by the Organization;<sup>6</sup>
- .2 The survey of marine diesel engines and equipment for compliance with regulation 13 of this Annex shall be conducted in accordance with the revised NO<sub>X</sub> Technical Code 2008;
- .3 When a nominated surveyor or recognized organization determines that the condition of the equipment does not correspond substantially with the particulars of the certificate, it shall ensure that corrective action is taken and shall in due course notify the Administration. If such corrective action is not taken, the certificate shall be withdrawn by the Administration. If the ship is in a port of another Party, the appropriate authorities of the port State shall also be notified immediately. When an officer of the Administration, a nominated surveyor or recognized organization has notified the appropriate authorities of the port State, the Government of the port State concerned shall give such officer, surveyor or organization any necessary assistance to carry out their obligations under this regulation; and
- .4 In every case, the Administration concerned shall fully guarantee the completeness and efficiency of the survey and shall undertake to ensure the necessary arrangements to satisfy this obligation.

<sup>&</sup>lt;sup>6</sup> Refer to the Code for Recognized Organizations (RO Code), as adopted by the Organization by resolution MEPC.237(65), as may be amended by the Organization. Refer also to the Survey Guidelines under the Harmonized System of Survey and Certification (HSSC), 2019 (resolution A.1140(31)).

4 Ships to which chapter 4 of this Annex applies shall also be subject to the surveys specified below, taking into account the guidelines adopted by the Organization:<sup>7</sup>

- .1 An initial survey carried out before a new ship is put in service and before the International Energy Efficiency Certificate is issued. The survey shall verify that the ship's attained EEDI is in accordance with the requirements in chapter 4 of this Annex, and that the SEEMP required by regulation 26 of this Annex is on board;
- .2 A general or partial survey, according to the circumstances, carried out after a major conversion of a new ship to which this regulation applies. The survey shall ensure that the attained EEDI is recalculated as necessary and meets the requirement of regulation 24 of this Annex, with the reduction factor applicable to the ship type and size of the converted ship in the phase corresponding to the date of contract or keel laying or delivery determined for the original ship in accordance with regulation 2.2.18 of this Annex;
- .3 In cases where the major conversion of a new or existing ship is so extensive that the ship is regarded by the Administration as a newly constructed ship, the Administration shall determine the necessity of an initial survey on attained EEDI. Such a survey, if determined necessary, shall ensure that the attained EEDI is calculated and meets the requirement of regulation 24 of this Annex, with the reduction factor applicable corresponding to the ship type and size of the converted ship at the date of the contract of the conversion, or in the absence of a contract, the commencement date of the conversion. The survey shall also verify that the SEEMP required by regulation 26 of this Annex is on board and, for a ship to which regulation 27 applies, has been revised appropriately to reflect a major conversion in those cases where the major conversion affects data collection methodology and/or reporting processes;
- .4 For existing ships, the verification of the requirement to have a SEEMP on board according to regulation 26 of this Annex shall take place at the first intermediate or renewal survey identified in paragraph 1 of this regulation, whichever is the first, on or after 1 January 2013;
- .5 The Administration shall ensure that for each ship to which regulation 27 applies, the SEEMP complies with regulation 26.2 of this Annex. This shall be done prior to collecting data under regulation 27 of this Annex in order to ensure the methodology and processes are in place prior to the beginning of the ship's first reporting period. Confirmation of compliance shall be provided to and retained on board the ship;
- .6 The Administration shall ensure that, for each ship to which regulation 28 applies, the SEEMP complies with regulation 26.3.1 of this Annex. This shall be done prior to 1 January 2023. Confirmation of compliance shall be provided to, and retained on board, the ship;
- .7 The verification that the ship's attained EEXI is in accordance with the requirements in regulations 23 and 25 of this Annex shall take place at the first annual, intermediate or renewal survey identified in paragraph 1 of this

<sup>&</sup>lt;sup>7</sup> Refer to the 2014 Guidelines on survey and certification of the Energy Efficiency Design Index (resolution MEPC.254(67), as amended by resolutions MEPC.261(68) and MEPC.309(73)); consolidated text: MEPC.1/Circ.855/Rev.2, as may be further amended.

regulation or the initial survey identified in paragraphs 4.1 and 4.3 of this regulation, whichever is the first, on or after 1 January 2023; and

.8 Notwithstanding paragraph 4.7 of this regulation, a general or partial survey, according to the circumstances, carried out after a major conversion of a ship to which regulation 23 of this Annex applies. The survey shall ensure that the attained EEXI is recalculated as necessary and meets the requirement of regulation 25 of this Annex.

5 The equipment shall be maintained to conform with the provisions of this Annex and no changes shall be made in the equipment, systems, fittings, arrangements or material covered by the survey, without the express approval of the Administration. The direct replacement of such equipment and fittings with equipment and fittings that conform with the provisions of this Annex is permitted.

6 Whenever an accident occurs to a ship or a defect is discovered that substantially affects the efficiency or completeness of its equipment covered by this Annex, the master or owner of the ship shall report at the earliest opportunity to the Administration, a nominated surveyor or recognized organization responsible for issuing the relevant certificate.

# **Regulation 6**

Issue or endorsement of Certificates and Statements of Compliance related to fuel oil consumption reporting and operational carbon intensity rating

# International Air Pollution Prevention Certificate

1 An International Air Pollution Prevention (IAPP) Certificate shall be issued, after an initial or renewal survey in accordance with the provisions of regulation 5 of this Annex, to:

- .1 any ship of 400 gross tonnage and above engaged in voyages to ports or offshore terminals under the jurisdiction of other Parties; and
- .2 platforms and drilling rigs engaged in voyages to waters under the sovereignty or jurisdiction of other Parties.

A ship constructed before the date this Annex enters into force for that particular ship's Administration, shall be issued with an IAPP Certificate in accordance with paragraph 1 of this regulation no later than the first scheduled dry-docking after the date of such entry into force, but in no case later than three years after this date.

3 Such certificate shall be issued or endorsed either by the Administration or by any person or organization duly authorized by it.<sup>8</sup> In every case, the Administration assumes full responsibility for the certificate.

# International Energy Efficiency Certificate

4 An International Energy Efficiency Certificate for the ship shall be issued after a survey in accordance with the provisions of regulation 5.4 of this Annex to any ship of 400 gross tonnage and above before that ship may engage in voyages to ports or offshore terminals under the jurisdiction of other Parties.

<sup>&</sup>lt;sup>8</sup> Refer to the Code for Recognized Organizations (RO Code), as adopted by the Organization by resolution MEPC.237(65), as may be amended by the Organization.

5 The certificate shall be issued or endorsed either by the Administration or any organization duly authorized by it.8 In every case, the Administration assumes full responsibility for the certificate.

# Statement of Compliance related to fuel oil consumption reporting and operational carbon intensity rating

6 Upon receipt of reported data pursuant to regulation 27.3 of this Annex and attained annual operational CII pursuant to regulation 28.2 of this Annex, the Administration or any organization duly authorized by it shall:

- .1 determine whether the data has been reported in accordance with regulation 27 of this Annex;
- .2 verify that the attained annual operational CII reported is based on the data submitted in accordance with regulation 27 of this Annex;
- .3 based on the verified attained annual operational CII, determine the operational carbon intensity rating of the ship in accordance with regulation 28.6 of this Annex; and
- .4 issue a Statement of Compliance related to fuel oil consumption reporting and operational carbon intensity rating to the ship no later than five months from the beginning of the calendar year, upon determination and verification pursuant to regulations 6.6.1 to 6.6.3 of this Annex. In every case, the Administration assumes full responsibility for this Statement of Compliance.

7 Upon receipt of reported data pursuant to regulations 27.4, 27.5 or 27.6 of this Annex, the Administration or any organization duly authorized by it<sup>9</sup> shall promptly determine whether the data has been reported in accordance with regulation 27 and, if so, issue a Statement of Compliance to the ship. In every case, the Administration assumes full responsibility for this Statement of Compliance.

8 Notwithstanding paragraph 6 of this regulation, a ship rated as D for three consecutive years or rated as E in accordance with regulation 28 of this Annex shall not be issued a Statement of Compliance unless a plan of corrective actions is duly developed and reflected in the SEEMP and verified by the Administration or any organization duly authorized by it in accordance with regulations 28.7 and 28.8 of this Annex.

# **Regulation 7**

Issue of a Certificate by another Party

1 A Party may, at the request of the Administration, cause a ship to be surveyed and, if satisfied that the provisions of this Annex are complied with, shall issue or authorize the issue of an IAPP Certificate or an International Energy Efficiency Certificate to the ship, and where appropriate, endorse or authorize the endorsement of such certificates on the ship, in accordance with this Annex.

2 A copy of the certificate and a copy of the survey report shall be transmitted as soon as possible to the requesting Administration.

<sup>&</sup>lt;sup>9</sup> Refer to the Code for Recognized Organizations (RO Code), as adopted by the Organization by resolution MEPC.237(65), as may be amended by the Organization.

3 A certificate so issued shall contain a statement to the effect that it has been issued at the request of the Administration and it shall have the same force and receive the same recognition as a certificate issued under regulation 6 of this Annex.

4 No IAPP Certificate, International Energy Efficiency Certificate or UNSP Exemption Certificate shall be issued to a ship which is entitled to fly the flag of a State which is not a Party.

# **Regulation 8**

Form of Certificates and Statements of Compliance related to fuel oil consumption reporting and operational carbon intensity rating

### International Air Pollution Prevention Certificate

1 The IAPP Certificate shall be drawn up in a form corresponding to the model given in appendix I to this Annex and shall be at least in English, French or Spanish. If an official language of the issuing country is also used, this shall prevail in case of a dispute or discrepancy.

### International Energy Efficiency Certificate

2 The International Energy Efficiency Certificate shall be drawn up in a form corresponding to the model given in appendix VIII to this Annex and shall be at least in English, French or Spanish. If an official language of the issuing Party is also used, this shall prevail in case of a dispute or discrepancy.

# Statement of Compliance related to fuel oil consumption reporting and operational carbon intensity rating

3 The Statement of Compliance pursuant to regulations 6.6 and 6.7 of this Annex shall be drawn up in a form corresponding to the model given in appendix X to this Annex and shall be at least in English, French or Spanish. If an official language of the issuing Party is also used, this shall prevail in case of a dispute or discrepancy.

### International Air Pollution Prevention Exemption Certificate for Unmanned Non-selfpropelled Barges

4 In accordance with regulation 3.4 of this Annex, the International Air Pollution Prevention Exemption Certificate for Unmanned Non-self-propelled Barges shall be drawn up in the form corresponding to the model given in appendix XI to this Annex and shall be at least in English, French or Spanish. If an official language of the issuing country is also used, this shall prevail in the event of a dispute or discrepancy.

### **Regulation 9**

Duration and validity of Certificates and Statements of Compliance related to fuel oil consumption reporting and operational carbon intensity rating

### International Air Pollution Prevention Certificate

1 An IAPP Certificate shall be issued for a period specified by the Administration, which shall not exceed five years.

- 2 Notwithstanding the requirements of paragraph 1 of this regulation:
  - .1 when the renewal survey is completed within three months before the expiry date of the existing certificate, the new certificate shall be valid from the date

of completion of the renewal survey to a date not exceeding five years from the date of expiry of the existing certificate;

- .2 when the renewal survey is completed after the expiry date of the existing certificate, the new certificate shall be valid from the date of completion of the renewal survey to a date not exceeding five years from the date of expiry of the existing certificate; and
- .3 when the renewal survey is completed more than three months before the expiry date of the existing certificate, the new certificate shall be valid from the date of completion of the renewal survey to a date not exceeding five years from the date of completion of the renewal survey.

3 If a certificate is issued for a period of less than five years, the Administration may extend the validity of the certificate beyond the expiry date to the maximum period specified in paragraph 1 of this regulation, provided that the surveys referred to in regulations 5.1.3 and 5.1.4 of this Annex applicable when a certificate is issued for a period of five years are carried out as appropriate.

4 If a renewal survey has been completed and a new certificate cannot be issued or placed on board the ship before the expiry date of the existing certificate, the person or organization authorized by the Administration may endorse the existing certificate and such a certificate shall be accepted as valid for a further period that shall not exceed five months from the expiry date.

If a ship, at the time when a certificate expires, is not in a port in which it is to be surveyed, the Administration may extend the period of validity of the certificate, but this extension shall be granted only for the purpose of allowing the ship to complete its voyage to the port in which it is to be surveyed, and then only in cases where it appears proper and reasonable to do so. No certificate shall be extended for a period longer than three months, and a ship to which an extension is granted shall not, on its arrival in the port in which it is to be surveyed, be entitled by virtue of such extension to leave that port without having a new certificate. When the renewal survey is completed, the new certificate shall be valid to a date not exceeding five years from the date of expiry of the existing certificate before the extension was granted.

6 A certificate issued to a ship engaged on short voyages that has not been extended under the foregoing provisions of this regulation may be extended by the Administration for a period of grace of up to one month from the date of expiry stated on it. When the renewal survey is completed, the new certificate shall be valid to a date not exceeding five years from the date of expiry of the existing certificate before the extension was granted.

7 In special circumstances, as determined by the Administration, a new certificate need not be dated from the date of expiry of the existing certificate as required by paragraph 2.1, 5 or 6 of this regulation. In these special circumstances, the new certificate shall be valid to a date not exceeding five years from the date of completion of the renewal survey.

8 If an annual or intermediate survey is completed before the period specified in regulation 5 of this Annex, then:

- .1 the anniversary date shown on the certificate shall be amended by endorsement to a date that shall not be more than three months later than the date on which the survey was completed;
- .2 the subsequent annual or intermediate survey required by regulation 5 of this Annex shall be completed at the intervals prescribed by that regulation using the new anniversary date; and

.3 the expiry date may remain unchanged, provided one or more annual or intermediate surveys, as appropriate, are carried out so that the maximum intervals between the surveys prescribed by regulation 5 of this Annex are not exceeded.

9 A certificate issued under regulation 6 or 7 of this Annex shall cease to be valid in any of the following cases:

- .1 if the relevant surveys are not completed within the periods specified under regulation 5.1 of this Annex;
- .2 if the certificate is not endorsed in accordance with regulation 5.1.3 or 5.1.4 of this Annex; and
- .3 upon transfer of the ship to the flag of another State. A new certificate shall only be issued when the Government issuing the new certificate is fully satisfied that the ship is in compliance with the requirements of regulation 5.4 of this Annex. In the case of a transfer between Parties, if requested within three months after the transfer has taken place, the Government of the Party whose flag the ship was formerly entitled to fly shall, as soon as possible, transmit to the Administration copies of the certificate carried by the ship before the transfer and, if available, copies of the relevant survey reports.

# International Energy Efficiency Certificate

10 The International Energy Efficiency Certificate shall be valid throughout the life of the ship subject to the provisions of paragraph 11 below.

11 An International Energy Efficiency Certificate issued under this Annex shall cease to be valid in any of the following cases:

- .1 if the ship is withdrawn from service or if a new certificate is issued following major conversion of the ship; or
- .2 upon transfer of the ship to the flag of another State. A new certificate shall only be issued when the Government issuing the new certificate is fully satisfied that the ship is in compliance with the requirements of chapter 4 of this Annex. In the case of a transfer between Parties, if requested within three months after the transfer has taken place, the Government of the Party whose flag the ship was formerly entitled to fly shall, as soon as possible, transmit to the Administration copies of the certificate carried by the ship before the transfer and, if available, copies of the relevant survey reports; or
- .3 if the ship's equipment, systems, fittings, arrangements, or material covered by the survey were changed without the express approval of the Administration, as provided for in regulation 5.5 of this Annex, unless regulation 3 of this Annex applies.

# Statement of Compliance related to fuel oil consumption reporting and operational carbon intensity rating

12 The Statement of Compliance issued pursuant to regulation 6.6 of this Annex shall be valid for the calendar year in which it is issued and for the first five months of the following

calendar year. The Statement of Compliance issued pursuant to regulation 6.7 of this Annex shall be valid for the calendar year in which it is issued, for the following calendar year, and for the first five months of the subsequent calendar year. All Statements of Compliance shall be kept on board for at least five years.

# **Regulation 10**

Port State control on operational requirements

1 A ship, when in a port or an offshore terminal under the jurisdiction of another Party, is subject to inspection by officers duly authorized by such Party concerning operational requirements under this Annex,<sup>10</sup> where there are clear grounds for believing that the master or crew are not familiar with essential shipboard procedures relating to the prevention of air pollution from ships.

2 In the circumstances given in paragraph 1 of this regulation, the Party shall take steps to ensure that the ship shall not sail until the situation has been brought to order in accordance with the requirements of this Annex.

3 Procedures relating to the port State control prescribed in article 5 of the present Convention shall apply to this regulation.

4 Nothing in this regulation shall be construed to limit the rights and obligations of a Party carrying out control over operational requirements specifically provided for in the present Convention.

5 In relation to chapter 4 of this Annex, any port State inspection may verify, when appropriate, that there is a valid Statement of Compliance related to fuel oil consumption reporting and operational carbon intensity rating, an International Energy Efficiency Certificate and a Ship Energy Efficiency Management Plan on board, in accordance with article 5 of the present Convention.

6 Notwithstanding the requirements in paragraph 5 of this regulation, any port State inspection may inspect whether the Ship Energy Efficiency Management Plan is duly implemented by the ship in accordance with regulation 28 of this Annex.

### **Regulation 11**

Detection of violations and enforcement

1 Parties shall cooperate in the detection of violations and the enforcement of the provisions of this Annex, using all appropriate and practicable measures of detection and environmental monitoring, and adequate procedures for reporting and accumulation of evidence.

2 A ship to which this Annex applies may, in any port or offshore terminal of a Party, be subject to inspection by officers appointed or authorized by that Party for the purpose of verifying whether the ship has emitted any of the substances covered by this Annex in violation of the provision of this Annex. If an inspection indicates a violation of this Annex, a report shall be forwarded to the Administration for any appropriate action.

3 Any Party shall furnish to the Administration evidence, if any, that the ship has emitted any of the substances covered by this Annex in violation of the provisions of this Annex. If it is

<sup>&</sup>lt;sup>10</sup> Refer to the *Procedures for port State control, 2019* (resolution A.1138(31)). Refer also to the *2019 Guidelines for port State control under MARPOL Annex VI Chapter 3* (resolution MEPC.321(74)).

practicable to do so, the competent authority of the former Party shall notify the master of the ship of the alleged violation.

4 Upon receiving such evidence, the Administration shall investigate the matter and may request the other Party to furnish further or better evidence of the alleged contravention. If the Administration is satisfied that sufficient evidence is available to enable proceedings to be brought in respect of the alleged violation, it shall cause such proceedings to be taken in accordance with its law as soon as possible. The Administration shall promptly inform the Party that has reported the alleged violation, as well as the Organization, of the action taken.

5 A Party may also inspect a ship to which this Annex applies when it enters the ports or offshore terminals under its jurisdiction, if a request for an investigation is received from any Party together with sufficient evidence that the ship has emitted any of the substances covered by the Annex in any place in violation of this Annex. The report of such investigation shall be sent to the Party requesting it and to the Administration so that the appropriate action may be taken under the present Convention.

6 The international law concerning the prevention, reduction and control of pollution of the marine environment from ships, including that law relating to enforcement and safeguards, in force at the time of application or interpretation of this Annex, applies, mutatis mutandis, to the rules and standards set forth in this Annex.

# Chapter 3 – Requirements for control of emissions from ships

# Regulation 12

Ozone-depleting substances

1 This regulation does not apply to permanently sealed equipment where there are no refrigerant charging connections or potentially removable components containing ozone-depleting substances.

2 Subject to the provisions of regulation 3.1, any deliberate emissions of ozone-depleting substances shall be prohibited. Deliberate emissions include emissions occurring in the course of maintaining, servicing, repairing or disposing of systems or equipment, except that deliberate emissions do not include minimal releases associated with the recapture or recycling of an ozone-depleting substance. Emissions arising from leaks of an ozone-depleting substance, whether or not the leaks are deliberate, may be regulated by Parties.

3.1 Installations that contain ozone-depleting substances, other than hydrochlorofluorocarbons, shall be prohibited:

- .1 on ships constructed on or after 19 May 2005; or
- .2 in the case of ships constructed before 19 May 2005 which have a contractual delivery date of the equipment to the ship on or after 19 May 2005 or, in the absence of a contractual delivery date, the actual delivery of the equipment to the ship on or after 19 May 2005.
- 3.2 Installations that contain hydrochlorofluorocarbons shall be prohibited:
  - .1 on ships constructed on or after 1 January 2020; or
  - .2 in the case of ships constructed before 1 January 2020 which have a contractual delivery date of the equipment to the ship on or

after 1 January 2020 or, in the absence of a contractual delivery date, the actual delivery of the equipment to the ship on or after 1 January 2020.

4 The substances referred to in this regulation, and equipment containing such substances, shall be delivered to appropriate reception facilities when removed from ships.

5 Each ship subject to regulation 6.1 shall maintain a list of equipment containing ozone-depleting substances.<sup>11</sup>

6 Each ship subject to regulation 6.1 that has rechargeable systems that contain ozone-depleting substances shall maintain an ozone-depleting substances record book. This record book may form part of an existing logbook or electronic record book<sup>12</sup> as approved by the Administration. An electronic recording system referred to in regulation 12.6, as adopted by resolution MEPC.176(58), shall be considered an electronic record book, provided the electronic recording system is approved by the Administration on or before the first IAPP Certificate renewal survey carried out on or after 1 October 2020, but not later than 1 October 2025, taking into account the guidelines developed by the Organization.12

7 Entries in the ozone-depleting substances record book shall be recorded in terms of mass (kg) of substance and shall be completed without delay on each occasion, in respect of the following:

- .1 recharge, full or partial, of equipment containing ozone-depleting substances;
- .2 repair or maintenance of equipment containing ozone-depleting substances;
- .3 discharge of ozone-depleting substances to the atmosphere:
  - .3.1 deliberate; and
  - .3.2 non-deliberate;
- .4 discharge of ozone-depleting substances to land-based reception facilities; and
- .5 supply of ozone-depleting substances to the ship.

# **Regulation 13**

Nitrogen oxides ( $NO_X$ )

### Application

- 1.1 This regulation shall apply to:
  - .1 each marine diesel engine with a power output of more than 130 kW installed on a ship; and
  - .2 each marine diesel engine with a power output of more than 130 kW that undergoes a major conversion on or after 1 January 2000 except when demonstrated to the satisfaction of the Administration that such engine is an

<sup>&</sup>lt;sup>11</sup> See appendix I, Supplement to International Air Pollution Prevention Certificate (IAPP Certificate), section 2.1.

<sup>&</sup>lt;sup>12</sup> Refer to the *Guidelines for the use of electronic record books under MARPOL* (resolution MEPC.312(74)).

identical replacement to the engine that it is replacing and is otherwise not covered under paragraph 1.1.1 of this regulation.

- 1.2 This regulation does not apply to:
  - .1 a marine diesel engine intended to be used solely for emergencies or solely to power any device or equipment intended to be used solely for emergencies on the ship on which it is installed, or a marine diesel engine installed in lifeboats intended to be used solely for emergencies; and
  - .2 a marine diesel engine installed on a ship solely engaged in voyages within waters subject to the sovereignty or jurisdiction of the State the flag of which the ship is entitled to fly, provided that such engine is subject to an alternative NO<sub>X</sub> control measure established by the Administration.

1.3 Notwithstanding the provisions of paragraph 1.1 of this regulation, the Administration may provide an exclusion from the application of this regulation for any marine diesel engine that is installed on a ship constructed, or for any marine diesel engine that undergoes a major conversion, before 19 May 2005, provided that the ship on which the engine is installed is solely engaged in voyages to ports or offshore terminals within the State the flag of which the ship is entitled to fly.

### Major conversion

2.1 For the purpose of this regulation, *major conversion* means a modification on or after 1 January 2000 of a marine diesel engine that has not already been certified to the standards set forth in paragraph 3, 4, or 5.1.1 of this regulation where:

- .1 the engine is replaced by a marine diesel engine or an additional marine diesel engine is installed, or
- .2 any substantial modification, as defined in the revised NO<sub>X</sub> Technical Code 2008, is made to the engine, or
- .3 the maximum continuous rating of the engine is increased by more than 10% compared to the maximum continuous rating of the original certification of the engine.

2.2 For a major conversion involving the replacement of a marine diesel engine with a nonidentical marine diesel engine, or the installation of an additional marine diesel engine, the standards in this regulation at the time of the replacement or addition of the engine shall apply. In the case of replacement engines only, if it is not possible for such a replacement engine to meet the standards set forth in paragraph 5.1.1 of this regulation (Tier III, as applicable), then that replacement engine shall meet the standards set forth in paragraph 4 of this regulation (Tier II), taking into account the guidelines developed by the Organization.<sup>13</sup>

2.3 A marine diesel engine referred to in paragraph 2.1.2 or 2.1.3 of this regulation shall meet the following standards:

.1 for ships constructed prior to 1 January 2000, the standards set forth in paragraph 3 of this regulation shall apply; and

<sup>&</sup>lt;sup>13</sup> Refer to the 2013 Guidelines as required by regulation 13.2.2 of MARPOL Annex VI in respect of non-identical replacement engines not required to meet the Tier III limit (resolution MEPC.230(65))

.2 for ships constructed on or after 1 January 2000, the standards in force at the time the ship was constructed shall apply.

# Tier I<sup>14</sup>

3 Subject to regulation 3 of this Annex, the operation of a marine diesel engine that is installed on a ship constructed on or after 1 January 2000 and prior to 1 January 2011 is prohibited, except when the emission of nitrogen oxides (calculated as the total weighted emission of NO<sub>2</sub>) from the engine is within the following limits, where n = rated engine speed (crankshaft revolutions per minute):

- .1 17.0 g/kWh when n is less than 130 rpm;
- .2  $45 \cdot n^{(-0.2)}$  g/kWh when *n* is 130 or more but less than 2,000 rpm;
- .3 9.8 g/kWh when *n* is 2,000 rpm or more.

### Tier II

Subject to regulation 3 of this Annex, the operation of a marine diesel engine that is installed on a ship constructed on or after 1 January 2011 is prohibited, except when the emission of nitrogen oxides (calculated as the total weighted emission of  $NO_2$ ) from the engine is within the following limits, where *n* = rated engine speed (crankshaft revolutions per minute):

- .1 14.4 g/kWh when n is less than 130 rpm;
- .2  $44 \cdot n^{(-0.2)}$  g/kWh when *n* is 130 or more but less than 2,000 rpm;
- .3 7.7 g/kWh when n is 2,000 rpm or more.

### Tier III

5.1 Subject to regulation 3 of this Annex, in an emission control area designated for Tier III  $NO_X$  control under paragraph 6 of this regulation ( $NO_X$  Tier III emission control area), the operation of a marine diesel engine that is installed on a ship is prohibited:

- .1 except when the emission of nitrogen oxides (calculated as the total weighted emission of  $NO_2$ ) from the engine is within the following limits, where n = rated engine speed (crankshaft revolutions per minute):
  - .1 3.4 g/kWh when *n* is less than 130 rpm;
  - .2  $9 \cdot n^{(-0.2)}$  g/kWh when *n* is 130 or more but less than 2,000 rpm;
  - .3 2.0 g/kWh when *n* is 2,000 rpm or more;

when

.2 that ship is constructed on or after:

<sup>&</sup>lt;sup>14</sup> Refer to the *Guidelines* for the application of the NO<sub>x</sub> Technical Code relative to certification and amendments of Tier I engines (MEPC.1/Circ.679).

- .1 1 January 2016 and is operating in the North American Emission Control Area or the United States Caribbean Sea Emission Control Area;
- .2 1 January 2021 and is operating in the Baltic Sea Emission Control Area or the North Sea Emission Control Area;
- .3 that ship is operating in a  $NO_X$  Tier III emission control area other than an emission control area described in paragraph 5.1.2 of this regulation, and is constructed on or after the date of adoption of such an emission control area, or a later date as may be specified in the amendment designating the  $NO_X$  Tier III emission control area, whichever is later.
- 5.2 The standards set forth in paragraph 5.1.1 of this regulation shall not apply to:
  - .1 a marine diesel engine installed on a ship with a length (*L*), as defined in regulation 1.19 of Annex I to the present Convention, of less than 24 metres when it has been specifically designed, and is used solely, for recreational purposes; or
  - .2 a marine diesel engine installed on a ship with a combined nameplate diesel engine propulsion power of less than 750 kW if it is demonstrated, to the satisfaction of the Administration, that the ship cannot comply with the standards set forth in paragraph 5.1.1 of this regulation because of design or construction limitations of the ship; or
  - .3 a marine diesel engine installed on a ship constructed prior to 1 January 2021 of less than 500 gross tonnage, with a length (*L*), as defined in regulation 1.19 of Annex I to the present Convention, of 24 metres or over when it has been specifically designed, and is used solely, for recreational purposes.

5.3 The tier and on/off status of marine diesel engines installed on board a ship to which paragraph 5.1 of this regulation applies which are certified to both Tier II and Tier III or which are certified to Tier II only shall be recorded in such logbook or electronic record book<sup>15</sup> as prescribed by the Administration at entry into and exit from a NO<sub>X</sub> Tier III emission control area, or when the on/off status changes within such an area, together with the date, time and position of the ship.

5.4 Emissions of nitrogen oxides from a marine diesel engine subject to paragraph 5.1 of this regulation that occur immediately following building and sea trials of a newly constructed ship, or before and following converting, repairing, and/or maintaining the ship, or maintenance or repair of a Tier II engine or a dual fuel engine when the ship is required to not have gas fuel or gas cargo on board due to safety requirements, for which activities take place in a shipyard or other repair facility located in a  $NO_X$  Tier III emission control area are temporarily exempted provided the following conditions are met:

- .1 the engine meets the Tier II NO<sub>X</sub> limits; and
- .2 the ship sails directly to or from the shipyard or other repair facility, does not load or unload cargo during the duration of the exemption, and follows any

<sup>&</sup>lt;sup>15</sup> Refer to the *Guidelines for the use of electronic record books under MARPOL* (resolution MEPC.312(74))

additional specific routeing requirements indicated by the port State in which the shipyard or other repair facility is located, if applicable.

5.5 The exemption described in paragraph 5.4 of this regulation applies only for the following period:

- .1 for a newly constructed ship, the period beginning at the time the ship is delivered from the shipyard, including sea trials, and ending at the time the ship directly exits the  $NO_X$  Tier III emission control area(s) or, with regard to a ship fitted with a dual fuel engine, the ship directly exits the  $NO_X$  Tier III emission control area(s) or proceeds directly to the nearest gas fuel bunkering facility appropriate to the ship located in the  $NO_X$  Tier III emission control area(s);
- .2 for a ship with a Tier II engine undergoing conversion, maintenance or repair, the period beginning at the time the ship enters the  $NO_X$  Tier III emission control area(s) and proceeds directly to the shipyard or other repair facility, and ending at the time the ship is released from the shipyard or other repair facility and directly exits the  $NO_X$  Tier III emission control area (s) after performing sea trials, if applicable; or
- .3 for a ship with a dual fuel engine undergoing conversion, maintenance or repair, when the ship is required to not have gas fuel or gas cargo on board due to safety requirements, the period beginning at the time the ship enters the NO<sub>X</sub> Tier III emission control area(s) or when it is degassed in the NO<sub>X</sub> Tier III emission control area(s) and proceeds directly to the shipyard or other repair facility, and ending at the time when the ship is released from the shipyard or other repair facility and directly exits the NO<sub>X</sub> Tier III emission control area(s) and proceeds directly to the shipyard or other shipyard or other repair facility and directly exits the NO<sub>X</sub> Tier III emission control area(s) or proceeds directly to the nearest gas fuel bunkering facility appropriate to the ship located in the NO<sub>X</sub> Tier III emission control area(s).

# **Emission control area**

For the purposes of this regulation, a  $NO_X$  Tier III emission control area shall be any sea area, including any port area, designated by the Organization in accordance with the criteria and procedures set forth in appendix III to this Annex. The  $NO_X$  Tier III emission control areas are:

- .1 the North American Emission Control Area, which means the area described by the coordinates provided in appendix VII to this Annex;
- .2 the United States Caribbean Sea Emission Control Area, which means the area described by the coordinates provided in appendix VII to this Annex;
- .3 the Baltic Sea area as defined in regulation 1.11.2 of Annex I of the present Convention; and
- .4 the North Sea area as defined in regulation 1.14.6 of Annex V of the present Convention.

### Marine diesel engines installed on a ship constructed prior to 1 January 2000

7.1 Notwithstanding paragraph 1.1.1 of this regulation, a marine diesel engine with a power output of more than 5,000 kW and a per cylinder displacement at or above 90 L installed on a ship constructed on or after 1 January 1990 but prior to 1 January 2000 shall comply with the emission limits set forth in paragraph 7.4 of this regulation, provided that an approved method<sup>16</sup> for that engine has been certified by an Administration of a Party and notification of such certification has been submitted to the Organization by the certifying Administration.<sup>17</sup> Compliance with this paragraph shall be demonstrated through one of the following:

- .1 installation of the certified approved method, as confirmed by a survey using the verification procedure specified in the approved method file, including appropriate notation on the ship's IAPP Certificate of the presence of the approved method; or
- .2 certification of the engine confirming that it operates within the limits set forth in paragraph 3, 4, or 5.1.1 of this regulation and an appropriate notation of the engine certification on the ship's IAPP Certificate.

7.2 Paragraph 7.1 of this regulation shall apply no later than the first renewal survey that occurs 12 months or more after deposit of the notification in paragraph 7.1. If a shipowner of a ship on which an approved method is to be installed can demonstrate to the satisfaction of the Administration that the approved method was not commercially available despite best efforts to obtain it, then that approved method shall be installed on the ship no later than the next annual survey of that ship that falls after the approved method is commercially available.

7.3 With regard to a marine diesel engine with a power output of more than 5,000 kW and a per cylinder displacement at or above 90 L installed on a ship constructed on or after 1 January 1990, but prior to 1 January 2000, the IAPP Certificate shall, for a marine diesel engine to which paragraph 7.1 of this regulation applies, indicate one of the following:

- .1 an approved method has been applied pursuant to paragraph 7.1.1 of this regulation;
- .2 the engine has been certified pursuant to paragraph 7.1.2 of this regulation;
- .3 an approved method is not yet commercially available as described in paragraph 7.2 of this regulation; or
- .4 an approved method is not applicable.

7.4 Subject to regulation 3 of this Annex, the operation of a marine diesel engine described in paragraph 7.1 of this regulation is prohibited, except when the emission of nitrogen oxides (calculated as the total weighted emission of NO<sub>2</sub>) from the engine is within the following limits, where n = rated engine speed (crankshaft revolutions per minute):

- .1 17.0 g/kWh when n is less than 130 rpm;
- .2  $45 \cdot n^{(-0.2)}$  g/kWh when *n* is 130 or more but less than 2,000 rpm; and

<sup>&</sup>lt;sup>16</sup> Refer to the 2014 Guidelines on the approved method process (resolution MEPC.243(66)).

<sup>&</sup>lt;sup>17</sup> Refer to the 2014 Guidelines in respect of the information to be submitted by an Administration to the Organization covering the certification of an approved method as required under regulation 13.7.1 of MARPOL Annex VI (resolution MEPC.242(66)).

.3 9.8 g/kWh when *n* is 2,000 rpm or more.

7.5 Certification of an approved method shall be in accordance with chapter 7 of the revised NO<sub>X</sub> Technical Code 2008 and shall include verification:

- .1 by the designer of the base marine diesel engine to which the approved method applies that the calculated effect of the approved method will not decrease engine rating by more than 1.0%, increase fuel consumption by more than 2.0% as measured according to the appropriate test cycle set forth in the revised NO<sub>X</sub> Technical Code 2008, or adversely affect engine durability or reliability; and
- .2 that the cost of the approved method is not excessive, which is determined by a comparison of the amount of  $NO_X$  reduced by the approved method to achieve the standard set forth in paragraph 7.4 of this regulation and the cost of purchasing and installing such approved method.<sup>18</sup>

# Certification

8 The revised NO<sub>X</sub> Technical Code 2008 shall be applied in the certification, testing and measurement procedures for the standards set forth in this regulation.

9 The procedures for determining  $NO_X$  emissions set out in the revised  $NO_X$  Technical Code 2008 are intended to be representative of the normal operation of the engine. Defeat devices and irrational emission control strategies undermine this intention and shall not be allowed. This regulation shall not prevent the use of auxiliary control devices that are used to protect the engine and/or its ancillary equipment against operating conditions that could result in damage or failure or that are used to facilitate the starting of the engine.

### **Regulation 14**

Sulphur oxides  $(SO_X)$  and particulate matter

### General requirements

1 The sulphur content of fuel oil used or carried for use on board a ship shall not exceed 0.50% m/m.

2 The worldwide average sulphur content of residual fuel oil supplied for use on board ships shall be monitored taking into account the guidelines developed by the Organization.<sup>19</sup>

 $Ce = \frac{Cost \ of \ approved \ method \ \cdot \ 10^{6}}{Power \ (KW) \ \cdot \ 0.768 \cdot \ 6,000 \ (hours/year) \cdot \ 5 \ (years) \ \cdot \ \Delta NOx \ (g/kWh)}$ 

Refer to the Definitions for the cost-effectiveness formula in regulation 13.7.5 of the revised MARPOL Annex VI (MEPC.1/Circ.678).

<sup>19</sup> Refer to the 2020 Guidelines for monitoring the worldwide average sulphur content of fuel oils supplied for use on board ships (resolution MEPC.326(75)).

<sup>&</sup>lt;sup>18</sup> The cost of an approved method shall not exceed 375 Special Drawing Rights/metric tonne NO<sub>X</sub> calculated in accordance with the cost-effectiveness (Ce) formula below:

### Requirements within emission control areas

3 For the purpose of this regulation, an emission control area shall be any sea area, including any port area, designated by the Organization in accordance with the criteria and procedures set forth in appendix III to this Annex. The emission control areas under this regulation are:

- .1 the Baltic Sea area as defined in regulation 1.11.2 of Annex I of the present Convention;
- .2 the North Sea area as defined in regulation 1.14.6 of Annex V of the present Convention;
- .3 the North American Emission Control Area, which means the area described by the coordinates provided in appendix VII to this Annex; and
- .4 the United States Caribbean Sea Emission Control Area, which means the area described by the coordinates provided in appendix VII to this Annex.

4 While a ship is operating within an emission control area, the sulphur content of fuel oil used on board that ship shall not exceed 0.10% m/m.

5 The sulphur content of fuel oil referred to in paragraph 1 and paragraph 4 of this regulation shall be documented by its supplier as required by regulation 18 of this Annex.

6 Those ships using separate fuel oils to comply with paragraph 4 of this regulation and entering or leaving an emission control area set forth in paragraph 3 of this regulation shall carry a written procedure showing how the fuel oil changeover is to be done, allowing sufficient time for the fuel oil service system to be fully flushed of all fuel oils exceeding the applicable sulphur content specified in paragraph 4 of this regulation prior to entry into an emission control area. The volume of low sulphur fuel oils in each tank as well as the date, time and position of the ship when any fuel oil changeover operation is completed prior to the entry into an emission control area or commenced after exit from such an area shall be recorded in such logbook or electronic record book<sup>20</sup> as prescribed by the Administration.

7 During the first 12 months immediately following entry into force of an amendment designating a specific emission control area under paragraph 3 of this regulation, ships operating in that emission control area are exempt from the requirements in paragraphs 4 and 6 of this regulation and from the requirements of paragraph 5 of this regulation insofar as they relate to paragraph 4 of this regulation.

### In-use and onboard fuel oil sampling and testing

8 If the competent authority of a Party requires the in-use or onboard sample to be analysed, it shall be done in accordance with the verification procedure set forth in appendix VI to this Annex to determine whether the fuel oil being used or carried for use on board meets the requirements in paragraph 1 or paragraph 4 of this regulation. The in-use sample shall be

<sup>&</sup>lt;sup>20</sup> Refer to the *Guidelines for the use of electronic record books under MARPOL* (resolution MEPC.312(74))

drawn taking into account the guidelines developed by the Organization.<sup>21</sup> The onboard sample shall be drawn taking into account the guidelines developed by the Organization.<sup>22</sup>

9 The sample shall be sealed by the representative of the competent authority with a unique means of identification installed in the presence of the ship's representative. The ship shall be given the option of retaining a duplicate sample.

# In-use fuel oil sampling point

10 For each ship subject to regulations 5 and 6 of this Annex, sampling point(s) shall be fitted or designated for the purpose of taking representative samples of the fuel oil being used on board the ship taking into account the guidelines developed by the Organization.<sup>23</sup>

11 For a ship constructed before 1 April 2022, the sampling point(s) referred to in paragraph 10 shall be fitted or designated not later than the first renewal survey as identified in regulation 5.1.2 of this Annex on or after 1 April 2023.

12 The requirements of paragraphs 10 and 11 above are not applicable to a fuel oil service system for a low-flashpoint fuel for combustion purposes for propulsion or operation on board the ship.

13 The competent authority of a Party shall, as appropriate, utilize the sampling point(s) which is(are) fitted or designated for the purpose of taking representative sample(s) of the fuel oil being used on board in order to verify that the fuel oil complies with this regulation. Taking fuel oil samples by the competent authority of the Party shall be performed as expeditiously as possible without causing the ship to be unduly delayed.

# **Regulation 15**

Volatile organic compounds

1 If the emissions of volatile organic compounds (VOCs) from a tanker are to be regulated in a port or ports or a terminal or terminals under the jurisdiction of a Party, they shall be regulated in accordance with the provisions of this regulation.

2 A Party regulating tankers for VOC emissions shall submit a notification to the Organization.<sup>24</sup> This notification shall include information on the size of tankers to be controlled, the cargoes requiring vapour emission control systems and the effective date of such control. The notification shall be submitted at least six months before the effective date.

3 A Party that designates ports or terminals at which VOC emissions from tankers are to be regulated shall ensure that vapour emission control systems, approved by that Party taking into account the safety standards for such systems developed by the Organization,<sup>25</sup> are provided in any designated port and terminal and are operated safely and in a manner so as to avoid undue delay to a ship.

<sup>&</sup>lt;sup>21</sup> Refer to the 2019 Guidelines for on board sampling for the verification of the sulphur content of the fuel oil used on board ships (MEPC.1/Circ.864/Rev.1).

<sup>&</sup>lt;sup>22</sup> Refer to the 2020 Guidelines for on board sampling of fuel oil intended to be used or carried for use on board a ship (MEPC.1/Circ.889).

<sup>&</sup>lt;sup>23</sup> Refer to the 2019 Guidelines for on board sampling for the verification of the sulphur content of the fuel oil used on board ships (MEPC.1/Circ.864/Rev.1).

<sup>&</sup>lt;sup>24</sup> Refer to the Notification to the Organization on ports or terminals where volatile organic compounds (VOCs) emissions are to be regulated (MEPC.1/Circ.509).

<sup>&</sup>lt;sup>25</sup> Refer to the *Standards for vapour emission control systems* (MSC/Circ.585).

4 The Organization shall circulate a list of the ports and terminals designated by Parties to other Parties and Member States of the Organization for their information.

5 A tanker to which paragraph 1 of this regulation applies shall be provided with a vapour emission collection system approved by the Administration taking into account the safety standards for such systems developed by the Organization,<sup>25</sup> and shall use this system during the loading of relevant cargoes. A port or terminal that has installed vapour emission control systems in accordance with this regulation may accept tankers that are not fitted with vapour collection systems for a period of three years after the effective date identified in paragraph 2 of this regulation.

6 A tanker carrying crude oil shall have on board and implement a VOC management plan approved by the Administration.<sup>26</sup> Such a plan shall be prepared taking into account the guidelines developed by the Organization. The plan shall be specific to each ship and shall at least:

- .1 provide written procedures for minimizing VOC emissions during the loading, sea passage and discharge of cargo;
- .2 give consideration to the additional VOC generated by crude oil washing;
- .3 identify a person responsible for implementing the plan; and
- .4 for ships on international voyages, be written in the working language of the master and officers and, if the working language of the master and officers is not English, French or Spanish, include a translation into one of these languages.

7 This regulation shall also apply to gas carriers only if the types of loading and containment systems allow safe retention of non-methane VOCs on board or their safe return ashore.<sup>27</sup>

### **Regulation 16**

### Shipboard incineration

1 Except as provided in paragraph 4 of this regulation, shipboard incineration shall be allowed only in a shipboard incinerator.

- 2 Shipboard incineration of the following substances shall be prohibited:
  - .1 residues of cargoes subject to Annex I, II or III or related contaminated packing materials;
  - .2 polychlorinated biphenyls (PCBs);
  - .3 garbage, as defined by Annex V, containing more than traces of heavy metals;

<sup>&</sup>lt;sup>26</sup> Refer to the Guidelines for the development of a VOC management plan (resolution MEPC.185(59)). Refer also to the Technical information on systems and operation to assist development of VOC management plans (MEPC.1/Circ.680), and the Technical information on a vapour pressure control system in order to facilitate the development and the update of VOC management plans (MEPC.1/Circ.719).

<sup>27</sup> Refer to the International Code for the Construction and Equipment of Ships Carrying Liquefied Gases in Bulk.

- .4 refined petroleum products containing halogen compounds;
- .5 sewage sludge and sludge oil neither of which is generated on board the ship; and
- .6 exhaust gas cleaning system residues.

3 Shipboard incineration of polyvinyl chlorides (PVCs) shall be prohibited, except in shipboard incinerators for which IMO Type Approval Certificates<sup>28</sup> have been issued.

4 Shipboard incineration of sewage sludge and sludge oil generated during normal operation of a ship may also take place in the main or auxiliary power plant or boilers, but in those cases, shall not take place inside ports, harbours or estuaries.

- 5 Nothing in this regulation either:
  - .1 affects the incineration at sea prohibitions of the Convention on the Prevention of Marine Pollution by Dumping of Wastes and Other Matter, 1972, as amended, and the 1996 Protocol thereto, or other requirements thereof,
  - or
  - .2 precludes the development, installation and operation of alternative design shipboard thermal waste treatment devices that meet or exceed the requirements of this regulation.

6.1 Except as provided in paragraph 6.2 of this regulation, each incinerator on a ship constructed on or after 1 January 2000 or incinerator that is installed on board a ship on or after 1 January 2000 shall meet the requirements contained in appendix IV to this Annex. Each incinerator subject to this paragraph shall be approved by the Administration taking into account the standard specification for shipboard incinerators developed by the Organization;<sup>29</sup>

6.2 The Administration may allow exclusion from the application of paragraph 6.1 of this regulation to any incinerator installed on board a ship before 19 May 2005, provided that the ship is solely engaged in voyages within waters subject to the sovereignty or jurisdiction of the State the flag of which the ship is entitled to fly.

7 Incinerators installed in accordance with the requirements of paragraph 6.1 of this regulation shall be provided with a manufacturer's operating manual, which is to be retained with the unit and which shall specify how to operate the incinerator within the limits described in paragraph 2 of appendix IV of this Annex.

8 Personnel responsible for the operation of an incinerator installed in accordance with the requirements of paragraph 6.1 of this regulation shall be trained to implement the guidance provided in the manufacturer's operating manual as required by paragraph 7 of this regulation.

<sup>&</sup>lt;sup>28</sup> Type Approval Certificates issued in accordance with the *Revised guidelines for the implementation of Annex V of MARPOL* (resolution MEPC.59(33), as amended by resolution MEPC.92(45)), or *Standard specification for shipboard incinerators* (resolution MEPC.76(40), as amended by resolution MEPC.93(45)), or the 2012 Guidelines for the implementation of MARPOL Annex V (resolution MEPC.219(63), as amended by resolution MEPC.239(65)), or the 2014 Standard specification for shipboard incinerators (resolution MEPC 244(66)), or the 2017 Guidelines for the implementation of MARPOL Annex V (resolution MEPC.295(71)).

<sup>&</sup>lt;sup>29</sup> Refer to the 2014 Standard specification for shipboard incinerators (resolution MEPC.244(66)), or Standard specification for shipboard incinerators (resolution MEPC.76(40), as amended by resolution MEPC.93(45)), and Type approval of shipboard incinerators (MEPC.1/Circ.793).

9 For incinerators installed in accordance with the requirements of paragraph 6.1 of this regulation the combustion chamber gas outlet temperature shall be monitored at all times the unit is in operation. Where that incinerator is of the continuous-feed type, waste shall not be fed into the unit when the combustion chamber gas outlet temperature is below 850°C. Where that incinerator is of the batch-loaded type, the unit shall be designed so that the combustion chamber gas outlet temperature shall be designed so that the combustion chamber gas outlet temperature shall be designed so that the combustion chamber gas outlet temperature shall be designed so that the combustion chamber gas outlet temperature shall reach 600°C within five minutes after start-up. and will thereafter stabilize at a temperature not less than 850°C.

### **Regulation 17**

Reception facilities

- 1 Each Party undertakes to ensure the provision of facilities adequate to meet the:
  - .1 needs of ships using its repair ports for the reception of ozone-depleting substances and equipment containing such substances when removed from ships;
  - .2 needs of ships using its ports, terminals or repair ports for the reception of exhaust gas cleaning residues from an exhaust gas cleaning system;

without causing undue delay to ships, and

.3 needs in ship-breaking facilities for the reception of ozone-depleting substances and equipment containing such substances when removed from ships.

2 Small island developing States<sup>30</sup> may satisfy the requirements in paragraph 1 of this regulation through regional arrangements when, because of those States' unique circumstances, such arrangements are the only practical means to satisfy these requirements. Parties participating in a regional arrangement shall develop a Regional Reception Facilities Plan, taking into account the guidelines developed by the Organization.<sup>31</sup>

The Government of each Party participating in the arrangement shall consult with the Organization for circulation to the Parties of the present Convention:

- .1 how the Regional Reception Facilities Plan takes into account the Guidelines;
- .2 particulars of the identified Regional Ships Waste Reception Centres; and
- .3 particulars of those ports with only limited facilities.

3 If a particular port or terminal of a Party is, taking into account the guidelines to be developed by the Organization, remotely located from, or lacking in, the industrial infrastructure necessary to manage and process those substances referred to in paragraph 1 of this regulation and therefore cannot accept such substances, then the Party shall inform the Organization of any such port or terminal so that this information may be circulated to all Parties and Member States of the Organization for their information and any appropriate action. Each Party that has provided the Organization with such information shall also notify the Organization of its ports and terminals where reception facilities are available to manage and process such substances.

<sup>&</sup>lt;sup>30</sup> Refer to the 2012 Guidelines for the development of a regional reception facilities plan (resolution MEPC.221(63)).

<sup>&</sup>lt;sup>31</sup> Refer to the 2011 Guidelines for reception facilities under MARPOL Annex VI (resolution MEPC.199(62)).

4 Each Party shall notify the Organization for circulation to the Members of the Organization of all cases where the facilities provided under this regulation are unavailable or alleged to be inadequate.

# **Regulation 18**

Fuel oil availability and quality

# Fuel oil availability

1 Each Party shall take all reasonable steps to promote the availability of fuel oils that comply with this Annex and inform the Organization of the availability of compliant fuel oils in its ports and terminals.

2.1 If a ship is found by a Party not to be in compliance with the standards for compliant fuel oils set forth in this Annex, the competent authority of the Party is entitled to require the ship to:

- .1 present a record of the actions taken to attempt to achieve compliance; and
- .2 provide evidence that it attempted to purchase compliant fuel oil in accordance with its voyage plan and, if it was not made available where planned, that attempts were made to locate alternative sources for such fuel oil and that despite best efforts to obtain compliant fuel oil, no such fuel oil was made available for purchase.

2.2 The ship should not be required to deviate from its intended voyage or to delay unduly the voyage in order to achieve compliance.

2.3 If a ship provides the information set forth in paragraph 2.1 of this regulation, a Party shall take into account all relevant circumstances and the evidence presented to determine the appropriate action to take, including not taking control measures.

2.4 A ship shall notify its Administration and the competent authority of the relevant port of destination when it cannot purchase compliant fuel oil.

2.5 A Party shall notify the Organization when a ship has presented evidence of the nonavailability of compliant fuel oil.

### Fuel oil quality

3 Fuel oil for combustion purposes delivered to and used on board ships to which this Annex applies shall meet the following requirements:

- .1 except as provided in paragraph 3.2 of this regulation:
  - .1.1 the fuel oil shall be blends of hydrocarbons derived from petroleum refining. This shall not preclude the incorporation of small amounts of additives intended to improve some aspects of performance;
  - .1.2 the fuel oil shall be free from inorganic acid; and
  - .1.3 the fuel oil shall not include any added substance or chemical waste that:
- .1 jeopardizes the safety of ships or adversely affects the performance of the machinery, or
- .2 is harmful to personnel, or
- .3 contributes overall to additional air pollution.
- .2 fuel oil for combustion purposes derived by methods other than petroleum refining shall not:
  - .2.1 exceed the applicable sulphur content set forth in regulation 14 of this Annex;
  - .2.2 cause an engine to exceed the applicable NO<sub>X</sub> emission limit set forth in paragraphs 3, 4, 5.1.1 and 7.4 of regulation 13;
  - .2.3 contain inorganic acid; or
  - .2.4.1 jeopardize the safety of ships or adversely affect the performance of the machinery, or
  - .2.4.2 be harmful to personnel, or
  - .2.4.3 contribute overall to additional air pollution.

4 This regulation does not apply to coal in its solid form or nuclear fuels. Paragraphs 5, 6, 7.1, 7.2, 8.1, 8.2, 9.2, 9.3, and 9.4 of this regulation do not apply to gas fuels such as liquefied natural gas, compressed natural gas or liquefied petroleum gas. The sulphur content of gas fuels delivered to a ship specifically for combustion purposes on board that ship shall be documented by the supplier.

5 For each ship subject to regulations 5 and 6 of this Annex, details of fuel oil for combustion purposes delivered to and used on board shall be recorded by means of a bunker delivery note that shall contain at least the information specified in appendix V to this Annex.

6 The bunker delivery note shall be kept on board the ship in such a place as to be readily available for inspection at all reasonable times. It shall be retained for a period of three years after the fuel oil has been delivered on board.

7.1 The competent authority of a Party may inspect the bunker delivery notes on board any ship to which this Annex applies while the ship is in its port or offshore terminal, may make a copy of each delivery note, and may require the master or person in charge of the ship to certify that each copy is a true copy of such bunker delivery note. The competent authority may also verify the contents of each note through consultations with the port where the note was issued.

7.2 The inspection of the bunker delivery notes and the taking of certified copies by the competent authority under paragraph 7.1 of this regulation shall be performed as expeditiously as possible without causing the ship to be unduly delayed.

8.1 The bunker delivery note shall be accompanied by a representative sample of the fuel oil delivered taking into account the guidelines developed by the Organization.<sup>32</sup> The sample is to be sealed and signed by the supplier's representative and the master or officer in charge of the bunker operation on completion of bunkering operations and retained under the ship's control until the fuel oil is substantially consumed, but in any case for a period of not less than 12 months from the time of delivery.

8.2 If a Party requires the representative sample to be analysed, it shall be done in accordance with the verification procedure set forth in appendix VI to this Annex to determine whether the fuel oil meets the requirements of this Annex.

- 9 Parties undertake to ensure that appropriate authorities designated by them:
  - .1 maintain a register of local suppliers of fuel oil;
  - .2 require local suppliers to provide the bunker delivery note and sample as required by this regulation, certified by the fuel oil supplier that the fuel oil meets the requirements of regulations 14 and 18 of this Annex;
  - .3 require local suppliers to retain a copy of the bunker delivery note for at least three years for inspection and verification by the port State as necessary;
  - .4 take action as appropriate against fuel oil suppliers that have been found to deliver fuel oil that does not comply with that stated on the bunker delivery note;
  - .5 inform the Administration of any ship receiving fuel oil found to be noncompliant with the requirements of regulation 14 or 18 of this Annex; and
  - .6 inform the Organization for circulation to Parties and Member States of the Organization of all cases where fuel oil suppliers have failed to meet the requirements specified in regulations 14 or 18 of this Annex.

10 In connection with port State inspections carried out by Parties, the Parties further undertake to:

- .1 inform the Party or non-Party under whose jurisdiction a bunker delivery note was issued of cases of delivery of non-compliant fuel oil, giving all relevant information; and
- .2 ensure that remedial action as appropriate is taken to bring non-compliant fuel oil discovered into compliance.

11 For every ship of 400 gross tonnage and above on scheduled services with frequent and regular port calls, an Administration may decide after application and consultation with affected States that compliance with paragraph 6 of this regulation may be documented in an alternative manner that gives similar certainty of compliance with regulations 14 and 18 of this Annex.

<sup>&</sup>lt;sup>32</sup> Refer to 2009 Guidelines for the sampling of fuel oil for determination of compliance with the revised MARPOL Annex VI (resolution MEPC.182(59)).

# CHAPTER 4 – REGULATIONS ON THE CARBON INTENSITY OF INTERNATIONAL SHIPPING

## **Regulation 19**

Application

- 1 This chapter shall apply to all ships of 400 gross tonnage and above.
- 2 The provisions of this chapter shall not apply to:
  - .1 ships solely engaged in voyages within waters subject to the sovereignty or jurisdiction of the State the flag of which the ship is entitled to fly. However, each Party should ensure, by the adoption of appropriate measures, that such ships are constructed and act in a manner consistent with the requirements of chapter 4 of this Annex, so far as is reasonable and practicable.
  - .2 ships not propelled by mechanical means, and platforms including FPSOs and FSUs and drilling rigs, regardless of their propulsion.

3 Regulations 22, 23, 24 and 25 of this Annex shall not apply to ships which have non-conventional propulsion, except that regulations 22 and 24 shall apply to cruise passenger ships having non-conventional propulsion and LNG carriers having conventional or non-conventional propulsion, delivered on or after 1 September 2019, as defined in regulation 2.2.1, and regulations 23 and 25 shall apply to cruise passenger ships having non-conventional propulsion and LNG carriers having conventional or non-conventional propulsion. Regulations 22, 23, 24, 25 and 28 shall not apply to category A ships as defined in the Polar Code.

4 Notwithstanding the provisions of paragraph 1 of this regulation, the Administration may waive the requirement for a ship of 400 gross tonnage and above to comply with regulations 22 and 24 of this Annex.

5 The provision of paragraph 4 of this regulation shall not apply to ships of 400 gross tonnage and above:

- .1 for which the building contract is placed on or after 1 January 2017; or
- .2 in the absence of a building contract, the keel of which is laid or which is at a similar stage of construction on or after 1 July 2017; or
- .3 the delivery of which is on or after 1 July 2019; or
- .4 in cases of a major conversion of a new or existing ship, as defined in regulation 2.2.17 of this Annex, on or after 1 January 2017, and in which regulations 5.4.2 and 5.4.3 of this Annex apply.

6 The Administration of a Party to the present Convention which allows the application of paragraph 4, or suspends, withdraws or declines the application of that paragraph, to a ship entitled to fly its flag shall forthwith communicate to the Organization for circulation to the Parties to the present Protocol particulars thereof, for their information.

# **Regulation 20**

Goal

The goal of this chapter is to reduce the carbon intensity of international shipping, working towards the levels of ambition set out in the *Initial IMO Strategy on reduction of GHG emissions* from ships.<sup>33</sup>

## **Regulation 21**

## Functional requirements

In order to achieve the goal set out in regulation 20 of this Annex, a ship to which this chapter applies shall comply, as applicable, with the following functional requirements to reduce its carbon intensity:

- .1 the technical carbon intensity requirements in accordance with regulations 22, 23, 24 and 25 of this Annex; and
- .2 the operational carbon intensity requirements in accordance with regulations 26, 27 and 28 of this Annex.

## **Regulation 22**

Attained Energy Efficiency Design Index (attained EEDI)

- 1 The attained EEDI shall be calculated for:
  - .1 each new ship;
  - .2 each new ship which has undergone a major conversion; and
  - .3 each new or existing ship which has undergone a major conversion that is so extensive that the ship is regarded by the Administration as a newly constructed ship

which falls into one or more of the categories in regulations 2.2.5, 2.2.7, 2.2.9, 2.2.11, 2.2.14 to 2.2.16, 2.2.20, 2.2.22, and 2.2.26 to 2.2.29 of this Annex. The attained EEDI shall be specific to each ship and shall indicate the estimated performance of the ship in terms of energy efficiency, and be accompanied by the EEDI technical file that contains the information necessary for the calculation of the attained EEDI and that shows the process of calculation. The attained EEDI shall be verified, based on the EEDI technical file, either by the Administration or by any organization duly authorized by it.<sup>34</sup>

2 The attained EEDI shall be calculated taking into account the guidelines<sup>35</sup> developed by the Organization.

3 For each ship subject to regulation 24 of this Annex, the Administration or any organization duly authorized by it shall report to the Organization the required and attained

<sup>&</sup>lt;sup>33</sup> Initial IMO Strategy on reduction of GHG emissions from ships (resolution MEPC.304(72))

<sup>&</sup>lt;sup>34</sup> Refer to the Code for Recognized Organizations (RO Code), as adopted by the Organization by resolution MEPC.237(65), as may be amended by the Organization.

<sup>&</sup>lt;sup>35</sup> Refer to the 2018 Guidelines on the method of calculation of the attained Energy Efficiency Design Index (EEDI) for new ships (resolution MEPC.308(73), as amended by resolutions MEPC.322(74) and MEPC.332(76)).

EEDI values and relevant information, taking into account the guidelines developed by the Organization,<sup>36</sup> via electronic communication:

- .1 within seven months of completing the survey required under regulation 5.4 of this Annex; or
- .2 within seven months following 1 April 2022 for a ship delivered prior to 1 April 2022.

## **Regulation 23**

Attained Energy Efficiency Existing Ship Index (attained EEXI)

- 1 The attained EEXI shall be calculated for:
  - .1 each ship; and
  - .2 each ship which has undergone a major conversion

which falls into one or more of the categories in regulations 2.2.5, 2.2.7, 2.2.9, 2.2.11, 2.2.14 to 2.2.16, 2.2.22, and 2.2.26 to 2.2.29 of this Annex. The attained EEXI shall be specific to each ship and shall indicate the estimated performance of the ship in terms of energy efficiency, and be accompanied by the EEXI technical file which contains the information necessary for the calculation of the attained EEXI and which shows the process of the calculation. The attained EEXI shall be verified, based on the EEXI technical file, either by the Administration or by any organization duly authorized by it.<sup>37</sup>

2 The attained EEXI shall be calculated taking into account the guidelines<sup>38</sup> developed by the Organization.

3 Notwithstanding paragraph 1 of this regulation, for each ship to which regulation 22 of this Annex applies, the attained EEDI verified by the Administration or by any organization duly authorized by it in accordance with regulation 22.1 of this Annex may be taken as the attained EEXI if the value of the attained EEDI is equal to or less than that of the required EEXI required by regulation 25 of this Annex. In this case, the attained EEXI shall be verified based on the EEDI technical file.

## **Regulation 24**

Required EEDI

1 For each:

- .1 new ship,
- .2 new ship which has undergone a major conversion, and

<sup>&</sup>lt;sup>36</sup> Refer to the 2018 Guidelines on the method of calculation of the attained Energy Efficiency Design Index (*EEDI*) for new ships (resolution MEPC.308(73), as amended by resolutions MEPC.322(74) and MEPC.332(76)).

<sup>&</sup>lt;sup>37</sup> Refer to the Code for Recognized Organizations (RO Code), as adopted by the Organization by resolution MEPC.237(65), as may be amended by the Organization.

<sup>&</sup>lt;sup>38</sup> 2021 Guidelines on the method of calculation of the attained Energy Efficiency Existing Ship Index (EEXI) (resolution MEPC.333(76)).

.3 new or existing ship which has undergone a major conversion that is so extensive that the ship is regarded by the Administration as a newly constructed ship

which falls into one of the categories in regulations 2.2.5, 2.2.7, 2.2.9, 2.2.11, 2.2.14 to 2.2.16, 2.2.22, and 2.2.26 to 2.2.29 and to which this chapter is applicable, the attained EEDI shall be as follows:

Attained EEDI  $\leq$  Required EEDI  $= (1 - \frac{x}{100}) \cdot$  Reference line value

where X is the reduction factor specified in table 1 for the required EEDI compared to the EEDI reference line.

2 For each new and existing ship that has undergone a major conversion which is so extensive that the ship is regarded by the Administration as a newly constructed ship, the attained EEDI shall be calculated and meet the requirement of paragraph 1 of this regulation with the reduction factor applicable corresponding to the ship type and size of the converted ship at the date of the contract of the conversion, or in the absence of a contract, the commencement date of the conversion.

**Table 1** - Reduction factors (in percentage) for the EEDI relative to the EEDI reference line

Ship Type	Size	Phase 0 1 Jan 2013 – 31 Dec 2014	Phase 1 1 Jan 2015 – 31 Dec 2019	Phase 2 1 Jan 2020 – 31 Mar 2022	Phase 2 1 Jan 2020 – 31 Dec 2024	Phase 3 1 Apr 2022 and onwards	Phase 3 1 Jan 2025 and onwards
	20,000 DWT and above	0	10		20		30
Bulk carrier	10,000 and above but less than 20,000 DWT	n/a	0-10 <sup>*</sup>		0-20*		0-30*
	15,000 DWT and above	0	10	20		30	
Gas carrier	10,000 and above but less than 15,000 DWT	0	10		20		30
	2,000 and above but less than 10,000 DWT	n/a	0-10*		0-20*		0-30*
	20,000 DWT and above	0	10		20		30
Tanker	4,000 and above but less than 20,000 DWT	n/a	0-10*		0-20*		0-30*
	200,000 DWT and above	0	10	20		50	
	120,000 and above but less than 200,000 DWT	0	10	20		45	
	80,000 and above but less than 120,000 DWT	0	10	20		40	
Containership	40,000 and above but less than 80,000 DWT	0	10	20		35	
	15,000 and above but less than 40,000 DWT	0	10	20		30	
	10,000 and above but less than 15,000 DWT	n/a	0-10*	0-20*		15-30*	
General Cargo ships	15,000 DWT and above	0	10	15		30	

Ship Type	Size	Phase 0 1 Jan 2013 – 31 Dec 2014	Phase 1 1 Jan 2015 – 31 Dec 2019	Phase 2 1 Jan 2020 – 31 Mar 2022	Phase 2 1 Jan 2020 – 31 Dec 2024	Phase 3 1 Apr 2022 and onwards	Phase 3 1 Jan 2025 and onwards
	3,000 and above but less than 15,000 DWT	n/a	0-10*	0-15*		0-30*	
	5,000 DWT and above	0	10		15		30
Refrigerated cargo carrier	3,000 and above but less than 5,000 DWT	n/a	0-10*		0-15*		0-30*
	20,000 DWT and above	0	10		20		30
Combination carrier	4,000 and above but less than 20,000 DWT	n/a	0-10*		0-20*		0-30*
LNG carrier***	10,000 DWT and above	n/a	10**	20		30	
Ro-ro cargo ship (vehicle carrier)***	10,000 DWT and above	n/a	5**		15		30
	2,000 DWT and above	n/a	5**		20		30
Ro-ro cargo ship***	1,000 and above but less than 2,000 DWT	n/a	0-5*,**		0-20*		0-30*
Ro-ro	1,000 DWT and above	n/a	5**		20		30
passenger ship***	250 and above but less than 1,000 DWT	n/a	0-5*,**		0-20*		0-30*
Cruise passenger	85,000 GT and above	n/a	5**	20		30	
having non- conventional propulsion	25,000 and above but less than 85,000 GT	n/a	0-5*,**	0-20*		0-30*	

Reduction factor to be linearly interpolated between the two values dependent upon ship size. The lower value of the reduction factor is to be applied to the smaller ship size.

\*\* Phase 1 commences for those ships on 1 September 2015.

\*\*\* Reduction factor applies to those ships delivered on or after 1 September 2019, as defined in paragraph 2.1 of regulation 2.

Note: n/a means that no required EEDI applies.

3 The reference line values shall be calculated as follows:

Reference line value =  $a \cdot b^{-c}$ 

where *a*, *b* and *c* are the parameters given in table 2.

Table 2 -	Parameters	for the det	ermination o	of reference	values	for the	different	ship	types
-----------	------------	-------------	--------------	--------------	--------	---------	-----------	------	-------

Ship type defined in regulation 2	а	b	С
2.2.5 Bulk carrier	961.79	DWT of the ship	0.477
		where	
		DWT≤279,000	
		279.000 where	
		DWT > 279,000	
2.2.7 Combination carrier	1,219.00	DWT of the ship	0.488
2.2.9 Containership	174.22	DWT of the ship	0.201
2.2.11 Cruise passenger ship	170.84	GT of the ship	0.214
having non-conventional			
propulsion	4 4 0 0 0 0	DWT of the object	0.450
2.2.14 Gas carrier	1,120.00	DWI of the ship	0.456
2.2.15 General cargo ship	107.48	DWT of the ship	0.216
2.2.16 LNG carrier	2,253.7	DWT of the ship	0.474
2.2.22 Refrigerated cargo carrier	227.01	DWT of the ship	0.244
2.2.26 Ro-ro cargo ship	1405.15	DWT of the ship	
	1686.17*	DWT of the ship	
		where	0 409
		DWT≤17,000*	0.490
		17 000 where DWT	
		> 17.000*	
2.2.27 Ro-ro cargo ship (vehicle	(DWT/GT) <sup>-0.7</sup> • 780.36	DWT of the ship	
carrier)	where DWT/GT < 0.3		
	1,812.63		0.471
	where DWT/GT ≥ 0.3		
2.2.28 Ro-ro passenger ship	752.16	DWT of the ship	
	902.59*	DWT of the ship	
			0.381
		DVV1 > 10,000"	0.001
		10.000 where DWT	
		> 10,000*	
2.2.29 Tanker	1,218.80	DWT of the ship	0.488

\* to be used from phase 2 and thereafter.

4 If the design of a ship allows it to fall into more than one of the ship type definitions specified in table 2, the required EEDI for the ship shall be the most stringent (the lowest) required EEDI.

5 For each ship to which this regulation applies, the installed propulsion power shall not be less than the propulsion power needed to maintain the manoeuvrability of the ship under adverse conditions as defined in the guidelines to be developed by the Organization.<sup>39</sup>

6 At the beginning of phase 1 and at the midpoint of phase 2, the Organization shall review the status of technological developments and, if proven necessary, amend the time periods, the EEDI reference line parameters for relevant ship types and reduction rates set out in this regulation.

## **Regulation 25**

Required EEXI

- 1 For:
  - .1 each ship; and
  - .2 each ship which has undergone a major conversion

which falls into one of the categories in regulations 2.2.5, 2.2.7, 2.2.9, 2.2.11, 2.2.14 to 2.2.16, 2.2.22, and 2.2.26 to 2.2.29 and to which this chapter is applicable, the attained EEXI shall be as follows:

Attained EEXI  $\leq$  Required EEXI  $= \left(1 - \frac{y}{100}\right) \cdot$  EEDI reference line value

where Y is the reduction factor specified in Table 3 for the required EEXI compared to the EEDI reference line.

**Table 3** - Reduction factors (in percentage) for the EEXI relative to the EEDI reference line

Ship type	Size	Reduction factor
	200,000 DWT and above	15
Bulk carrier	20,000 and above but less than 200,000 DWT	20
	10,000 and above but less than 20,000 DWT	0-20*
	15,000 DWT and above	30
Gas carrier	10,000 and above but less than 15,000 DWT	20
	2,000 and above but less than 10,000 DWT	0-20*
Tankar	200,000 DWT and above	15
i alikei	20,000 and above but less than 200,000 DWT	20

<sup>&</sup>lt;sup>39</sup> Refer to the 2013 Interim guidelines for determining minimum propulsion power to maintain the manoeuvrability of ships in adverse conditions (resolution MEPC.232(65), as amended by resolutions MEPC.255(67) and MEPC.262(68)): consolidated text: MEPC.1/Circ.850/Rev.2, and the Guidelines for determining minimum propulsion power to maintain the manoeuvrability of ships in adverse conditions (MEPC.1/Circ.850/Rev.3).

Ship type	Size	Reduction factor
	4,000 and above but less than 20,000 DWT	0-20*
	200,000 DWT and above	50
	120,000 and above but less than 200,000 DWT	45
O antain anakin	80,000 and above but less than 120,000 DWT	35
Containership	40,000 and above but less than 80,000 DWT	30
	15,000 and above but less than 40,000 DWT	20
	10,000 and above but less than 15,000 DWT	0-20*
Conoral corgo chin	15,000 DWT and above	30
General cargo ship	3,000 and above but less than 15,000 DWT	0-30*
Pofrigorotod oprao oprior	5,000 DWT and above	15
Reingerated cargo carner	3,000 and above but less than 5,000 DWT	0-15*
Combination corrier	20,000 DWT and above	20
Combination carrier	4,000 and above but less than 20,000 DWT	0-20*
LNG carrier	10,000 DWT and above	30
Ro-ro cargo ship (vehicle carrier)	10,000 DWT and above	15
	2,000 DWT and above	5
Ro-ro cargo snip	1,000 and above but less than 2,000 DWT	0-5*
Do ro possonaor chin	1,000 DWT and above	5
Ro-ro passenger snip	250 and above but less than 1,000 DWT	0-5*
Cruise passenger ship	85,000 GT and above	30
propulsion	25,000 and above but less than 85,000 GT	0-30*

\* Reduction factor to be linearly interpolated between the two values dependent upon ship size. The lower value of the reduction factor is to be applied to the smaller ship size.

2 The EEDI reference line values shall be calculated in accordance with regulations 24.3 and 24.4 of this Annex. For ro-ro cargo ships and ro-ro passenger ships, the reference line value to be used from phase 2 and thereafter under regulation 24.3 of this Annex shall be referred to.

3 A review shall be completed by 1 January 2026 by the Organization to assess the effectiveness of this regulation taking into account any guidelines developed by the Organization. If, based on the review, the Parties decide to adopt amendments to this regulation, such amendments shall be adopted and brought into force in accordance with the provisions of article 16 of the present Convention.

## **Regulation 26**

Ship Energy Efficiency Management Plan (SEEMP)

1 Each ship shall keep on board a ship specific Ship Energy Efficiency Management Plan (SEEMP). This may form part of the ship's Safety Management System (SMS). The SEEMP shall be developed and reviewed, taking into account the guidelines adopted by the Organization.<sup>40</sup>

2 In the case of a ship of 5,000 gross tonnage and above, the SEEMP shall include a description of the methodology that will be used to collect the data required by regulation 27.1 of this Annex and the processes that will be used to report the data to the ship's Administration.

3 In the case of a ship of 5,000 gross tonnage and above, which falls into one or more of the categories in regulations 2.2.5, 2.2.7, 2.2.9, 2.2.11, 2.2.14 to 2.2.16, 2.2.22, and 2.2.26 to 2.2.29 of this Annex:

- .1 On or before 1 January 2023 the SEEMP shall include:
  - .1 a description of the methodology that will be used to calculate the ship's attained annual operational CII required by regulation 28 of this Annex and the processes that will be used to report this value to the ship's Administration;
  - .2 the required annual operational CII, as specified in regulation 28 of this Annex, for the next three years;
  - .3 an implementation plan documenting how the required annual operational CII will be achieved during the next three years; and
  - .4 a procedure for self-evaluation and improvement.
- .2 For a ship rated as D for three consecutive years or rated as E in accordance with regulation 28 of this Annex, the SEEMP shall be reviewed in accordance with regulation 28.8 of this Annex to include a plan of corrective actions to achieve the required annual operational CII.
- .3 The SEEMP shall be subject to verification and company audits taking into account the guidelines to be developed by the Organization.

## **Regulation 27**

Collection and reporting of ship fuel oil consumption data

1 From calendar year 2019, each ship of 5,000 gross tonnage and above shall collect the data specified in appendix IX to this Annex, for that and each subsequent calendar year or portion thereof, as appropriate according to the methodology included in the SEEMP.

<sup>&</sup>lt;sup>40</sup> Refer to the 2016 Guidelines for the development of a ship energy efficiency management plan (SEEMP) (resolution MEPC.282(70)).

2 Except as provided for in paragraphs 4, 5 and 6 of this regulation, at the end of each calendar year, the ship shall aggregate the data collected in that calendar year or portion thereof, as appropriate.

3 Except as provided for in paragraphs 4, 5 and 6 of this regulation, within three months after the end of each calendar year, the ship shall report to its Administration or any organization duly authorized by it,<sup>41</sup> the aggregated value for each datum specified in appendix IX to this Annex, via electronic communication and using a standardized format to be developed by the Organization.<sup>42</sup>

4 In the event of the transfer of a ship from one Administration to another, the ship shall on the day of completion of the transfer or as close as practical thereto report to the losing Administration or any organization duly authorized by it<sup>41</sup>, the aggregated data for the period of the calendar year corresponding to that Administration, as specified in appendix IX to this Annex and, upon prior request of that Administration, the disaggregated data.

5 In the event of a change from one company to another, the ship shall on the day of completion of the change or as close as practical thereto report to its Administration or any organization duly authorized by it,<sup>411</sup> the aggregated data for the portion of the calendar year corresponding to the company, as specified in appendix IX to this Annex and, upon request of its Administration, the disaggregated data.

6 In the event of change from one Administration to another and from one company to another concurrently, paragraph 4 of this regulation shall apply.

7 The data shall be verified according to procedures established by the Administration, taking into account the guidelines developed by the Organization.<sup>43</sup>

8 Except as provided for in paragraphs 4, 5 and 6 of this regulation, the disaggregated data that underlies the reported data noted in appendix IX to this Annex for the previous calendar year shall be readily accessible for a period of not less than 12 months from the end of that calendar year and be made available to the Administration upon request.

9 The Administration shall ensure that the reported data noted in appendix IX to this Annex by its registered ships of 5,000 gross tonnage and above are transferred to the IMO Ship Fuel Oil Consumption Database via electronic communication and using a standardized format to be developed by the Organization not later than one month after issuing the Statements of Compliance of these ships.

10 On the basis of the reported data submitted to the IMO Ship Fuel Oil Consumption Database, the Secretary-General of the Organization shall produce an annual report to the Marine Environment Protection Committee summarizing the data collected, the status of missing data, and such other relevant information as may be requested by the Committee.

11 The Secretary-General of the Organization shall grant the Administration of a ship to which regulation 28 of this Annex applies access to all the reported data for all the preceding calendar year in the IMO Ship Fuel Oil Consumption Database for that ship.

<sup>&</sup>lt;sup>41</sup> Refer to the Code for Recognized Organizations (RO Code), as adopted by the Organization by resolution MEPC.237(65), as may be amended by the Organization.

<sup>&</sup>lt;sup>42</sup> Refer to the 2016 Guidelines for the development of a Ship Energy Efficiency Management Plan (SEEMP Guidelines) (resolution MEPC.282(70)).

<sup>&</sup>lt;sup>43</sup> Refer to the 2017 Guidelines for Administration verification of ship fuel oil consumption data (resolution MEPC.292(71))

12 The Secretary-General of the Organization shall maintain an anonymized database such that identification of a specific ship will not be possible. Parties shall have access to the anonymized data strictly for their analysis and consideration.

13 The IMO Ship Fuel Oil Consumption Database shall be undertaken and managed by the Secretary-General of the Organization, pursuant to guidelines to be developed by the Organization.

# **Regulation 28**

Operational carbon intensity

# Attained annual operational carbon intensity indicator (attained annual operational CII)

1 After the end of calendar year 2023 and after the end of each following calendar year, each ship of 5,000 gross tonnage and above which falls into one or more of the categories in regulations 2.2.5, 2.2.7, 2.2.9, 2.2.11, 2.2.14 to 2.2.16, 2.2.22, and 2.2.26 to 2.2.29 of this Annex shall calculate the attained annual operational CII over a 12-month period from 1 January to 31 December for the preceding calendar year, using the data collected in accordance with regulation 27 of this Annex, taking into account the guidelines to be developed by the Organization.

2 Within three months after the end of each calendar year, the ship shall report to its Administration, or any organization duly authorized by it, the attained annual operational CII via electronic communication and using a standardized format to be developed by the Organization.

3 Notwithstanding 1 and 2 of this regulation, in the event of any transfer of a ship addressed in regulations 27.4, 27.5 or 27.6 completed after 1 January 2023, a ship shall, after the end of the calendar year in which the transfer takes place, calculate and report the attained annual operational CII for the full 12-month period from 1 January to 31 December in the calendar year during which the transfer took place, in accordance with regulations 28.1 and 28.2, for verification in accordance with regulation 6.6 of this Annex, taking into account guidelines to be developed by the Organization. Nothing in this regulation relieves any ship of its reporting obligations under regulation 27 or this regulation of this Annex.

# Required annual operational carbon intensity indicator (required annual operational CII)

4 For each ship of 5,000 gross tonnage and above which falls into one or more of the categories in regulations 2.2.5, 2.2.7, 2.2.9, 2.2.11, 2.2.14 to 2.2.16, 2.2.22, and 2.2.26 to 2.2.29 of this Annex, the required annual operational CII shall be determined as follows:

Required annual operational CII = 
$$\left(1 - \frac{z}{100}\right) \cdot \text{CII}_{R}$$

where,

Z is the annual reduction factor to ensure continuous improvement of the ship's operational carbon intensity within a specific rating level; and

 $CII_R$  is the reference value.

5 The annual reduction factor  $Z^{44}$  and the reference value CII<sub>R</sub> shall be the values defined taking into account the guidelines to be developed by the Organization.

<sup>&</sup>lt;sup>44</sup> The annual reduction factor is specific to each category of ship. This factor is defined to increase progressively to meet the objectives of the *Initial IMO Strategy on reduction of GHG emissions from ships* (resolution MEPC.304(72)).

## Operational carbon intensity rating

6 The attained annual operational CII shall be documented and verified against the required annual operational CII to determine operational carbon intensity rating A, B, C, D or E, indicating a major superior, minor superior, moderate, minor inferior, or inferior performance level, either by the Administration or by any organization duly authorized by it, taking into account the guidelines developed by the Organization. The middle point of rating level C shall be the value equivalent to the required annual operational CII set out in paragraph 4 of this regulation.

## Corrective actions and incentives

7 A ship rated as D for three consecutive years or rated as E shall develop a plan of corrective actions to achieve the required annual operational CII.

8 The SEEMP shall be reviewed to include the plan of corrective actions accordingly, taking into account the guidelines to be developed by the Organization. The revised SEEMP shall be submitted to the Administration or any organization duly authorized by it for verification, preferably together with, but in no case later than 1 month after reporting the attained annual operational CII in accordance with paragraph 2 of this regulation.

9 A ship rated as D for three consecutive years or rated as E shall duly undertake the planned corrective actions in accordance with the revised SEEMP.

10 Administrations, port authorities and other stakeholders as appropriate, are encouraged to provide incentives to ships rated as A or B.

### Review

- 11 A review shall be completed by 1 January 2026 by the Organization to assess:
  - .1 the effectiveness of this regulation in reducing the carbon intensity of international shipping;
  - .2 the need for reinforced corrective actions or other means of remedy, including possible additional EEXI requirements;
  - .3 the need for enhancement of the enforcement mechanism;
  - .4 the need for enhancement of the data collection system; and
  - .5 the revision of the Z factor and CII<sub>R</sub> values.

If based on the review the Parties decide to adopt amendments to this regulation, such amendments shall be adopted and brought into force in accordance with the provisions of article 16 of the present Convention.

## **Regulation 29**

Promotion of technical cooperation and transfer of technology relating to the improvement of energy efficiency of ships<sup>45</sup>

1 Administrations shall, in cooperation with the Organization and other international bodies, promote and provide support, as appropriate, directly or through the Organization to States that request technical assistance, especially developing States.

2 The Administration of a Party shall cooperate actively with other Parties, subject to its national laws, regulations and policies, to promote the development and transfer of technology and exchange of information to States which request technical assistance, particularly developing States, in respect of the implementation of measures to fulfil the requirements of chapter 4 of this Annex, in particular regulations 19.4 to 19.6.

## Chapter 5 – Verification of compliance with the provisions of this Annex

## **Regulation 30**

Application

Parties shall use the provisions of the Code for Implementation in the execution of their obligations and responsibilities contained in this Annex.

## **Regulation 31**

Verification of compliance

1 Every Party shall be subject to periodic audits by the Organization in accordance with the audit standard to verify compliance with and implementation of this Annex.

2 The Secretary-General of the Organization shall have responsibility for administering the Audit Scheme, based on the guidelines developed by the Organization.<sup>46</sup>

3 Every Party shall have responsibility for facilitating the conduct of the audit and implementation of a programme of actions to address the findings, based on the guidelines developed by the Organization.<sup>46</sup>

4 The audits of all Parties shall be:

- .1 based on an overall schedule developed by the Secretary-General of the Organization, taking into account the guidelines developed by the Organization;46 and
- .2 conducted at periodic intervals, taking into account the guidelines developed by the Organization.46

<sup>&</sup>lt;sup>45</sup> Refer to Promotion of technical cooperation and transfer of technology relating to the improvement of energy efficiency of ships (resolution MEPC.229(65)), and the Model agreement between governments on technological cooperation for the implementation of the regulations in chapter 4 of MARPOL Annex VI (MEPC.1/Circ.861).

<sup>&</sup>lt;sup>46</sup> Refer to the *Framework and procedures for the IMO Member State Audit Scheme* (resolution A.1067(28)).

Appendix I

## Form of International Air Pollution Prevention (IAPP) Certificate (regulation 8)

INTERNATIONAL AIR POLLUTION PREVENTION CERTIFICATE

Issued under the provisions of the Protocol of 1997, as amended, to amend the International Convention for the Prevention of Pollution from Ships, 1973, as modified by the Protocol of 1978 relating thereto (hereinafter referred to as "the Convention") under the authority of the Government of:

(full designation of the country) by...... (full designation of the competent person or organization authorized under the provisions of the Convention)

Particulars of ship<sup>1</sup>

Name of ship
Distinctive number or letters
IMO Number <sup>2</sup>

Port of registry .....

Gross tonnage .....

THIS IS TO CERTIFY:

1 That the ship has been surveyed in accordance with regulation 5 of Annex VI of the Convention; and

2 That the survey shows that the equipment, systems, fittings, arrangements and materials fully comply with the applicable requirements of Annex VI of the Convention.

This Certificate is valid until (dd/mm/yyyy)<sup>3</sup>.....subject to surveys in accordance with regulation 5 of Annex VI of the Convention.

Completion date of the survey on which this Certificate is based (dd/mm/yyyy).....

Issued at .....

(place of issue of Certificate)

Date (dd/mm/yyyy) ..... (date of issue)

(signature of duly authorized official issuing the Certificate)

<sup>&</sup>lt;sup>1</sup> Alternatively, the particulars of the ship may be placed horizontally in boxes.

<sup>&</sup>lt;sup>2</sup> In accordance with the IMO Ship Identification Number Scheme (resolution A.1117(30)).

<sup>&</sup>lt;sup>3</sup> Insert the date of expiry as specified by the Administration in accordance with regulation 9.1 of Annex VI of the Convention. The day and the month of this date correspond to the anniversary date as defined in regulation 2.1.3 of Annex VI of the Convention, unless amended in accordance with regulation 9.8 of Annex VI of the Convention.

## (seal or stamp of the authority, as appropriate)

## ENDORSEMENT FOR ANNUAL AND INTERMEDIATE SURVEYS

THIS IS TO CERTIFY that, at a survey required by regulation 5 of Annex VI of the Convention, the ship was found to comply with the relevant provisions of that Annex:

Annual survey	Signed (signature of duly authorized official)
	Place
(seal or stamp of the a	Date (dd/mm/yyyy) authority, as appropriate)
Annual/Intermediate <sup>4</sup> survey	Signed (signature of duly authorized official)
	Place
(seal or stamp of the a	Date (dd/mm/yyyy) authority, as appropriate)
Annual/Intermediate <sup>4</sup> survey	Signed (signature of duly authorized official)
	Place
(seal or stamp of the a	Date (dd/mm/yyyy) authority, as appropriate)
Annual survey	Signed (signature of duly authorized official)
	Place
(seal or stamp of the a	Date (dd/mm/yyyy) authority, as appropriate)
ANNUAL/INTERMEDIATE	SURVEY IN ACCORDANCE

WITH REGULATION 9.8.3

THIS IS TO CERTIFY that, at an annual/intermediate<sup>4</sup> survey in accordance with regulation 9.8.3 of Annex VI of the Convention, the ship was found to comply with the relevant provisions of that Annex:

Signed.....(signature of duly authorized official)

Place .....

<sup>4</sup> Delete as appropriate.

Date (dd/mm/yyyy) .....

(seal or stamp of the authority, as appropriate)

## ENDORSEMENT TO EXTEND THE CERTIFICATE IF VALID FOR LESS THAN FIVE YEARS WHERE REGULATION 9.3 APPLIES

The ship complies with the relevant provisions of the Annex, and this Certificate shall, in accordance with regulation 9.3 of Annex VI of the Convention, be accepted as valid until (dd/mm/yyyy) .....

Signed.
(signature of duly authorized official)
Place
Date (dd/mm/yyyy)

## ENDORSEMENT WHERE THE RENEWAL SURVEY HAS BEEN COMPLETED AND REGULATION 9.4 APPLIES

The ship complies with the relevant provisions of the Annex, and this Certificate shall, in accordance with regulation 9.4 of Annex VI of the Convention, be accepted as valid until (dd/mm/yyyy) .....

Place .....

Date (dd/mm/yyyy) .....

## (seal or stamp of the authority, as appropriate)

## ENDORSEMENT TO EXTEND THE VALIDITY OF THE CERTIFICATE UNTIL REACHING THE PORT OF SURVEY OR FOR A PERIOD OF GRACE WHERE REGULATION 9.5 OR 9.6 APPLIES

This Certificate shall, in accordance with regulation 9.5 or 9.6<sup>5</sup> of Annex VI of the Convention, be accepted as valid until (dd/mm/yyyy).....

Signed. ...... (signature of duly authorized official)

Place .....

Date (dd/mm/yyyy) .....

<sup>&</sup>lt;sup>5</sup> Delete as appropriate.

## ENDORSEMENT FOR ADVANCEMENT OF ANNIVERSARY DATE WHERE REGULATION 9.8 APPLIES

In accordance with regulation 9.8 of Annex VI of the Convention, the new anniversary date is (dd/mm/yyyy).....

Signed. .....(signature of duly authorized official)

Place .....

Date (dd/mm/yyyy) .....

## (seal or stamp of the authority, as appropriate)

In accordance with regulation 9.8 of Annex VI of the Convention, the new anniversary date is (dd/mm/yyyy).....

Signed..... (signature of duly authorized official) Place

Date (dd/mm/yyyy) .....

## SUPPLEMENT TO INTERNATIONAL AIR POLLUTION PREVENTION CERTIFICATE (IAPP CERTIFICATE)

# RECORD OF CONSTRUCTION AND EQUIPMENT

#### Notes

- 1 This Record shall be permanently attached to the IAPP Certificate. The IAPP Certificate shall be available on board the ship at all times.
- 2 The Record shall be at least in English, French or Spanish. If an official language of the issuing country is also used, this shall prevail in case of a dispute or discrepancy.
- 3 Entries in boxes shall be made by inserting either: a cross (x) for the answers "yes" and "applicable"; or a dash (–) for the answers "no" and "not applicable", as appropriate.
- 4 Unless otherwise stated, regulations mentioned in this Record refer to regulations of Annex VI of the Convention and resolutions or circulars refer to those adopted by the International Maritime Organization.
- 1 Particulars of ship
- 1.1 Name of ship.....
- 1.2 IMO Number.....
- 1.3 Date on which keel was laid or ship was at a similar stage of construction (dd/mm/yyyy).....
- 1.4 Length (*L*)<sup>6</sup> metres .....
- 2 Control of emissions from ships
- 2.1 *Ozone-depleting substances* (regulation 12)
- 2.1.1 The following fire-extinguishing systems, other systems and equipment containing ozone-depleting substances, other than hydrochlorofluorocarbons (HCFCs), installed before 19 May 2005 may continue in service:

System or equipment	Location on board	Substance

<sup>&</sup>lt;sup>6</sup> Completed only in respect of ships constructed on or after 1 January 2016 that are specially designed, and used solely for recreational purposes and to which, in accordance with regulation 13.5.2.1 or regulation 13.5.2.3, the NO<sub>X</sub> emission limit as given by regulation 13.5.1.1 will not apply.

2.1.2 The following systems containing HCFCs installed before 1 January 2020 may continue in service:

System or equipment	Location on board	Substance

## 2.2 Nitrogen oxides (NO<sub>X</sub>) (regulation 13)

2.2.1 The following marine diesel engines installed on this ship are in accordance with the requirements of regulation 13, as indicated:

	Applicable regulation (NTC = NO <sub>X</sub> Ter (AM = appr	n of MARPOL Annex VI chnical Code 2008) oved method)	Engine #1	Engine #2	Engine #3	Engine #4	Engine #5
1	Manufacturer and mode						
2	Serial number						
3	Use (applicable applicat	ion cycle(s) – NTC 3.2)					
4	Rated power (kW) (NTC	1.3.11)					
5	Rated speed (rpm) (NTC	1.3.12)					
6	Identical engine installe	ed $\geq$ 1/1/2000 exempted by					
7	Identical engine installat 13.1.1.2	ion date (dd/mm/yyyy) as per					
8a		13.2.1.1 & 13.2.2					
8b	(dd/mm/yyyy)	13.2.1.2 & 13.2.3					
8c		13.2.1.3 & 13.2.3					
9a		13.3					
9b	Tier I	13.2.2					
9c		13.2.3.1					
9d		13.2.3.2					
9e		13.7.1.2					
10a		13.4					
10b		13.2.2					
10c	Tier II	13.2.2 (Tier III not possible)					
10d		13.2.3.2					
10e		13.5.2 (Exemptions)					
10f		13.7.1.2					
11a	NO <sub>x</sub> Tier III Emission	13.5.1.1					
11b	Control Areas	13.2.2					
11c		13.2.3.2					

Applicable regulation of MARPOL Annex VI (NTC = NO <sub>X</sub> Technical Code 2008) (AM = approved method)			Engine #1	Engine #2	Engine #3	Engine #4	Engine #5
11d		13.7.1.2					
12		installed					
13	AM <sup>7</sup>	not commercially available at this survey					
14		not applicable					

- 2.3 Sulphur oxides (SO<sub>X</sub>) and particulate matter (regulation 14)
- 2.3.1 When the ship operates outside of an emission control area specified in regulation 14.3, the ship uses:

  - .2 an equivalent arrangement approved in accordance with regulation 4.1 as listed in paragraph 2.6 that is at least as effective in terms of SO<sub>X</sub> emission reductions as compared to using a fuel oil with a sulphur content limit value of 0.50% m/m .....□
- 2.3.2 When the ship operates inside an emission control area specified in regulation 14.3, the ship uses:
  - .1 fuel oil with a sulphur content as documented by bunker delivery notes that does not exceed the limit value of 0.10% m/m, and/or
  - .2 an equivalent arrangement approved in accordance with regulation 4.1 as listed in paragraph 2.6 that is at least as effective in terms of SO<sub>X</sub> emission reductions as compared to using a fuel oil with a sulphur content limit value of 0.10% m/m ......
- 2.3.4 The ship is fitted with designated sampling point(s) in accordance with regulation 14.10 or 14.11......
- 2.4 Volatile organic compounds (VOCs) (regulation 15)
- 2.4.1 The tanker has a vapour collection system installed and approved in accordance with MSC/Circ.585......

<sup>&</sup>lt;sup>7</sup> Refer to 2014 Guidelines on the approved method process (resolution MEPC.243(66)).

2.4.2.1 For a tanker carrying crude oil, there is an approved VOC management plan

2.4.2.2 VOC management plan approval reference .....

2.5 *Shipboard incineration* (regulation 16)

The ship has an incinerator:

- .1 installed on or after 1 January 2000 that complies with:
  - .1 resolution MEPC.76(40), as amended<sup>8</sup> ......
  - .2 resolution MEPC.244(66) .....
- .2 installed before 1 January 2000 that complies with:
  - .1 resolution MEPC.59(33), as amended<sup>9</sup> ......
  - .2 resolution MEPC.76(40), as amended<sup>10</sup>......

## 2.2 *Equivalents* (regulation 4)

The ship has been allowed to use the following fitting, material, appliance or apparatus to be fitted in a ship or other procedures, alternative fuel oils, or compliance methods used as an alternative to that required by this Annex:

System or equipment	Equivalent used	Approval reference	

THIS IS TO CERTIFY that this Record is correct in all respects.

(date of issue)

(signature of duly authorized official issuing the Record)

<sup>&</sup>lt;sup>8</sup> As amended by resolution MEPC.93(45).

<sup>&</sup>lt;sup>9</sup> As amended by resolution MEPC.92(45).

<sup>&</sup>lt;sup>10</sup> As amended by resolution MEPC.93(45).

# Appendix II

## Test cycles and weighting factors (regulation 13)

The following test cycles and weighting factors shall be applied for verification of compliance of marine diesel engines with the applicable  $NO_X$  limit in accordance with regulation 13 of this Annex using the test procedure and calculation method as specified in the revised  $NO_X$  Technical Code 2008.

- .1 For constant-speed marine engines for ship main propulsion, including dieselelectric drive, test cycle E2 shall be applied.
- .2 For controllable-pitch propeller sets test cycle E2 shall be applied.
- .3 For propeller-law-operated main and propeller-law-operated auxiliary engines the test cycle E3 shall be applied.
- .4 For constant-speed auxiliary engines test cycle D2 shall be applied.
- .5 For variable-speed, variable-load auxiliary engines, not included above, test cycle C1 shall be applied.

Test cycle for *constant-speed main propulsion* application (including diesel-electric drive and all controllable-pitch propeller installations)

	Speed	100%	100%	100%	100%
Test cycle type	Power	100%	75%	50%	25%
EZ	Weighting factor	0.2	0.5	0.15	0.15

Test cycle for propeller-law-operated main and propeller-law-operated auxiliary engine application

	Speed	100%	91%	80%	63%
Test cycle type E3	Power	100%	75%	50%	25%
	Weighting factor	0.2	0.5	0.15	0.15

Test cycle for constant-speed auxiliary engine application

	Speed	100%	100%	100%	100%	100%
Test cycle type	Power	100%	75%	50%	25%	10%
D2	Weighting factor	0.05	0.25	0.3	0.3	0.1

Test cycle for variable-speed and variable-load auxiliary engine application

	Speed		Ra	ated		Int	ermedia	ate	Idle
Test cycle type C1	Torque	100%	75%	50%	10%	100%	75%	50%	0%
	Weighting factor	0.15	0.15	0.15	0.1	0.1	0.1	0.1	0.15

In the case of an engine to be certified in accordance with paragraph 5.1.1 of regulation 13, the specific emission at each individual mode point shall not exceed the applicable  $NO_X$  emission limit value by more than 50% except as follows:

- .1 The 10% mode point in the D2 test cycle.
- .2 The 10% mode point in the C1 test cycle.
- .3 The idle mode point in the C1 test cycle.

# Appendix III

# Criteria and procedures for the designation of emission control areas (regulations 13.6 and 14.3)

## 1 **Objectives**

1.1 The purpose of this appendix is to provide Parties with the criteria and procedures for formulating and submitting proposals for the designation of emission control areas and to set forth the factors to be considered in the assessment of such proposals by the Organization.

1.2 Emissions of  $NO_X$ ,  $SO_X$  and particulate matter from ocean-going ships contribute to ambient concentrations of air pollution in cities and coastal areas around the world. Adverse public health and environmental effects associated with air pollution include premature mortality, cardiopulmonary disease, lung cancer, chronic respiratory ailments, acidification and eutrophication.

1.3 An emission control area should be considered for adoption by the Organization if supported by a demonstrated need to prevent, reduce and control emissions of  $NO_X$  or  $SO_X$  and particulate matter or all three types of emissions (hereinafter emissions) from ships.

## 2 **Process for the designation of emission control areas**

2.1 A proposal to the Organization for the designation of an emission control area for  $NO_X$  or  $SO_X$  and particulate matter or all three types of emissions may be submitted only by Parties. Where two or more Parties have a common interest in a particular area, they should formulate a coordinated proposal.

2.2 A proposal to designate a given area as an emission control area should be submitted to the Organization in accordance with the rules and procedures established by the Organization.

## 3 Criteria for designation of an emission control area

## 3.1 The proposal shall include:

- .1 a clear delineation of the proposed area of application, along with a reference chart on which the area is marked;
- .2 the type or types of emission(s) that is or are being proposed for control (i.e.  $NO_X$  or  $SO_X$  and particulate matter or all three types of emissions);
- .3 a description of the human populations and environmental areas at risk from the impacts of ship emissions;
- .4 an assessment that emissions from ships operating in the proposed area of application are contributing to ambient concentrations of air pollution or to adverse environmental impacts. Such assessment shall include a description of the impacts of the relevant emissions on human health and the environment, such as adverse impacts on terrestrial and aquatic ecosystems, areas of natural productivity, critical habitats, water quality, human health, and areas of cultural and scientific significance, if applicable. The sources of relevant data including methodologies used shall be identified;

- .5 relevant information, pertaining to the meteorological conditions in the proposed area of application, to the human populations and environmental areas at risk, in particular prevailing wind patterns, or to topographical, geological, oceanographic, morphological or other conditions that contribute to ambient concentrations of air pollution or adverse environmental impacts;
- .6 the nature of the ship traffic in the proposed emission control area, including the patterns and density of such traffic;
- .7 a description of the control measures taken by the proposing Party or Parties addressing land-based sources of  $NO_X$ ,  $SO_X$  and particulate matter emissions affecting the human populations and environmental areas at risk that are in place and operating concurrently with the consideration of measures to be adopted in relation to provisions of regulations 13 and 14 of Annex VI; and
- .8 the relative costs of reducing emissions from ships when compared with landbased controls, and the economic impacts on shipping engaged in international trade.

3.2 The geographical limits of an emission control area will be based on the relevant criteria outlined above, including emissions and deposition from ships navigating in the proposed area, traffic patterns and density, and wind conditions.

# 4 Procedures for the assessment and adoption of emission controlareas by the Organization

4.1 The Organization shall consider each proposal submitted to it by a Party or Parties.

4.2 In assessing the proposal, the Organization shall take into account the criteria that are to be included in each proposal for adoption as set forth in section 3 above.

4.3 An emission control area shall be designated by means of an amendment to this Annex, considered, adopted and brought into force in accordance with article 16 of the present Convention.

# 5 **Operation of emission control areas**

5.1 Parties that have ships navigating in the area are encouraged to bring to the Organization any concerns regarding the operation of the area.

# Appendix IV

## Type approval and operating limits for shipboard incinerators (regulation 16)

1 Shipboard incinerators described in regulation 16.6.1 shall possess an IMO Type Approval Certificate for each incinerator. In order to obtain such certificate, the incinerator shall be designed and built to an approved standard as described in regulation 16.6.1. Each model shall be subject to a specified type approval test operation at the factory or an approved test facility, and under the responsibility of the Administration, using the following standard fuel/waste specification for the type approval test for determining whether the incinerator operates within the limits specified in paragraph 2 of this appendix:

Sludge oil consisting of:	75% sludge oil from heavy fuel oil (HFO); 5% waste lubricating oil; and 20% `emulsified water.		
Solid waste consisting of:	50% food waste; 50% rubbish containing: approx. 30% paper, " 40% cardboard, " 10% rags, " 20% plastic.		
	The mixture will have up to		

The mixture will have up to 50% moisture and 7% incombustible solids.

2 Incinerators described in regulation 16.6.1 shall operate within the following limits:

O <sub>2</sub> in combustion chamber:	6–12%
CO in flue gas maximum average:	200 mg/MJ
Soot number maximum average:	Bacharach 3 or Ringelmann 1 (20% opacity) (a higher soot number is acceptable only during very short periods such as starting up)
Unburned components in ash residues:	Maximum 10% by weight
Combustion chamber flue gas outlet temperature range:	850–1200°C

# Appendix V

## Information to be included in the bunker delivery note (regulation 18.5)

- 1 Name and IMO Number of receiving ship
- 2 Port
- 3 Date of commencement of delivery
- 4 Name, address and telephone number of marine fuel oil supplier
- 5 Product name(s)
- 6 Quantity in metric tonnes
- 7 Density at 15°C (kg/m<sup>3</sup>)<sup>1</sup>
- 8 Sulphur content (% m/m)<sup>2</sup>
- 9 A declaration signed and certified by the fuel oil supplier's representative that the fuel oil supplied is in conformity with regulation 18.3 of this Annex and that the sulphur content of the fuel oil supplied does not exceed:
  - the limit value given by regulation 14.1 of this Annex;
  - the limit value given by regulation 14.4 of this Annex; or
  - the purchaser's specified limit value of \_\_\_\_\_ (% m/m), as completed by the fuel oil supplier's representative and on the basis of the purchaser's notification that the fuel oil:
    - .1 is intended to be used in combination with an equivalent means of compliance in accordance with regulation 4 of this Annex; or
    - .2 is subject to a relevant exemption for a ship to conduct trials for sulphur oxides emission reduction and control technology research in accordance with regulation 3.2 of this Annex.

The declaration shall be completed by the fuel oil supplier's representative by marking the applicable box(es) with a cross (x).

<sup>&</sup>lt;sup>1</sup> Fuel oil shall be tested in accordance with ISO 3675:1998 or ISO 12185:1996.

<sup>&</sup>lt;sup>2</sup> Fuel oil shall be tested in accordance with ISO 8754:2003.

## Appendix VI

# Verification procedures for a MARPOL Annex VI fuel oil sample (regulation 18.8.2 or regulation 14.8)

The following relevant verification procedure shall be used to determine whether the fuel oil delivered to, in use or carried for use on board a ship has met the applicable sulphur limit of regulation 14 of this Annex.

This appendix refers to the following representative MARPOL Annex VI fuel oil samples:

Part 1 – sample of fuel oil delivered<sup>1</sup> in accordance with regulation 18.8.1, hereafter referred to as the "MARPOL delivered sample" as defined in regulation 2.1.22.

Part 2 – sample of fuel oil in use,<sup>2</sup> intended to be used or carried for use on board in accordance with regulation 14.8, hereafter referred to as the "in-use sample" as defined in regulation 2.1.16 and "onboard sample"<sup>3</sup> as defined in regulation 2.1.24.

## Part 1 – MARPOL delivered sample

## 1 General Requirements

1.1 The representative sample of the fuel oil, which is required by regulation 18.8.1 (the MARPOL delivered sample), shall be used to verify the sulphur content of the fuel oil delivered to a ship.

1.2 A Party, through its competent authority, shall manage the verification procedure.

1.3 A laboratory undertaking the sulphur testing procedure given in this appendix shall have valid accreditation<sup>4</sup> in respect of the test method to be used.

### 2 Verification Procedure Part 1

2.1 The MARPOL delivered sample shall be conveyed by the competent authority to the laboratory.

- 2.2 The laboratory shall:
  - .1 record the details of the seal number and the sample label on the test record;
  - .2 record the condition of the seal of the sample as received on the test record; and
  - .3 reject any sample where the seal has been broken prior to receipt and record that rejection on the test record.

<sup>&</sup>lt;sup>1</sup> Samples taken in accordance with the 2009 Guidelines for the sampling of fuel oil for determination of compliance with the revised MARPOL Annex VI (resolution MEPC.182(59)).

<sup>&</sup>lt;sup>2</sup> Samples taken in accordance with the 2019 Guidelines for on board sampling for the verification of the sulphur content of the fuel oil used on board ships (MEPC.1/Circ.864/Rev.1).

<sup>&</sup>lt;sup>3</sup> Refer to the 2020 Guidelines for on board sampling of fuel oil intended to be used or carried for use on board a ship (MEPC.1/Circ.889).

<sup>&</sup>lt;sup>4</sup> The laboratory is to be accredited to ISO/IEC 17025:2017 or an equivalent standard for the performance of the given sulphur content test ISO 8754:2003.

2.3 If the seal of the sample as received has not been broken, the laboratory shall proceed with the verification procedure and shall:

- .1 unseal the sample;
- .2 ensure that the sample is thoroughly homogenized;
- .3 draw two subsamples from the sample; and
- .4 reseal the sample and record the new reseal details on the test record.

2.4 The two subsamples shall be tested in succession, in accordance with the specified test method referred to in regulation 2.1.30 of this Annex. For the purposes of this Part 1 verification procedure, the results of the test analysis shall be referred to as '1A' and '1B':

- .1 results 1A and 1B shall be recorded on the test record in accordance with the requirements of the test method; and
- .2 if the results of 1A and 1B are within the repeatability (r)<sup>5</sup> of the test method, the results shall be considered valid; or
- .3 if the results 1A and 1B are not within the repeatability (r) of the test method, both results shall be rejected and two new subsamples shall be taken by the laboratory and tested. The sample bottle shall be resealed in accordance with paragraph 2.3.4 after the new subsamples have been taken.
- .4 in the case of two failures to achieve repeatability between 1A and 1B, the cause of that failure shall be investigated by the laboratory and resolved before further testing of the sample is undertaken. On resolution of that repeatability issue, two new subsamples shall be taken in accordance with paragraph 2.3. The sample shall be resealed in accordance with paragraph 2.3.4 after the new subsamples have been taken.

2.5 If the test results of 1A and 1B are valid, an average of these two results shall be calculated. The average value shall be referred to as 'X' and shall be recorded on the test record:

- .1 if the result X is equal to or less than the applicable limit required by regulation 14, the fuel oil shall be considered to have met the requirement; or
- .2 if the result X is greater than the applicable limit required by regulation 14, the fuel oil shall be considered to have not met the requirement.

<sup>&</sup>lt;sup>5</sup> Repeatability (r) calculation in accordance with ISO 4259:2017-2 and as defined in the test method used.

On the basis of the test method referred to in regulation 2.1.30 of this Annex							
Applicable limit % m/m: VResult 2.5.1: X ≤ VResult 2.5.2: X > V							
0.10	0.10 Met the requirement Not met the requirement						
0.50	<u> </u>						
	Result X reported to 2 decimal places						

## Table 1: Summary of Part 1 MARPOL delivered sample procedure

2.6 The final results obtained from this verification procedure shall be evaluated by the competent authority.

2.7 The laboratory shall provide a copy of the test record to the competent authority managing the verification procedure.

## Part 2 – In-use and onboard samples

## 3 General Requirements

3.1 The in-use or onboard sample, as appropriate, shall be used to verify the sulphur content of the fuel oil as represented by that sample of fuel oil at the point of sampling.

3.2 A Party, through its competent authority, shall manage the verification procedure.

3.3 A laboratory undertaking the sulphur testing procedure given in this appendix shall have valid accreditation<sup>6</sup> in respect of the test method to be used.

## 4 Verification Procedure Part 2

4.1 The in-use or onboard sample shall be conveyed by the competent authority to the laboratory.

- 4.2 The laboratory shall:
  - .1 record the details of the seal number and the sample label on the test record;
  - .2 record the condition of the seal of the sample as received on the test record; and
  - .3 reject any sample where the seal has been broken prior to receipt and record that rejection on the test record.

4.3 If the seal of the sample as received has not been broken, the laboratory shall proceed with the verification procedure and shall:

- .1 unseal the sample;
- .2 ensure that the sample is thoroughly homogenized;
- .3 draw two subsamples from the sample; and

<sup>&</sup>lt;sup>6</sup> The laboratory is to be accredited to ISO/IEC 17025:2017 or an equivalent standard for the performance of the given sulphur content test ISO 8754:2003.

.4 reseal the sample and record the new reseal details on the test record.

4.4 The two subsamples shall be tested in succession, in accordance with the specified test method referred to in regulation 2.1.30 of this Annex. For the purposes of this Part 2 verification procedure, the results obtained shall be referred to as '2A' and '2B':

- .1 results 2A and 2B shall be recorded on the test record in accordance with the requirements of the test method; and
- .2 if the results of 2A and 2B are within the repeatability (r)<sup>7</sup> of the test method, the results shall be considered valid; or
- .3 if the results of 2A and 2B are not within the repeatability (r) of the test method, both results shall be rejected and two new subsamples shall be taken by the laboratory and tested. The sample bottle shall be resealed in accordance with paragraph 4.3.4 after the new subsamples have been taken; and
- .4 in the case of two failures to achieve repeatability between 2A and 2B, the cause of that failure shall be investigated by the laboratory and resolved before further testing of the sample is undertaken. On resolution of that repeatability issue, two new subsamples shall be taken in accordance with paragraph 4.3. The sample shall be resealed in accordance with paragraph 4.3.4 after the new subsamples have been taken.

4.5 If the test results of 2A and 2B are valid, an average of these two results shall be calculated. That average value shall be referred to as 'Z' and shall be recorded on the test record:

- .1 if Z is equal to or less than the applicable limit required by regulation 14, the sulphur content of the fuel oil as represented by the tested sample shall be considered to have met the requirement;
- .2 if Z is greater than the applicable limit required by regulation 14 but less than or equal to that applicable limit + 0.59R (where R is the reproducibility of the test method),<sup>8</sup> the sulphur content of the fuel oil as represented by the tested sample shall be considered to have met the requirement; or
- .3 if Z is greater than the applicable limit required by regulation 14 + 0.59R, the sulphur content of the fuel oil as represented by the tested sample shall be considered to have not met the requirement.

<sup>&</sup>lt;sup>7</sup> Repeatability (r) calculation in accordance with ISO 4259:2017-2 and as defined in the test method used.

<sup>&</sup>lt;sup>8</sup> Reproducibility (R) calculation in accordance with ISO 4259:2017-2 and as defined in the test method used.

On the basis of the test method referred to in regulation 2.1.30 of this Annex							
Applicable limit %m/m: V	Test margin value: W	Result 4.5.1: Z ≤ V	Result 4.5.2: V < Z ≤ W	Result 4.5.3: Z > W			
0.10	0.11	Met the	Met the	Not met the			
0.50	0.53	requirement	requirement	requirement			
		Result Z reported to 2 decimal places					

## Table 2: Summary of in-use or onboard sample procedure<sup>9</sup>

4.6 The final results obtained from this verification procedure shall be evaluated by the competent authority.

4.7 The laboratory shall provide a copy of the test record to the competent authority managing the verification procedure.

<sup>&</sup>lt;sup>9</sup> Results of testing undertaken by the company or other entities are outside the MARPOL process and hence should be considered within the approach given by ISO 4259:2017-2 regarding recipient drawn samples.
# Appendix VII

# Emission control areas (regulations 13.6 and 14.3)

1 The boundaries of emission control areas designated under regulations 13.6 and 14.3, other than the Baltic Sea and the North Sea areas, are set forth in this appendix.

- 2 The North American area comprises:
  - .1 the sea area located off the Pacific coasts of the United States and Canada, enclosed by geodesic lines connecting the following coordinates:

Point	Latitude	Longitude
1	32°32′.10 N	117°06′.11 W
2	32°32′.04 N	117°07′.29 W
3	32°31′.39 N	117°14′.20 W
4	32°33′.13 N	117°15′.50 W
5	32°34′.21 N	117°22′.01 W
6	32°35′.23 N	117°27′.53 W
7	32°37′.38 N	117°49′.34 W
8	31°07′.59 N	118°36′.21 W
9	30°33′.25 N	121°47′.29 W
10	31°46′.11 N	123°17′.22 W
11	32°21′.58 N	123°50′.44 W
12	32°56′.39 N	124°11′.47 W
13	33°40′.12 N	124°27′.15 W
14	34°31′.28 N	125°16′.52 W
15	35°14′.38 N	125°43′.23 W
16	35°44′.00 N	126°18′.53 W
17	36°16′.25 N	126°45′.30 W
18	37°01′.35 N	127°07′.18 W
19	37°45′.39 N	127°38′.02 W
20	38°25′.08 N	127°53′.00 W
21	39°25′.05 N	128°31′.23 W
22	40°18′.47 N	128°45′.46 W
23	41°13′.39 N	128°40′.22 W
24	42°12′.49 N	129°00′.38 W
25	42°47′.34 N	129°05′.42 W
26	43°26′.22 N	129°01′.26 W
27	44°24′.43 N	128°41′.23 W
28	45°30′.43 N	128°40′.02 W
29	46°11′.01 N	128°49′.01 W
30	46°33′.55 N	129°04′.29 W
31	47°39′.55 N	131°15′.41 W
32	48°32′.32 N	132°41′.00 W
33	48°57′.47 N	133°14′.47 W

Point	Latitude	Longitude
34	49°22′.39 N	134°15′.51 W
35	50°01′.52 N	135°19′.01 W
36	51°03′.18 N	136°45′.45 W
37	51°54′.04 N	137°41′.54 W
38	52°45′.12 N	138°20′.14 W
39	53°29′.20 N	138°40′.36 W
40	53°40′.39 N	138°48′.53 W
41	54°13′.45 N	139°32′.38 W
42	54°39′.25 N	139°56′.19 W
43	55°20′.18 N	140°55′.45 W
44	56°07′.12 N	141°36′.18 W
45	56°28′.32 N	142°17′.19 W
46	56°37′.19 N	142°48′.57 W
47	58°51′.04 N	153°15′.03 W

.2 the sea areas located off the Atlantic coasts of the United States, Canada and France (Saint- Pierre-et-Miquelon), and the Gulf of Mexico coast of the United States enclosed by geodesic lines connecting the following coordinates:

Point	Latitude	Longitude
1	60°00′.00 N	64°09′.36 W
2	60°00′.00 N	56°43′.00 W
3	58°54′.01 N	55°38′.05 W
4	57°50′.52 N	55°03′.47 W
5	57°35′.13 N	54°00′.59 W
6	57°14′.20 N	53°07′.58 W
7	56°48′.09 N	52°23′.29 W
8	56°18′.13 N	51°49′.42 W
9	54°23′.21 N	50°17′.44 W
10	53°44′.54 N	50°07′.17 W
11	53°04′.59 N	50°10′.05 W
12	52°20′.06 N	49°57′.09 W
13	51°34′.20 N	48°52′.45 W
14	50°40′.15 N	48°16′.04 W
15	50°02′.28 N	48°07′.03 W
16	49°24′.03 N	48°09′.35 W
17	48°39′.22 N	47°55′.17 W
18	47°24′.25 N	47°46′.56 W
19	46°35′.12 N	48°00′.54 W
20	45°19′.45 N	48°43′.28 W
21	44°43′.38 N	49°16′.50 W

Point	Latitude	Longitude
22	44°16′.38 N	49°51′.23 W
23	43°53′.15 N	50°34′.01 W
24	43°36′.06 N	51°20′.41 W
25	43°23′.59 N	52°17′.22 W
26	43°19′.50 N	53°20′.13 W
27	43°21′.14 N	54°09′.20 W
28	43°29′.41 N	55°07′.41 W
29	42°40′.12 N	55°31′.44 W
30	41°58′.19 N	56°09′.34 W
31	41°20′.21 N	57°05′.13 W
32	40°55′.34 N	58°02′.55 W
33	40°41′.38 N	59°05′.18 W
34	40°38′.33 N	60°12′.20 W
35	40°45′.46 N	61°14′.03 W
36	41°04′.52 N	62°17′.49 W
37	40°36′.55 N	63°10′.49 W
38	40°17′.32 N	64°08′.37 W
39	40°07′.46 N	64°59′.31 W
40	40°05′.44 N	65°53′.07 W
41	39°58′.05 N	65°59′.51 W
42	39°28′.24 N	66°21′.14 W
43	39°01′.54 N	66°48′.33 W
44	38°39′.16 N	67°20′.59 W
45	38°19′.20 N	68°02′.01 W
46	38°05′.29 N	68°46′.55 W
47	37°58′.14 N	69°34′.07 W
48	37°57′.47 N	70°24′.09 W
49	37°52′.46 N	70°37′.50 W
50	37°18′.37 N	71°08′.33 W
51	36°32′.25 N	71°33′.59 W
52	35°34′.58 N	71°26′.02 W
53	34°33′.10 N	71°37′.04 W
54	33°54′.49 N	71°52′.35 W
55	33°19′.23 N	72°17′.12 W
56	32°45′.31 N	72°54′.05 W
57	31°55′.13 N	74°12′.02 W
58	31°27′.14 N	75°15′.20 W
59	31°03′.16 N	75°51′.18 W
60	30°45′.42 N	76°31′.38 W
61	30°12′.48 N	77°18′.29 W
62	29°25′.17 N	76°56′.42 W

Point	Latitude	Longitude
63	28°36′.59 N	76°48′.00 W
64	28°17′.13 N	76°40′.10 W
65	28°17′.12 N	79°11′.23 W
66	27°52′.56 N	79°28′.35 W
67	27°26′.01 N	79°31′.38 W
68	27°16′.13 N	79°34′.18 W
69	27°11′.54 N	79°34′.56 W
70	27°05′.59 N	79°35′.19 W
71	27°00′.28 N	79°35′.17 W
72	26°55′.16 N	79°34′.39 W
73	26°53′.58 N	79°34′.27 W
74	26°45′.46 N	79°32′.41 W
75	26°44′.30 N	79°32′.23 W
76	26°43′.40 N	79°32′.20 W
77	26°41′.12 N	79°32′.01 W
78	26°38′.13 N	79°31′.32 W
79	26°36′.30 N	79°31′.06 W
80	26°35′.21 N	79°30′.50 W
81	26°34′.51 N	79°30′.46 W
82	26°34′.11 N	79°30′.38 W
83	26°31′.12 N	79°30′.15 W
84	26°29′.05 N	79°29′.53 W
85	26°25′.31 N	79°29′.58 W
86	26°23′.29 N	79°29′.55 W
87	26°23′.21 N	79°29′.54 W
88	26°18′.57 N	79°31′.55 W
89	26°15′.26 N	79°33′.17 W
90	26°15′.13 N	79°33′.23 W
91	26°08′.09 N	79°35′.53 W
92	26°07′.47 N	79°36′.09 W
93	26°06′.59 N	79°36′.35 W
94	26°02′.52 N	79°38′.22 W
95	25°59′.30 N	79°40′.03 W
96	25°59′.16 N	79°40′.08 W
97	25°57′.48 N	79°40′.38 W
98	25°56′.18 N	79°41′.06 W
99	25°54′.04 N	79°41′.38 W
100	25°53′.24 N	79°41′.46 W
101	25°51′.54 N	79°41′.59 W
102	25°49′.33 N	79°42′.16 W
103	25°48′.24 N	79°42′.23 W

Point	Latitude	Longitude
104	25°48′.20 N	79°42′.24 W
105	25°46′.26 N	79°42′.44 W
106	25°46′.16 N	79°42′.45 W
107	25°43′.40 N	79°42′.59 W
108	25°42′.31 N	79°42′.48 W
109	25°40′.37 N	79°42′.27 W
110	25°37′.24 N	79°42′.27 W
111	25°37′.08 N	79°42′.27 W
112	25°31′.03 N	79°42′.12 W
113	25°27′.59 N	79°42′.11 W
114	25°24′.04 N	79°42′.12 W
115	25°22′.21 N	79°42′.20 W
116	25°21′.29 N	79°42′.08 W
117	25°16′.52 N	79°41′.24 W
118	25°15′.57 N	79°41′.31 W
119	25°10′.39 N	79°41′.31 W
120	25°09′.51 N	79°41′.36 W
121	25°09′.03 N	79°41′.45 W
122	25°03′.55 N	79°42′.29 W
123	25°03′.00 N	79°42′.56 W
124	25°00′.30 N	79°44′.05 W
125	24°59′.03 N	79°44′.48 W
126	24°55′.28 N	79°45′.57 W
127	24°44′.18 N	79°49′.24 W
128	24°43′.04 N	79°49′.38 W
129	24°42′.36 N	79°50′.50 W
130	24°41′.47 N	79°52′.57 W
131	24°38′.32 N	79°59′.58 W
132	24°36′.27 N	80°03′.51 W
133	24°33′.18 N	80°12′.43 W
134	24°33′.05 N	80°13′.21 W
135	24°32′.13 N	80°15′.16 W
136	24°31′.27 N	80°16′.55 W
137	24°30′.57 N	80°17′.47 W
138	24°30′.14 N	80°19′.21 W
139	24°30′.06 N	80°19′.44 W
140	24°29′.38 N	80°21′.05 W
141	24°28′.18 N	80°24′.35 W
142	24°28′.06 N	80°25′.10 W
143	24°27′.23 N	80°27′.20 W
144	24°26′.30 N	80°29′.30 W

Point	Latitude	Longitude
145	24°25′.07 N	80°32′.22 W
146	24°23′.30 N	80°36′.09 W
147	24°22′.33 N	80°38′.56 W
148	24°22′.07 N	80°39′.51 W
149	24°19′.31 N	80°45′.21 W
150	24°19′.16 N	80°45′.47 W
151	24°18′.38 N	80°46′.49 W
152	24°18′.35 N	80°46′.54 W
153	24°09′.51 N	80°59′.47 W
154	24°09′.48 N	80°59′.51 W
155	24°08′.58 N	81°01′.07 W
156	24°08′.30 N	81°01′.51 W
157	24°08′.26 N	81°01′.57 W
158	24°07′.28 N	81°03′.06 W
159	24°02′.20 N	81°09′.05 W
160	24°00′.00 N	81°11′.16 W
161	23°55′.32 N	81°12′.55 W
162	23°53′.52 N	81°19′.43 W
163	23°50′.52 N	81°29′.59 W
164	23°50′.02 N	81°39′.59 W
165	23°49′.05 N	81°49′.59 W
166	23°49′.05 N	82°00′.11 W
167	23°49′.42 N	82°09′.59 W
168	23°51′.14 N	82°24′.59 W
169	23°51′.14 N	82°39′.59 W
170	23°49′.42 N	82°48′.53 W
171	23°49′.32 N	82°51′.11 W
172	23°49′.24 N	82°59′.59 W
173	23°49′.52 N	83°14′.59 W
174	23°51′.22 N	83°25′.49 W
175	23°52′.27 N	83°33′.01 W
176	23°54′.04 N	83°41′.35 W
177	23°55′.47 N	83°48′.11 W
178	23°58′.38 N	83°59′.59 W
179	24°09′.37 N	84°29′.27 W
180	24°13′.20 N	84°38′.39 W
181	24°16′.41 N	84°46′.07 W
182	24°23′.30 N	84°59′.59 W
183	24°26′.37 N	85°06′.19 W
184	24°38′.57 N	85°31′.54 W
185	24°44′.17 N	85°43′.11 W

Point	Latitude	Longitude
186	24°53′.57 N	85°59′.59 W
187	25°10′.44 N	86°30′.07 W
188	25°43′.15 N	86°21′.14 W
189	26°13′.13 N	86°06′.45 W
190	26°27′.22 N	86°13′.15 W
191	26°33′.46 N	86°37′.07 W
192	26°01′.24 N	87°29′.35 W
193	25°42′.25 N	88°33′.00 W
194	25°46′.54 N	90°29′.41 W
195	25°44′.39 N	90°47′.05 W
196	25°51′.43 N	91°52′.50 W
197	26°17′.44 N	93°03′.59 W
198	25°59′.55 N	93°33′.52 W
199	26°00′.32 N	95°39′.27 W
200	26°00′.33 N	96°48′.30 W
201	25°58′.32 N	96°55′.28 W
202	25°58′.15 N	96°58′.41 W
203	25°57′.58 N	97°01′.54 W
204	25°57′.41 N	97°05′.08 W
205	25°57′.24 N	97°08′.21 W
206	25°57′.24 N	97°08′.47 W

.3 the sea area located off the coasts of the Hawaiian Islands of Hawai'i, Maui, Oahu, Meloka'i, Ni'ihau, Kaua'i, Lana'i and Kaho'olawe, enclosed by geodesic lines connecting the following coordinates:

Point	Latitude	Longitude
1	22°32′.54 N	153°00′.33 W
2	23°06′.05 N	153°28′.36 W
3	23°32′.11 N	154°02′.12 W
4	23°51′.47 N	154°36′.48 W
5	24°21′.49 N	155°51′.13 W
6	24°41′.47 N	156°27′.27 W
7	24°57′.33 N	157°22′.17 W
8	25°13′.41 N	157°54′.13 W
9	25°25′.31 N	158°30′.36 W
10	25°31′.19 N	159°09′.47 W
11	25°30′.31 N	159°54′.21 W
12	25°21′.53 N	160°39′.53 W
13	25°00′.06 N	161°38′.33 W
14	24°40′.49 N	162°13′.13 W
15	24°15′.53 N	162°43′.08 W

Point	Latitude	Longitude
16	23°40′.50 N	163°13′.00 W
17	23°03′.20 N	163°32′.58 W
18	22°20′.09 N	163°44′.41 W
19	21°36′.45 N	163°46′.03 W
20	20°55′.26 N	163°37′.44 W
21	20°13′.34 N	163°19′.13 W
22	19°39′.03 N	162°53′.48 W
23	19°09′.43 N	162°20′.35 W
24	18°39′.16 N	161°19′.14 W
25	18°30′.31 N	160°38′.30 W
26	18°29′.31 N	159°56′.17 W
27	18°10′.41 N	159°14′.08 W
28	17°31′.17 N	158°56′.55 W
29	16°54′.06 N	158°30′.29 W
30	16°25′.49 N	157°59′.25 W
31	15°59′.57 N	157°17′.35 W
32	15°40′.37 N	156°21′.06 W
33	15°37′.36 N	155°22′.16 W
34	15°43′.46 N	154°46′.37 W
35	15°55′.32 N	154°13′.05 W
36	16°46′.27 N	152°49′.11 W
37	17°33′.42 N	152°00′.32 W
38	18°30′.16 N	151°30′.24 W
39	19°02′.47 N	151°22′.17 W
40	19°34′.46 N	151°19′.47 W
41	20°07′.42 N	151°22′.58 W
42	20°38′.43 N	151°31′.36 W
43	21°29′.09 N	151°59′.50 W
44	22°06′.58 N	152°31′.25 W
45	22°32′.54 N	153°00′.33 W

- 3 The United States Caribbean Sea area includes:
  - .1 the sea area located off the Atlantic and Caribbean coasts of the Commonwealth of Puerto Rico and the United States Virgin Islands, enclosed by geodesic lines connecting the following coordinates:

Point	Latitude	Longitude
1	17°18′.37 N	67°32′.14 W
2	19°11′.14 N	67°26′.45 W
3	19°30′.28 N	65°16′.48 W
4	19°12′.25 N	65°06′.08 W
5	18°45′.13 N	65°00′.22 W
6	18°41′.14 N	64°59′.33 W

Point	Latitude	Longitude
7	18°29′.22 N	64°53′.51 W
8	18°27′.35 N	64°53′.22 W
9	18°25′.21 N	64°52′.39 W
10	18°24′.30 N	64°52′.19 W
11	18°23′.51 N	64°51′.50 W
12	18°23′.42 N	64°51′.23 W
13	18°23′.36 N	64°50′.17 W
14	18°23′.48 N	64°49′.41 W
15	18°24′.11 N	64°49′.00 W
16	18°24′.28 N	64°47′.57 W
17	18°24′.18 N	64°47′.01 W
18	18°23′.13 N	64°46′.37 W
19	18°22′.37 N	64°45′.20 W
20	18°22′.39 N	64°44′.42 W
21	18°22′.42 N	64°44′.36 W
22	18°22′.37 N	64°44′.24 W
23	18°22′.39 N	64°43′.42 W
24	18°22′.30 N	64°43′.36 W
25	18°22′.25 N	64°42′.58 W
26	18°22′.26 N	64°42′.28 W
27	18°22′.15 N	64°42′.03 W
28	18°22′.22 N	64°40′.60 W
29	18°21′.57 N	64°40′.15 W
30	18°21′.51 N	64°38′.23 W
31	18°21′.22 N	64°38′.16 W
32	18°20′.39 N	64°38′.33 W
33	18°19′.15 N	64°38′.14 W
34	18°19′.07 N	64°38′.16 W
35	18°17′.23 N	64°39′.38 W
36	18°16′.43 N	64°39′.41 W
37	18°11′.33 N	64°38′.58 W
38	18°03′.02 N	64°38′.03 W
39	18°02′.56 N	64°29′.35 W
40	18°02′.51 N	64°27′.02 W
41	18°02′.30 N	64°21′.08 W
42	18°02′.31 N	64°20′.08 W
43	18°02′.03 N	64°15′.57 W
44	18°00′.12 N	64°02′.29 W
45	17°59′.58 N	64°01′.04 W
46	17°58′.47 N	63°57′.01 W
47	17°57′.51 N	63°53′.54 W
48	17°56′.38 N	63°53′.21 W
49	17°39′.40 N	63°54′.53 W

Point	Latitude	Longitude	
50	17°37′.08 N	63°55′.10 W	
51	17°30′.21 N	63°55′.56 W	
52	17°11′.36 N	63°57′.57 W	
53	17°05′.00 N	63°58′.41 W	
54	16°59′.49 N	63°59′.18 W	
55	17°18′.37 N	67°32′.14 W	

# Appendix VIII

# Form of International Energy Efficiency (IEE) Certificate

# INTERNATIONAL ENERGY EFFICIENCY CERTIFICATE

Issued under the provisions of the Protocol of 1997, as amended, to amend the International Convention for the Prevention of Pollution from Ships, 1973, as modified by the Protocol of 1978 relating thereto (hereinafter referred to as "the Convention") under the authority of the Government of:

(full designation of the country)

by .....

(full designation of the competent person or organization authorized under the provisions of the Convention)

### Particulars of ship<sup>1</sup>

Name of ship
Distinctive number or letters
Port of registry
Gross tonnage
IMO Number <sup>2</sup>
THIS IS TO CERTIFY:

- 1 That the ship has been surveyed in accordance with regulation 5.4 of Annex VI to the Convention; and
- 2 That the survey shows that the ship complies with the applicable requirements in regulations 22, 23, 24, 25 and 26.

Issued at .....

(place of issue of certificate)

(dd/mm/yyyy): ..... (date of issue)

(signature of duly authorized official issuing the certificate)

(seal or stamp of the authority, as appropriate)

<sup>&</sup>lt;sup>1</sup> Alternatively, the particulars of the ship may be placed horizontally in boxes.

<sup>&</sup>lt;sup>2</sup> In accordance with the IMO Ship Identification Number Scheme (resolution A.1117(30)).

# Supplement to the International Energy Efficiency Certificate (IEE Certificate)

# RECORD OF CONSTRUCTION RELATING TO ENERGY EFFICIENCY

Notes:

- 1 This Record shall be permanently attached to the IEE Certificate. The IEE Certificate shall be available on board the ship at all times.
- 2 The Record shall be at least in English, French or Spanish. If an official language of the issuing Party is also used, this shall prevail in case of a dispute or discrepancy.
- 3 Entries in boxes shall be made by inserting either: a cross (x) for the answers "yes" and "applicable"; or a dash (-) for the answers "no" and "not applicable", as appropriate.
- 4 Unless otherwise stated, regulations mentioned in this Record refer to regulations in Annex VI of the Convention, and resolutions or circulars refer to those adopted by the International Maritime Organization.

### 1 Particulars of ship

1.1	Name of ship	
1.2	IMO Number	
1.3	Date of building contract	
1.4	Date of major conversion (if applicable)	
1.5	Gross tonnage	
1.6	Deadweight	
1.7	Type of ship <sup>3</sup>	
2	Propulsion system	
<b>2</b> 2.1	Propulsion system Diesel propulsion	
<b>2</b> 2.1 2.2	Propulsion system Diesel propulsion Diesel-electric propulsion	
<b>2</b> 2.1 2.2 2.3	Propulsion system         Diesel propulsion         Diesel-electric propulsion         Turbine propulsion	
<b>2</b> 2.1 2.2 2.3 2.4	Propulsion system         Diesel propulsion         Diesel-electric propulsion         Turbine propulsion         Hybrid propulsion	
<b>2</b> 2.1 2.2 2.3 2.4 2.5	Propulsion system         Diesel propulsion         Diesel-electric propulsion         Turbine propulsion         Hybrid propulsion         Propulsion system other than any of the above	

<sup>&</sup>lt;sup>3</sup> Insert ship type in accordance with definitions specified in regulation 2. Ships falling into more than one of the ship types defined in regulation 2 should be considered as being the ship type with the most stringent (the lowest) required EEDI. If the ship does not fall into the ship types defined in regulation 2, insert "Ship other than ship types defined in regulation 2".

3	Attained Energy Efficiency Design Index (EEDI)		
3.1	The attained EEDI in accordance with regulation 22.1 is calculated based on the information contained in the EEDI technical file, which also shows the process calculating the attained EEDI		
	The attained EEDI is: grams-CO <sub>2</sub> /tonne-nautical mile		
3.2	The attained EEDI is not calculated, as:		
3.2.1	the ship is exempt under regulation 22.1 as it is not a new ship as defined in regulation 2.2.18 $\hfill \square$		
3.2.2	the type of propulsion system is exempt in accordance with regulation 19.3		
3.2.3	the requirement of regulation 22 is waived by the ship's Administration in accordance with regulation 19.4 $\dots$ $\square$		
3.2.4	the type of ship is exempt in accordance with regulation 22.1 $\ldots$		
4	Required EEDI		
4.1	Required EEDI is: grams-CO <sub>2</sub> /tonne-mile		
4.2	The required EEDI is not applicable, as:		
4.2.1	the ship is exempt under regulation 24.1 as it is not a new ship as defined in regulation 2.2.18		
4.2.2	the type of propulsion system is exempt in accordance with regulation 19.3		
4.2.3	the requirement of regulation 24 is waived by the ship's Administration in accordance with regulation 19.4 $\hfill\square$		
4.2.4	the type of ship is exempt in accordance with regulation 24.1 $\square$		
4.2.5	the ship's capacity is below the minimum capacity threshold in table 1 of regulation 24.2 $\hfill\square$		
5	Attained Energy Efficiency Existing Ship Index (EEXI)		
5.1	The attained EEXI in accordance with regulation 23.1 is calculated taking into account the guidelines <sup>4</sup> developed by the Organization		
	The attained EEXI is:grams-CO <sub>2</sub> /tonne-mile		
5.2	The attained EEXI is not calculated, as:		
5.2.1	the type of propulsion system is exempt in accordance with regulation 19.3 $\Box$		
5.2.2	the type of ship is exempt in accordance with regulation 23.1 $\Box$		

<sup>&</sup>lt;sup>4</sup> Refer to the 2021 Guidelines on the method of calculation of the attained Energy Efficiency Existing Ship Index (EEXI) (resolution MEPC.333(76))

#### 6 Required EEXI

- 6.1 The required EEXI is:.....grams-CO<sub>2</sub>/tonne-mile in accordance with regulation 25
- 6.2 The required EEXI is not applicable, as:
- 6.2.1 the type of propulsion system is exempt in accordance with regulation 19.3......
- 6.2.2 the type of ship is exempt in accordance with regulation 25.1.....  $\hfill\square$
- 6.2.3 the ship's capacity is below the minimum capacity threshold in table 3 of regulation 25.1.

## 7 Ship Energy Efficiency Management Plan

7.1 The ship is provided with a Ship Energy Efficiency Management Plan (SEEMP) in compliance with regulation 26.....

#### 8 EEDI technical file

8.1 The IEE Certificate is accompanied by the EEDI technical file in compliance with regulation 22.1.....

# 8.1.1 The EEDI technical file identification/verification number.....

8.1.2 The EEDI technical file verification date.....

### 9 EEXI technical file

- 9.1 The IEE Certificate is accompanied by the EEXI technical file in compliance with regulation 23.1.....
- 9.1.1 The EEXI technical file identification/verification number.....
- 9.1.2 The EEXI technical file verification date.....

THIS IS TO CERTIFY that this Record is correct in all respects.

(dd/mm/yyyy): ..... (date of issue)

(signature of duly authorized official issuing the Record)

(seal or stamp of the authority, as appropriate)

# Appendix IX

#### Information to be submitted to the IMO Ship Fuel Oil Consumption Database

Identity of the ship
IMO Number
Period of calendar year for which the data is submitted
Start date (dd/mm/yyyy)
End date (dd/mm/yyyy)
Technical characteristics of the ship
Ship type, as defined in regulation 2 of this Annex or other (to be stated)
Gross tonnage (GT) <sup>1</sup>
Net tonnage (NT) <sup>2</sup>
Deadweight tonnage (DWT) <sup>3</sup>
Power output (rated power) <sup>4</sup> of main and auxiliary reciprocating internal combustion engines over 130 kW (to be stated in kW)
EEDI (if applicable)
Ice class <sup>5</sup>
Fuel oil consumption, by fuel oil type <sup>6</sup> in metric tonnes and methods used for collecting fuel oil consumption data
Distance travelled
Hours under way

<sup>&</sup>lt;sup>1</sup> Gross tonnage should be calculated in accordance with the International Convention on Tonnage Measurement of Ships, 1969.

<sup>&</sup>lt;sup>2</sup> Net tonnage should be calculated in accordance with the International Convention on Tonnage Measurement of Ships, 1969. If not applicable, note "N/A".

<sup>&</sup>lt;sup>3</sup> DWT means the difference in tonnes between the displacement of a ship in water of relative density of 1,025 kg/m3 at the summer load draught and the lightweight of the ship. The summer load draught should be taken as the maximum summer draught as certified in the stability booklet approved by the Administration or an organization aby it. If not applicable, note "N/A".

<sup>&</sup>lt;sup>4</sup> Rated power means the maximum continuous rated power as specified on the nameplate of the engine.

<sup>&</sup>lt;sup>5</sup> Ice class should be consistent with the definition set out in the International Code for Ships Operating in Polar Waters (Polar Code) (resolutions MEPC.264(68) and MSC.385(94)). If not applicable, note "N/A".

<sup>&</sup>lt;sup>6</sup> Refer to the 2018 Guidelines on the method of calculation of the attained Energy Efficiency Design Index (EEDI) for new ships (resolution MEPC.308(73), as amended by resolutions MEPC.322(74) and MEPC.332(76)).

Appendix X

### Form of Statement of Compliance – Fuel Oil Consumption Reporting and Operational Carbon Intensity rating

# STATEMENT OF COMPLIANCE – FUEL OIL CONSUMPTION REPORTING AND OPERATIONAL CARBON INTENSITY RATING

Issued under the provisions of the Protocol of 1997, as amended, to amend the International Convention for the Prevention of Pollution from Ships, 1973, as modified by the Protocol of 1978 relating thereto (hereinafter referred to as "the Convention") under the authority of the Government of:

(full designation of the	e country)

by.....(full designation of the competent person or organization authorized under the provisions of the Convention)

# Particulars of ship<sup>1</sup>

Name of ship
Distinctive number or letters
IMO Number <sup>2</sup>
Port of registry
Gross tonnage
Deadweight
Type of ship

### THIS IS TO DECLARE THAT:

- 1 the ship has submitted to this Administration the data required by regulation 27 of Annex VI to the Convention, covering ship operations from (dd/mm/yyyy) to (dd/mm/yyyy);
- 2 the data was collected and reported in accordance with the methodology and processes set out in the ship's SEEMP that was in effect over the period from (dd/mm/yyyy) to (dd/mm/yyyy);

<sup>&</sup>lt;sup>1</sup> Alternatively, the particulars of the ship may be placed horizontally in boxes.

<sup>&</sup>lt;sup>2</sup> In accordance with the IMO Ship Identification Number Scheme (resolution A.1117(30)).

- 3 the attained annual operational CII of the ship from (dd/mm/yyyy) through (dd/mm/yyyy) was: ..... pursuant to regulations 28.1 and 28.2 of Annex VI of the Convention, for ships to which regulation 28 applies;<sup>3</sup>
- 4 the annual operational carbon intensity of the ship in this period is rated as

 $\Box A \Box B \Box C \Box D \Box E$ 

in accordance with regulation 28 of Annex VI to the Convention, for a ship to which regulation 28 applies<sup>3</sup>; and

5 a corrective action plan has been developed and included in the SEEMP (for a ship to which regulation 28 applies, rated as D for three consecutive years or rated as E)\*

This Statement of Compliance is valid until (dd/mm/yyyy) .....

Issued at			
(place of issue of the Statement)			
(dd/mm/yyyy):			
(date of issue)	(signature of duly authorized official issuing the Statement)		

(seal or stamp of the authority, as appropriate)

<sup>&</sup>lt;sup>3</sup> In the event of any transfer of a ship addressed in regulations 27.4, 27.5 or 27.6, these sections should be completed consistent with regulation 28.3 of MARPOL Annex VI.

Appendix XI

# Form of Exemption Certificate for UNSP Barges

# INTERNATIONAL AIR POLLUTION PREVENTION EXEMPTION CERTIFICATE FOR UNMANNED NON-SELF-PROPELLED (UNSP) BARGES

Issued under the provisions of the Protocol of 1997, as amended, to amend the International Convention for the Prevention of Pollution from Ships, 1973, as modified by the Protocol of 1978 relating thereto (hereinafter referred to as "the Convention") under the authority of the Government of:

(full designation of the country)

# Particulars of ship<sup>1</sup>

Name of ship..... Distinctive number or letters..... IMO Number<sup>2</sup>.... Port of registry..... Gross tonnage....

# THIS IS TO CERTIFY THAT:

1 the UNSP barge has been surveyed in accordance with regulation 3.4 of Annex VI to the Convention;

2 the survey shows that the UNSP barge:

- .1 is not propelled by mechanical means;
- .2 has no system, equipment and/or machinery fitted that may generate emissions controlled by Annex VI to the Convention; and
- .3 has neither persons nor living animals on board; and
- 3 the UNSP barge is exempted, under regulation 3.4 of Annex VI to the Convention from the certification and related survey requirements of regulations 5.1 and 6.1 of Annex VI to the Convention.

<sup>&</sup>lt;sup>1</sup> Alternatively, the particulars of the ship may be placed horizontally in boxes.

<sup>&</sup>lt;sup>2</sup> In accordance with the IMO Ship Identification Number Scheme (resolution A.1117(30)).

This Certificate is valid until (dd/mm/yyyy) .....

subject to the exemption conditions being maintained.

Completion date of the survey on which this Certificate is based (dd/mm/yyyy)

Issued at .....

(place of issue of certificate)

(dd/mm/yyyy): ..... (date of issue)

(signature of duly authorized official issuing the certificate)

(seal or stamp of the authority, as appropriate)



MARINE ENVIRONMENT PROTECTION COMMITTEE 76th session Agenda item 15 MEPC 76/15/Add.2 12 July 2021 Original: ENGLISH

F

# REPORT OF THE MARINE ENVIRONMENT PROTECTION COMMITTEE ON ITS SEVENTY-SIXTH SESSION

Attached are annexes 2 to 20 to the report of the Marine Environment Protection Committee on its seventy-sixth session (MEPC 76/15).

(See document MEPC 76/15/Add.1 for annex 1)



#### LIST OF ANNEXES

- ANNEX 2 RESOLUTION MEPC.329(76) AMENDMENTS TO MARPOL ANNEX I (PROHIBITION ON THE USE AND CARRIAGE FOR USE AS FUEL OF HEAVY FUEL OIL BY SHIPS IN ARCTIC WATERS)
- ANNEX 3 RESOLUTION MEPC.330(76) AMENDMENTS TO MARPOL ANNEXES I AND IV (EXEMPTION OF UNMANNED NON-SELF-PROPELLED BARGES FROM CERTAIN SURVEY AND CERTIFICATION REQUIREMENTS)
- ANNEX 4 RESOLUTION MEPC.331(76) AMENDMENTS TO THE AFS I CONVENTION (CONTROLS ON CYBUTRYNE AND FORM OF THE INTERNATIONAL ANTI-FOULING SYSTEM CERTIFICATE)
- ANNEX 5 RESOLUTION MEPC.332(76) AMENDMENTS TO THE 2018 GUIDELINES ON THE METHOD OF CALCULATION OF THE ATTAINED ENERGY EFFICIENCY DESIGN INDEX (EEDI) FOR NEW SHIPS (RESOLUTION MEPC.308(73), AS AMENDED BY RESOLUTION MEPC.322(74)
- ANNEX 6 UNIFIED INTERPRETATION TO REGULATION 2.23 OF MARPOL ANNEX VI
- ANNEX 7 RESOLUTION MEPC.333(76) 2021 GUIDELINES ON THE METHOD OF CALCULATION OF THE ATTAINED ENERGY EFFICIENCY EXISTING SHIP INDEX (EEXI)
- ANNEX 8 RESOLUTION MEPC.334(76) 2021 GUIDELINES ON SURVEY AND CERTIFICATION OF THE ENERGY EFFICIENCY EXISTING SHIP INDEX (EEXI)
- ANNEX 9 RESOLUTION MEPC.335(76) 2021 GUIDELINES ON THE SHAFT/ENGINE POWER LIMITATION SYSTEM TO COMPLY WITH THE EEXI REQUIREMENTS AND USE OF A POWER RESERVE
- ANNEX 10 RESOLUTION MEPC.336(76) 2021 GUIDELINES ON OPERATIONAL CARBON INTENSITY INDICATORS AND THE CALCULATION METHODS (CII GUIDELINES, G1)
- ANNEX 11 RESOLUTION MEPC.337(76) 2021 GUIDELINES ON THE REFERENCE LINES FOR USE WITH OPERATIONAL CARBON INTENSITY INDICATORS (CII REFERENCE LINES GUIDELINES, G2)
- ANNEX 12 RESOLUTION MEPC.338(76) 2021 GUIDELINES ON THE OPERATIONAL CARBON INTENSITY REDUCTION FACTORS RELATIVE TO REFERENCE LINES (CII REDUCTION FACTOR GUIDELINES, G3)
- ANNEX 13 RESOLUTION MEPC.339(76) 2021 GUIDELINES ON THE OPERATIONAL CARBON INTENSITY RATING OF SHIPS (CII RATING GUIDELINES, G4)
- ANNEX 14 WORK PLAN FOR DEVELOPMENT OF MID- AND LONG-TERM MEASURES AS A FOLLOW-UP OF THE INITIAL IMO STRATEGY ON REDUCTION OF GHG EMISSIONS FROM SHIPS

- ANNEX 15 BIENNIAL AGENDA OF THE PPR SUB-COMMITTEE FOR THE 2022-2023 BIENNIUM
- ANNEX 16 PROVISIONAL AGENDA FOR PPR 9
- ANNEX 17 STATUS REPORT OF THE OUTPUTS OF MEPC FOR THE 2020-2021 BIENNIUM
- ANNEX 18 POST-BIENNIAL AGENDA OF MEPC
- ANNEX 19 ITEMS TO BE INCLUDED IN THE AGENDA OF MEPC 77
- ANNEX 20 STATEMENTS BY DELEGATIONS AND OBSERVERS

\*\*\*

#### RESOLUTION MEPC.329(76) (adopted on 17 June 2021)

#### AMENDMENTS TO THE ANNEX OF THE INTERNATIONAL CONVENTION FOR THE PREVENTION OF POLLUTION FROM SHIPS, 1973, AS MODIFIED BY THE PROTOCOL OF 1978 RELATING THERETO

#### Amendments to MARPOL Annex I

#### (Prohibition on the use and carriage for use as fuel of heavy fuel oil by ships in Arctic waters)

THE MARINE ENVIRONMENT PROTECTION COMMITTEE,

RECALLING Article 38(a) of the Convention on the International Maritime Organization concerning the functions of the Marine Environment Protection Committee conferred upon it by international conventions for the prevention and control of marine pollution from ships,

RECALLING ALSO article 16 of the International Convention for the Prevention of Pollution from Ships, 1973, as modified by the Protocol of 1978 relating thereto (MARPOL), which specifies the amendment procedure and confers upon the appropriate body of the Organization the function of considering and adopting amendments thereto,

HAVING CONSIDERED, at its seventy-sixth session, proposed amendments to MARPOL Annex I concerning the prohibition on the use and carriage for use as fuel of heavy fuel oil by ships in Arctic waters, which were circulated in accordance with article 16(2)(a) of MARPOL,

1 ADOPTS, in accordance with article 16(2)(d) of MARPOL, amendments to MARPOL Annex I, the text of which is set out in the annex to the present resolution;

2 DETERMINES, in accordance with article 16(2)(f)(iii) of MARPOL, that the amendments shall be deemed to have been accepted on 1 May 2022 unless prior to that date not less than one third of the Parties or Parties the combined merchant fleets of which constitute not less than 50% of the gross tonnage of the world's merchant fleet have communicated to the Organization their objection to the amendments;

3 INVITES the Parties to note that, in accordance with article 16(2)(g)(ii) of MARPOL, the said amendments shall enter into force on 1 November 2022 upon their acceptance in accordance with paragraph 2 above;

4 REQUESTS the Secretary-General, for the purposes of article 16(2)(e) of MARPOL, to transmit certified copies of the present resolution and the text of the amendments contained in the annex to all Parties to MARPOL;

5 REQUESTS ALSO the Secretary-General to transmit copies of the present resolution and its annex to Members of the Organization which are not Parties to MARPOL.

# AMENDMENTS TO MARPOL ANNEX I

# (Prohibition on the use and carriage for use as fuel of heavy fuel oil by ships in Arctic waters)

1 The title of chapter 9 is amended as follows:

"Chapter 9 – Special requirements for the use or carriage of oils in polar waters"

2 A new regulation 43A is added in chapter 9 after existing regulation 43, as follows:

#### "Regulation 43A

Special requirements for the use and carriage of oils as fuel in Arctic waters

1 With the exception of ships engaged in securing the safety of ships or in search and rescue operations, and ships dedicated to oil spill preparedness and response, the use and carriage of oils listed in regulation 43.1.2 of this Annex as fuel by ships shall be prohibited in Arctic waters, as defined in regulation 46.2 of this Annex, on or after 1 July 2024.

2 Notwithstanding the provisions of paragraph 1 of this regulation, for ships to which regulation 12A of this Annex or regulation 1.2.1 of chapter 1 of part II-A of the Polar Code applies, the use and carriage of oils listed in regulation 43.1.2 of this Annex as fuel by those ships shall be prohibited in Arctic waters, as defined in regulation 46.2 of this Annex, on or after 1 July 2029.

3 When prior operations have included the use and carriage of oils listed in regulation 43.1.2 of this Annex as fuel, the cleaning or flushing of tanks or pipelines is not required.

4 Notwithstanding the provisions of paragraphs 1 and 2 of this regulation, the Administration of a Party to the present Convention the coastline of which borders on Arctic waters may temporarily waive the requirements of paragraph 1 of this regulation for ships flying the flag of that Party while operating in waters subject to the sovereignty or jurisdiction of that Party, taking into account the guidelines to be developed by the Organization. No waivers issued under this paragraph shall apply on or after 1 July 2029.

5 The Administration of a Party to the present Convention which allows application of paragraph 4 of this regulation shall communicate to the Organization for circulation to the Parties particulars of the waiver thereof, for their information and appropriate action, if any."

\*\*\*

#### RESOLUTION MEPC.330(76) (adopted on 17 June 2021)

#### AMENDMENTS TO THE ANNEX OF THE INTERNATIONAL CONVENTION FOR THE PREVENTION OF POLLUTION FROM SHIPS, 1973, AS MODIFIED BY THE PROTOCOL OF 1978 RELATING THERETO

#### Amendments to MARPOL Annexes I and IV

# (Exemption of unmanned non-self-propelled barges from certain survey and certification requirements)

#### THE MARINE ENVIRONMENT PROTECTION COMMITTEE,

RECALLING Article 38(a) of the Convention on the International Maritime Organization concerning the functions of the Marine Environment Protection Committee conferred upon it by international conventions for the prevention and control of marine pollution from ships,

RECALLING ALSO article 16 of the International Convention for the Prevention of Pollution from Ships, 1973, as modified by the Protocol of 1978 relating thereto (MARPOL), which specifies the amendment procedure and confers upon the appropriate body of the Organization the function of considering and adopting amendments thereto,

HAVING CONSIDERED, at its seventy-sixth session, proposed amendments to MARPOL Annexes I and IV concerning the exemption of unmanned non-self-propelled (UNSP) barges from survey and certification requirements, which were circulated in accordance with article 16(2)(a) of MARPOL,

1 ADOPTS, in accordance with article 16(2)(d) of MARPOL, amendments to MARPOL Annexes I and IV, the text of which is set out in the annex to the present resolution;

2 DETERMINES, in accordance with article 16(2)(f)(iii) of MARPOL, that the amendments shall be deemed to have been accepted on 1 May 2022 unless prior to that date not less than one third of the Parties or Parties the combined merchant fleets of which constitute not less than 50% of the gross tonnage of the world's merchant fleet have communicated to the Organization their objection to the amendments;

3 INVITES the Parties to note that, in accordance with article 16(2)(g)(ii) of MARPOL, the said amendments shall enter into force on 1 November 2022 upon their acceptance in accordance with paragraph 2 above;

4 REQUESTS the Secretary-General, for the purposes of article 16(2)(e) of MARPOL, to transmit certified copies of the present resolution and the text of the amendments contained in the annexes to all Parties to MARPOL;

5 REQUESTS ALSO the Secretary-General to transmit copies of the present resolution and its annexes to Members of the Organization which are not Parties to MARPOL.

### AMENDMENTS TO MARPOL ANNEX I

#### (Exemption of UNSP barges from certain survey and certification requirements)

#### **Regulation 1**

Definitions

1 A new paragraph 40 is added as follows:

"40 *Unmanned non-self-propelled (UNSP) barge* means a barge that:

- .1 is not propelled by mechanical means;
- .2 carries no oil (as defined in regulation 1.1 of this Annex);
- .3 has no machinery fitted that may use oil or generate oil residue (sludge);
- .4 has no oil fuel tank, lubricating oil tank, oily bilge water holding tank and oil residue (sludge) tank; and
- .5 has neither persons nor living animals on board."

#### **Regulation 3**

Exemptions and waivers

2 Paragraph 2 is replaced by the following:

"2 Particulars of any such exemption, except those under paragraph 7 of this regulation, granted by the Administration shall be indicated in the Certificate referred to in regulation 7 of this Annex."

3 A new paragraph 7 is added as follows:

"7 The Administration may exempt a UNSP barge\* from the requirements of regulations 6.1 and 7.1 of this Annex, by an International Oil Pollution Prevention Exemption Certificate for Unmanned Non-self-propelled Barges, for a period not exceeding five years provided that the UNSP barge has undergone a survey to confirm that the conditions referred to in regulations 1.40.1 to 1.40.5 of this Annex are met.

### **Regulation 8**

Issue of endorsement of certificate by another Government

4 Paragraph 4 is replaced by the following:

"4 No International Oil Pollution Prevention Certificate or UNSP Exemption Certificate shall be issued to a ship which is entitled to fly the flag of a State which is not a Party."

<sup>\*</sup> Refer to the Guidelines for exemption of unmanned non-self-propelled barges from certain survey and certification requirements under the MARPOL Convention (MEPC.1/Circ.892)."

# **Regulation 9**

# Form of certificate

5 The existing paragraph is numbered as paragraph 1 and a new paragraph 2 is added as follows:

"2 The International Oil Pollution Prevention Exemption Certificate for Unmanned Non-self-propelled Barges shall be drawn up in the form corresponding to the model given in appendix IV to this Annex and shall be at least in English, French or Spanish. If an official language of the issuing country is also used, this shall prevail in the event of a dispute or discrepancy."

6 A new appendix IV is added as follows:

"APPENDIX IV

#### Form of Exemption Certificate for UNSP Barges

#### INTERNATIONAL OIL POLLUTION PREVENTION EXEMPTION CERTIFICATE FOR UNMANNED NON-SELF-PROPELLED (UNSP) BARGES

Issued under the provisions of the International Convention for the Prevention of Pollution from Ships, 1973, as modified by the Protocol of 1978, as amended, relating thereto (hereinafter referred to as "the Convention") under the authority of the Government of:

(full designation of the country)

#### Particulars of ship\*

Name of ship
Distinctive number or letters
Port of registry
Gross tonnage

### THIS IS TO CERTIFY:

1 That the UNSP barge has been surveyed in accordance with regulation 3.7 of Annex I to the Convention;

- 2 That the survey shows that the UNSP barge:
  - .1 is not propelled by mechanical means;
  - .2 carries no oil (as defined in regulation 1.1 of Annex I to the Convention);
  - .3 has no machinery fitted that may use oil or generate oil residue (sludge);
  - .4 has no oil fuel tank, lubricating oil tank, oily bilge water holding tank and oil residue (sludge) tank; and
  - .5 has neither persons nor living animals on board; and

3 That the UNSP barge is exempted, under regulation 3.7 of Annex I to the Convention, from the certification and related survey requirements of regulations 6.1 and 7.1 of Annex I to the Convention.

This certificate is valid until (*dd/mm/yyyy*).....

subject to the exemption conditions being maintained.

Completion date of the survey on which this certificate is based (*dd/mm/yyyy*).....

<sup>\*</sup> Alternatively, the particulars of the ship may be placed horizontally in boxes.

(date of issue) (dd/mm/yyyy)

(signature of duly authorized official issuing the certificate)

(seal or stamp of the authority, as appropriate)"

# AMENDMENTS TO MARPOL ANNEX IV

### (Exemption of UNSP barges from certain survey and certification requirements)

#### **Regulation 1**

Definitions

1 A new paragraph 16 is added as follows:

"16 Unmanned non-self-propelled (UNSP) barge means a barge that:

- .1 is not propelled by mechanical means;
- .2 has neither persons nor living animals on board;
- .3 is not used for holding sewage during transport; and
- .4 has no arrangements that could produce sewage as defined in regulation 1.3 of this Annex."

# **Regulation 3**

Exceptions

2 The title of the regulation is replaced by the following:

"Exceptions and exemptions"

3 A new paragraph 2 is added as follows:

"2 The Administration may exempt an unmanned non-self-propelled (UNSP) barge\* from the requirements of regulations 4.1 and 5.1 of this Annex, by an International Sewage Pollution Prevention Exemption Certificate for Unmanned Non-self-propelled (UNSP) Barges, for a period not exceeding 5 years provided that the barge has undergone a survey to confirm that the conditions referred to in regulations 1.16.1 to 1.16.4 of this Annex are met."

\* Refer to the Guidelines for exemption of unmanned non-self-propelled barges from the survey and certification requirements under the MARPOL Convention (MEPC.1/Circ.892).

#### **Regulation 6**

Issue of endorsement of a Certificate by another Government

4 Paragraph 4 is replaced by the following:

"4 No International Sewage Pollution Prevention Certificate or UNSP Exemption Certificate shall be issued to a ship which is entitled to fly the flag of a State which is not a Party."

#### **Regulation 7**

Form of Certificate

5 The existing paragraph is numbered as paragraph 1 and the reference to "appendix" is replaced by "appendix 1".

6 A new paragraph 2 is added as follows:

"2 The International Sewage Pollution Prevention Exemption Certificate for Unmanned Non-self-propelled (UNSP) Barges shall be drawn up in the form corresponding to the model given in appendix II to this Annex and shall be at least in English, French or Spanish. If an official language of the issuing country is also used, this shall prevail in the event of a dispute or discrepancy."

# Appendices

7 The existing appendix is numbered as appendix I and a new appendix II is added as follows:

# "APPENDIX II

#### Form of Exemption Certificate for UNSP Barges

#### INTERNATIONAL SEWAGE POLLUTION PREVENTION EXEMPTION CERTIFICATE FOR UNMANNED NON-SELF-PROPELLED (UNSP) BARGES

Issued under the provisions of the International Convention for the Prevention of Pollution from Ships, 1973, as modified by the Protocol of 1978, as amended, relating thereto (hereinafter referred to as "the Convention") under the authority of the Government of:

(full designation of the country)

#### Particulars of ship\*

Name of ship
Distinctive number or letters
Port of registry
Gross tonnage

#### THIS IS TO CERTIFY:

1 That the unmanned non-self-propelled (UNSP) barge has been surveyed in accordance with regulation 3.2 of Annex IV to the Convention;

2 That the survey shows that the unmanned non-self-propelled (UNSP) barge:

- .1 is not propelled by mechanical means;
- .2 has neither persons nor living animals on board;
- .3 is not used for holding sewage during transport; and
- .4 has no arrangements that could produce sewage as defined in regulation 1.3 of Annex IV to the Convention; and

3 That the UNSP barge is exempted, under regulation 3.2 of Annex IV to the Convention, from the certification and related survey requirements of regulations 4.1 and 5.1 of Annex IV to the Convention.

This certificate is valid until (*dd/mm/yyyy*).....

subject to the exemption conditions being maintained.

Completion date of the survey on which this certificate is based (*dd/mm/yyyy*).....

Issued at .....

(place of issue of certificate)

<sup>\*</sup> Alternatively, the particulars of the ship may be placed horizontally in boxes.

(date of issue) (dd/mm/yyyy):

(signature of duly authorized official issuing the certificate)

(seal or stamp of the authority, as appropriate)"

\*\*\*

#### RESOLUTION MEPC.331(76) (adopted on 17 June 2021)

#### AMENDMENTS TO THE INTERNATIONAL CONVENTION ON THE CONTROL OF HARMFUL ANTI-FOULING SYSTEMS ON SHIPS, 2001

#### Amendments to Annexes 1 and 4

### (Controls on cybutryne and form of the International Anti-fouling System Certificate)

#### THE MARINE ENVIRONMENT PROTECTION COMMITTEE,

RECALLING Article 38(a) of the Convention on the International Maritime Organization concerning the functions of the Marine Environment Protection Committee conferred upon it by international conventions for the prevention and control of marine pollution from ships,

RECALLING ALSO article 16 of the International Convention on the Control of Harmful Anti-fouling Systems on Ships, 2001 (the AFS Convention), which specifies the amendment procedure and confers upon the Marine Environment Protection Committee of the Organization the function of considering amendments thereto for adoption by the Parties,

HAVING CONSIDERED, at its seventy-sixth session, proposed amendments to the AFS Convention regarding controls on cybutryne and the form of the International Anti-fouling System Certificate,

1 ADOPTS, in accordance with article 16(2)(c) of the AFS Convention, amendments to Annexes 1 and 4, the text of which is set out in the annex to the present resolution;

2 DETERMINES, in accordance with article 16(2)(e)(ii) of the AFS Convention, that the amendments shall be deemed to have been accepted on 1 July 2022 unless prior to that date more than one third of the Parties have notified the Secretary-General that they object to the amendments;

3 INVITES the Parties to note that, in accordance with articles 16(2)(f)(ii) and (iii) of the AFS Convention, the said amendments shall enter into force on 1 January 2023 upon their acceptance in accordance with paragraph 2 above;

4 INVITES ALSO the Parties to remind ships that fly their flag and that are confirmed to be affected by the amendments to Annex 1 to the AFS Convention adopted through the present resolution to make a timely request for a survey for the issuance of an International Anti-fouling System Certificate, in the amended model form adopted through the present resolution, using the procedure outlined in paragraphs 4 and 5.3 of the annex to resolution MEPC.195(61), as may be amended by the Organization, so that ships have a valid International Anti-fouling System Certificate on board not later than 24 months after the entry into force of the amendments to Annex 1 to the AFS Convention adopted through the present resolution;

5 INVITES FURTHER the Parties to issue new International Anti-fouling System Certificates, in the amended model form adopted through the present resolution, at the next anti-fouling system application, in the case of ships that are confirmed not to be affected by the amendments to Annex 1 to the AFS Convention adopted through the present resolution; 6 REQUESTS the Secretary-General, for the purposes of article 16(2)(d) of the AFS Convention, to transmit certified copies of the present resolution and the text of the amendments contained in the annex to all Parties to the AFS Convention;

7 REQUESTS ALSO the Secretary-General to transmit copies of the present resolution and its annex to Members of the Organization which are not Parties to the AFS Convention;

8 REQUESTS FURTHER the Secretary-General to prepare a consolidated certified text of the AFS Convention.

..

# ANNEX

# AMENDMENTS TO THE INTERNATIONAL CONVENTION ON THE CONTROL OF HARMFUL ANTI-FOULING SYSTEMS ON SHIPS, 2001

# Annex 1

# Controls on anti-fouling systems

- 1
- The following rows are added to the table in Annex 1 to the 2001 AFS Convention:

Anti-fouling system	Control	Application	Effective date
	measures		
Cybutryne CAS No. 28159-98-0	Ships shall not apply or re-apply anti-fouling systems containing this substance	All ships	1 January 2023
Cybutryne CAS No. 28159-98-0	Ships bearing an anti-fouling system that contains this substance in the external coating layer of their hulls or external parts or surfaces on 1 January 2023 shall either: (1) remove the anti-fouling system; or (2) apply a coating that forms a barrier to this substance leaching from the underlying non-compliant anti-fouling system	All ships except: (1) fixed and floating platforms, FSUs, and FPSOs that have been constructed prior to 1 January 2023 and that have not been in dry-dock on or after 1 January 2023; (2) ships not engaged in international voyages; and (3) ships of less than 400 gross tonnage engaged in international voyages, if accepted by the coastal State(s)	At the next scheduled renewal of the anti-fouling system after 1 January 2023, but no later than 60 months following the last application to the ship of an anti-fouling system containing cybutryne
## Annex 4 Surveys and certification requirements for anti-fouling systems

2 Regulation 2(3) is replaced by the following:

"(3) For ships bearing an anti-fouling system controlled under Annex 1 that was applied before the date of entry into force of a control for such a system, the Administration shall issue a Certificate in accordance with paragraphs (1) and (2) of this regulation not later than two years after entry into force of that control. This paragraph shall not affect any requirement for ships to comply with Annex 1."

#### Appendix 1 to Annex 4 Model form of International Anti-fouling System Certificate

3 The section of the model form of the International Anti-fouling System Certificate (appendix 1) listing the compliance options for controlled anti-fouling systems on the ship is replaced by the following:

"An anti-fouling system controlled under Annex 1 containing:

	Has not been applied during or after construction of this ship	Has been applied on this ship previously, but has been removed by	Has been applied on this ship previously, but has been covered with a sealer coat applied by	Has been applied on this ship previously, but is not in the external coating layer of the hull or external parts or surfaces on	Was applied on this ship prior to
Organotin compound s which act as biocides		(insert name of the facility) ON 	(insert name of the facility) ON 	Not applicable	Not applicable
Cybutryne		(insert name of the facility) ON 	(insert name of the facility) ON 	1 January 2023	1 January 2023, but must be removed or covered with a sealer coat prior to 

\*\*\*

#### RESOLUTION MEPC.332(76) (adopted on 17 June 2021)

#### AMENDMENTS TO THE 2018 GUIDELINES ON THE METHOD OF CALCULATION OF THE ATTAINED ENERGY EFFICIENCY DESIGN INDEX (EEDI) FOR NEW SHIPS (RESOLUTION MEPC.308(73), AS AMENDED BY RESOLUTION MEPC.322(74))

THE MARINE ENVIRONMENT PROTECTION COMMITTEE,

RECALLING article 38(a) of the Convention on the International Maritime Organization concerning the functions of the Marine Environment Protection Committee (the Committee) conferred upon it by international conventions for the prevention and control of marine pollution from ships,

RECALLING ALSO that, at its sixty-second session, it adopted, by resolution MEPC.203(62), *Amendments to the annex of the Protocol of 1997 to amend the International Convention for the Prevention of Pollution from Ships, 1973, as modified by the Protocol of 1978 relating thereto* (inclusion of regulations on energy efficiency for ships in MARPOL Annex VI),

NOTING that the aforementioned amendments to MARPOL Annex VI entered into force on 1 January 2013,

NOTING ALSO that regulation 22 (Attained Energy Efficiency Design Index (attained EEDI)) of MARPOL Annex VI, as amended, requires that the EEDI shall be calculated taking into account the guidelines developed by the Organization,

NOTING FURTHER the 2012 Guidelines on the method of calculation of the attained Energy Efficiency Design Index (EEDI) for new ships, adopted at its sixty-third session by resolution MEPC.212(63), superseded by the 2014 Guidelines on the method of calculation of the attained Energy Efficiency Design Index (EEDI) for new ships (resolution MEPC.245(66)), which were subsequently superseded by the 2018 Guidelines on the method of calculation of the attained Energy Efficiency Design Index (EEDI) for new ships (resolution MEPC.245(66)), which were subsequently superseded by the 2018 Guidelines on the method of calculation of the attained Energy Efficiency Design Index (EEDI) for new ships (resolution MEPC.308(73)),

NOTING that, at its seventy-fourth session, it adopted, by resolution MEPC.322(74), Amendments to the 2018 Guidelines on the method of calculation of the attained Energy Efficiency Design Index (EEDI) for new ships,

HAVING CONSIDERED, at its seventy-sixth session, proposed amendments to the 2018 Guidelines on the method of calculation of the attained Energy Efficiency Design Index (EEDI) for new ships (resolution MEPC.308(73), as amended by resolution MEPC.322(74)),

1 ADOPTS amendments to the 2018 Guidelines on the method of calculation of the attained Energy Efficiency Design Index (EEDI) for new ships (resolution MEPC.308(73), as amended by resolution MEPC.322(74)), as set out in the annex to the present resolution;

2 INVITES Administrations to take into account the aforementioned amendments when developing and enacting national laws which give force to, and implement provisions set forth in regulation 20 of MARPOL Annex VI, as amended;

3 REQUESTS the Parties to MARPOL Annex VI and other Member Governments to bring the amendments to the attention of shipowners, ship operators, shipbuilders, ship designers and any other interested parties;

4 AGREES to keep these Guidelines, as amended, under review, in light of experience gained with their implementation.

## AMENDMENTS TO THE 2018 GUIDELINES ON THE METHOD OF CALCULATION OF THE ATTAINED ENERGY EFFICIENCY DESIGN INDEX (EEDI) FOR NEW SHIPS

1 A new section 3 is added, as follows:

#### "3 Mandatory Reporting of Attained EEDI Values and Related Information

- 3.1 In accordance with regulation 22.3 of MARPOL Annex VI, for each ship subject to regulation 24, the Administration or any organization duly authorized by it shall report the required and attained EEDI values and relevant information taking into account these Guidelines via electronic communication.
- 3.2 Information to be reported are as follows:
  - .1 applicable EEDI phase (e.g. Phase 1, Phase 2, etc.);
  - .2 identification number (IMO Secretariat use only);
  - .3 ship type;
  - .4 common commercial size reference<sup>\*</sup> (see Note (3) in appendix 5 to these Guidelines), if available;
  - .5 DWT or GT (as appropriate);
  - .6 year of delivery;
  - .7 required EEDI value;
  - .8 attained EEDI value;
  - .9 dimensional parameters (length  $L_{pp}$  (m), breadth  $B_s$  (m), and draught (m));
  - .10  $V_{ref}$  (knots) and  $P_{ME}$  (kW);
  - .11 use of innovative technologies (4th and 5th terms in the EEDI equation, if applicable);
  - .12 short statement<sup>\*</sup> describing the principal design elements or changes employed to achieve the attained EEDI (as appropriate), if available;
  - .13 type of fuel used in the calculation of the attained EEDI, and for dual-fuel engines, the f<sub>DFgas</sub> ratio; and
  - .14 ice class designation (if applicable).

<sup>\*</sup> Not subject to verification.

- 3.3 The information in paragraph 3.2 is not required to be reported for ships for which the required and attained EEDI values had been already reported to the Organization.
- 3.4 A standardized reporting format for Mandatory Reporting of Attained EEDI Values and Related Information is presented in appendix 5."
- 2 A new appendix 5 is added, as follows:

## "APPENDIX 5

## STANDARD FORMAT TO SUBMIT EEDI INFORMATION TO BE INCLUDED IN THE EEDI DATABASE

IMO	Type	Common	Capa (4	acity )	1	Dimensi parame	ional iters					Vref	PME	Туре	for	2	EEI (Ins innova te	DI 4th term stallation of ative electrical chnology)	EEI (Ins in me teo	DI 5th term tallation of novative echanical chnology)	Short statement as appropriate describing the principal
Number (1)	ship (2)	commercial size (3)	DWT	GT (5)	Lpp (m) (6)	Bs (m) (7)	Draught (m) (8)	Year of delivery	Applicable phase	Required EEDI	Attained EEDI	(knot) (9)	(kW)	of fuel (11)	iD⊧ gas (12)	class (13)	Yes/ No	Name, outline and means/ ways of performance of technology (14)	Yes/ No	Name, outline and means/ ways of performan ce of technology (14)	design elements or changes employed to achieve the attained EEDI (15)
											1										

#### Note:

- (1) IMO number to be submitted for Secretariat use only.
- (2) As defined in regulation 2 of MARPOL Annex VI.
- (3) Common commercial size reference (TEU for containership, CEU (RT43) for ro-ro cargo ship (vehicle carrier), cubic meter for gas carrier and LNG carrier), if available, should be provided.
- (4) The exact DWT or GT, as appropriate, should be provided. The Secretariat should round the DWT or GT data up to the nearest 500 when these data are subsequently provided to MEPC. (For containerships, 100% DWT should be provided while 70% of DWT should be used when calculating the EEDI value).
- (5) GT should be provided for a cruise passenger ship having non-conventional propulsion as defined in regulations 2.2.11and 2.2.19, respectively, of MARPOL Annex VI.
  Both DWT and GT should be provided for a ro-ro cargo ship (vehicle carrier) as defined in regulation 2.2.27 of MARPOL Annex VI.
- (6) As defined in paragraph 2.2.13 of the 2018 Guidelines on the method of calculation of the attained Energy Efficiency Design Index (EEDI) for new ships (resolution MEPC.308(73), as amended). The exact Lpp should be provided. The Secretariat will round the Lpp data up to the nearest 10 when these data are subsequently provided to MEPC.
- (7) As defined in paragraph 2.2.16 of the 2018 Guidelines on the method of calculation of the attained Energy Efficiency Design Index (EED) for new ships (resolution MEPC.308(73), as amended). The exact Bs should be provided. The Secretariat will round the Bs data up to the nearest 1 when these data are subsequently provided to MEPC.
- (8) As defined in paragraph 2.2.15 of the 2018 Guidelines on the method of calculation of the attained Energy Efficiency Design Index (EEDI) for new ships (resolution MEPC.308(73), as amended). The exact draught should be provided. The Secretariat will round the draught data up to the nearest 1 when these data are subsequently provided to MEPC.
- (9) As defined in paragraph 2.2.2 of the 2018 Guidelines on the method of calculation of the attained Energy Efficiency Design Index (EEDI) for new ships (resolution MEPC.308(73), as amended). The exact Vref should be provided. The Secretariat will round the Vref data up to the nearest 0.5 when these data are subsequently provided to MEPC.
- (10) As defined in paragraph 2.2.5.1 of the 2018 Guidelines on the method of calculation of the attained Energy Efficiency Design Index (EEDI) for new ships (resolution MEPC.308(73), as amended). The exact PME should be provided. The Secretariat will round the PME data up to the nearest 100 when these data are subsequently provided to MEPC.
- (11) As defined in paragraph 2.2.1 of the 2018 Guidelines on the method of calculation of the attained Energy Efficiency Design Index (EEDI) for new ships (resolution MEPC.308(73), as amended) or other (to be stated). In case of a ship equipped with a dual-fuel engine, type of "primary fuel" should be provided.
- (12) As defined in paragraph 2.2.1 of the 2018 Guidelines on the method of calculation of the attained Energy Efficiency Design Index (EEDI) for new ships (resolution MEPC.308 (73), as amended), if applicable.
- (13) Ice class, which was used to calculate correction factors for ice-classed ships as defined in paragraphs 2.2.8.1 and 2.2.11.1 of the 2018 Guidelines on the method of calculation of the attained Energy Efficiency Design Index (EEDI) for new ships (resolution MEPC.308(73), as amended), if applicable, should be provided.
- (14) In the case that the innovative energy efficiency technologies are already included in the 2013 Guidance on treatment of innovative energy efficiency technologies for calculation and verification of the attained EEDI (MEPC.1/Circ.815), the name of technology should be identified. Otherwise, name, outline and means/ways of performance of the technology should be identified.
- (15) To assist the IMO in assessing relevant design trends, provide a short statement as appropriate, describing the principal design elements or changes employed to achieve the attained EEDI.

#### UNIFIED INTERPRETATION TO REGULATION 2.23 OF MARPOL ANNEX VI

(update to the unified interpretation provided in paragraphs 1.2.3 and 1.2.4 of the annex to MEPC.1/Circ.795/Rev.4)

## 1 Definition of "new ship"

Regulation 2 *Definitions* 

Regulation 2.23 reads as follows:

"*New ship* means a ship:

- .1 for which the building contract is placed on or after 1 January 2013; or
- .2 in the absence of a building contract, the keel of which is laid or which is at a similar stage of construction on or after 1 July 2013; or
- .3 the delivery of which is on or after 1 July 2015."

## Interpretation:

1.1 For the application of the definition "new ship" as specified in regulation 2.23 to each phase specified in table 1 of regulation 21, it should be interpreted as follows:

- .1 the date specified in regulation 2.23.1 should be replaced with the start date of each phase;
- .2 the date specified in regulation 2.23.2 should be replaced with the date six months after the start date of each phase; and
- .3 the date specified in regulation 2.23.3 should, for Phase 1, 2 and 3, be replaced with the date 48 months after the start date of each phase.

1.2 With the above interpretations, the required EEDI of each phase is applied to the following new ship which falls into one of the categories defined in regulations 2.25 to 2.31 and to which chapter 4 is applicable:

(.....)

- .3 the required EEDI of Phase 2 is applied to the following new ship:
  - .1 for ship types where Phase 2 ends on 31 March 2022\*:
    - .1 the building contract of which is placed in Phase 2, and the delivery is before 1 April 2026; or

<sup>\*</sup> Unified Interpretation is applicable when resolution MEPC.324(75) enters into force on 1 April 2022.

.2 the building contract of which is placed before Phase 2, and the delivery is on or after 1 January 2024 and before 1 April 2026; or

in the absence of a building contract:

- .3 the keel of which is laid or which is at a similar stage of construction on or after 1 July 2020 and before 1 October 2022, and the delivery is before 1 April 2026; or
- .4 the keel of which is laid or which is at a similar stage of construction before 1 July 2020, and the delivery is on or after 1 January 2024 and before 1 April 2026.
- .2 for ship types where Phase 2 ends on 31 December 2024:
  - .1 the building contract of which is placed in Phase 2, and the delivery is before 1 January 2029; or
  - .2 the building contract of which is placed before Phase 2, and the delivery is on or after 1 January 2024 and before 1 January 2029; or

in the absence of a building contract:

- .3 the keel of which is laid or which is at a similar stage of construction on or after 1 July 2020 and before 1 July 2025, and the delivery is before 1 January 2029; or
- .4 the keel of which is laid or which is at a similar stage of construction before 1 July 2020, and the delivery is on or after 1 January 2024 and before 1 January 2029.
- .4 the required EEDI of Phase 3 is applied to the following new ship:
  - .1 for ship types where Phase 3 commences with 1 April 2022 and onwards:
    - .1 the building contract of which is placed in Phase 3; or
    - .2 the building contract of which is placed before Phase 3, and the delivery is on or after 1 April 2026; or

in the absence of a building contract:

- .3 the keel of which is laid or which is at a similar stage of construction on or after 1 October 2022; or
- .4 the keel of which is laid or which is at a similar stage of construction before 1 October 2022 and the delivery of which is on or after 1 April 2026.
- .2 for ship types where Phase 3 commences with 1 January 2025 and onwards:

- .1 the building contract of which is placed in Phase 3; or
- .2 the building contract of which is placed before Phase 3, and the delivery is on or after 1 January 2029; or

in the absence of a building contract:

- .3 the keel of which is laid or which is at a similar stage of construction on or after 1 July 2025; or
- .4 the keel of which is laid or which is at a similar stage of construction before 1 July 2025 and the delivery of which is on or after 1 January 2029.

(.....)

\*\*\*

#### RESOLUTION MEPC.333(76) (adopted on 17 June 2021)

## 2021 GUIDELINES ON THE METHOD OF CALCULATION OF THE ATTAINED ENERGY EFFICIENCY EXISTING SHIP INDEX (EEXI)

THE MARINE ENVIRONMENT PROTECTION COMMITTEE,

RECALLING Article 38(a) of the Convention on the International Maritime Organization concerning the functions of the Marine Environment Protection Committee conferred upon it by international conventions for the prevention and control of marine pollution from ships,

NOTING that it adopted, by resolution MEPC.328(76), the 2021 revised MARPOL Annex VI, which is expected to enter into force on 1 November 2022 upon its deemed acceptance on 1 May 2022,

NOTING IN PARTICULAR that the 2021 revised MARPOL Annex VI contains amendments concerning mandatory goal-based technical and operational measures to reduce carbon intensity of international shipping,

NOTING FURTHER that regulation 23 of MARPOL Annex VI requires that the attained EEXI shall be calculated taking into account the guidelines developed by the Organization,

RECOGNIZING that the aforementioned amendments to MARPOL Annex VI require relevant guidelines for uniform and effective implementation of the regulations and to provide sufficient lead time for industry to prepare,

HAVING CONSIDERED, at its seventy-sixth session, draft 2021 Guidelines on the method of calculation of the attained Energy Efficiency Existing Ship Index (EEXI),

1 ADOPTS the 2021 Guidelines on the method of calculation of the attained Energy *Efficiency Existing Ship Index (EEXI)*, as set out in the annex to the present resolution;

2 INVITES Administrations to take the annexed Guidelines into account when developing and enacting national laws which give force to and implement requirements set forth in regulation 23 of MARPOL Annex VI;

3 REQUESTS the Parties to MARPOL Annex VI and other Member Governments to bring the annexed Guidelines to the attention of masters, seafarers, shipowners, ship operators and any other interested parties;

4 AGREES to keep the Guidelines under review in light of experience gained with their implementation and in light of the review of EEXI regulations to be completed by the Organization by 1 January 2026 as identified in regulation 25.3 of MARPOL Annex VI.

## 2021 GUIDELINES ON THE METHOD OF CALCULATION OF THE ATTAINED ENERGY EFFICIENCY EXISTING SHIP INDEX (EEXI)

## CONTENTS

- 1 Definitions
- 2 Energy Efficiency Existing Ship Index (EEXI)
- 2.1 EEXI formula
- 2.2 Parameters
- 2.2.1  $P_{ME(i)}$ ; Power of main engines
- 2.2.2 *P*<sub>AE(i)</sub>; Power of auxiliary engines
- 2.2.3 V<sub>ref</sub>; Ship speed
- 2.2.4 SFC ; Certified specific fuel consumption
- 2.2.5  $C_F$ ; Conversion factor between fuel consumption and CO<sub>2</sub> emission
- 2.2.6 Correction factor for ro-ro cargo and ro-ro passenger ships ( $f_{jRoRo}$ )
- 2.2.7 Correction factor for ro-ro cargo ships (vehicle carrier) (*f*<sub>cVEHICLE</sub>)
- APPENDIX Parameters to calculate V<sub>ref,app</sub>

## 1 Definitions

1.1 *MARPOL* means the International Convention for the Prevention of Pollution from Ships, 1973, as modified by the Protocols of 1978 and 1997 relating thereto, as amended.

1.2 For the purpose of these Guidelines, the definitions in MARPOL Annex VI, as amended, apply.

## 2 Energy Efficiency Existing Ship Index (EEXI)

## 2.1 EEXI formula

The attained Energy Efficiency Existing Ship Index (EEXI) is a measure of ship's energy efficiency (g/t\*nm) and calculated by the following formula:



- \* If part of the Normal Maximum Sea Load is provided by shaft generators,  $SFC_{ME}$  and  $C_{FME}$  may for that part of the power be used instead of  $SFC_{AE}$  and  $C_{FAE}$
- \*\* In case of  $P_{PTI(i)} > 0$ , the average weighted value of  $(SFC_{ME} \cdot C_{FME})$  and  $(SFC_{AE} \cdot C_{FAE})$  to be used for calculation of  $P_{eff}$ 
  - **Note:** This formula may not be applicable to a ship having diesel-electric propulsion, turbine propulsion or hybrid propulsion system, except for cruise passenger ships and LNG carriers.

Ships falling into the scope of EEDI requirement can use their attained EEDI calculated in accordance with the 2018 Guidelines on the method of calculation of the attained EEDI for new ships (resolution MEPC.308(73), as amended, the "EEDI Calculation Guidelines" hereafter) as the attained EEXI if the value of the attained EEDI is equal to or less than that of the required EEXI.

## 2.2 Parameters

For calculation of the attained EEXI by the formula in paragraph 2.1, parameters under the EEDI Calculation Guidelines apply, unless expressly provided otherwise. In referring to the aforementioned guidelines, the terminology "EEDI" should be read as "EEXI".

## 2.2.1 $P_{ME(i)}$ ; Power of main engines

In cases where overridable Shaft / Engine Power Limitation is installed in accordance with the 2021 Guidelines on the shaft / engine power limit to comply with the EEXI requirements and use of a power reserve (resolution MEPC.335(76)),  $P_{ME(i)}$  is 83% of the limited installed power ( $MCR_{lim}$ ) or 75% of the original installed power (MCR), whichever is lower, for each main engine (*i*). In cases where the overridable Shaft / Engine Power Limitation and shaft generator(s) are installed, in referring to paragraph 2.2.5.2 (option 1) of the EEDI Calculation Guidelines, " $MCR_{ME}$ " should be read as " $MCR_{lim}$ ".

For LNG carriers having steam turbine or diesel electric propulsion,  $P_{ME(i)}$  is 83% of the limited installed power (*MCR*<sub>*lim*</sub>, *MPP*<sub>*lim*</sub>), divided by the electrical efficiency in case of diesel electric propulsion system, for each main engine (*i*). For LNG carriers, the power from combustion of

the excessive natural boil-off gas in the engines or boilers to avoid releasing to the atmosphere or unnecessary thermal oxidation should be deducted from  $P_{ME(i)}$  with the approval of the verifier.

## 2.2.2 *P*<sub>AE(i)</sub>; Power of auxiliary engines

2.2.2.1  $P_{AE(i)}$  is calculated in accordance with paragraph 2.2.5.6 of the EEDI Calculation Guidelines.

2.2.2.2 For ships where power of auxiliary engines ( $P_{AE}$ ) value calculated by paragraphs 2.2.5.6.1 to 2.2.5.6.3 of the EEDI Calculation Guidelines is significantly different from the total power used at normal seagoing, e.g. in cases of passenger ships, the  $P_{AE}$  value should be estimated by the consumed electric power (excluding propulsion) in conditions when the ship is engaged in a voyage at reference speed ( $V_{ref}$ ) as given in the electric power table, divided by the average efficiency of the generator(s) weighted by power (see appendix 2 of the EEDI Calculation Guidelines).

2.2.2.3 In cases where the electric power table is not available, the  $P_{AE}$  value may be approximated either by:

- .1 annual average figure of  $P_{AE}$  at sea from onboard monitoring obtained prior to the EEXI certification;
- .2 for cruise passenger ships, approximated value of power of auxiliary engines  $(P_{AE,app})$ , as defined below:

 $P_{AE,app} = 0.1193 \times GT + 1814.4$  [kW]

.3 for ro-ro passenger ships, approximated value of power of auxiliary engines  $(P_{AE,app})$ , as defined below:

 $P_{AE,app} = 0.866 \times GT^{0.732}$  [kW]

## 2.2.3 *V*<sub>ref</sub>; Ship speed

2.2.3.1 For ships falling into the scope of the EEDI requirement, the ship speed  $V_{ref}$  should be obtained from an approved speed-power curve as defined in the 2014 Guidelines on survey and certification of the Energy Efficiency Design Index (EEDI), as amended (resolution MEPC.254(67), as amended).

2.2.3.2 For ships not falling into the scope of the EEDI requirement, the ship speed  $V_{ref}$  should be obtained from an estimated speed-power curve as defined in the 2021 Guidelines on survey and certification of the attained EEXI (resolution MEPC.334(76)).

2.2.3.3 For ships not falling into the scope of the EEDI requirement but whose sea trial results, which may have been calibrated by the tank test, under the EEDI draught and the sea condition as specified in paragraph 2.2.2 of the EEDI Calculation Guidelines are included in the sea trial report, the ship speed  $V_{ref}$  may be obtained from the sea trial report:

$$V_{ref} = V_{S,EEDI} \times \left[\frac{P_{ME}}{P_{S,EEDI}}\right]^{\frac{1}{3}}$$
 [knot]

where,

 $V_{S, EEDI}$ , is the sea trial service speed under the EEDI draught; and

 $P_{S,EEDI}$  is power of the main engine corresponding to  $V_{S,EEDI}$ .

2.2.3.4 For containerships, bulk carriers or tankers not falling into the scope of the EEDI requirement but whose sea trial results, which may have been calibrated by the tank test, under the design load draught and sea condition as specified in paragraph 2.2.2 of the EEDI Calculation Guidelines are included in the sea trial report, the ship speed  $V_{ref}$  may be obtained from the sea trial report:

$$V_{ref} = k^{\frac{1}{3}} \times \left(\frac{DWT_{S,service}}{Capacity}\right)^{\frac{2}{9}} \times V_{S,service} \times \left[\frac{P_{ME}}{P_{S,service}}\right]^{\frac{1}{3}} \quad [\text{knot}]$$

where,

 $V_{S,service}$  is the sea trial service speed under the design load draught;

DWT<sub>S,service</sub> is the deadweight under the design load draught;

 $P_{S,service}$  is the power of the main engine corresponding to  $V_{S,service}$ ;

*k* is the scale coefficient, which should be:

- .1 0.95 for containerships with 120,000 DWT or less;
- .2 0.93 for containerships with more than 120,000 DWT;
- .3 0.97 for bulk carrier with 200,000 DWT or less;
- .4 1.00 for bulk carrier with more than 200,000 DWT;
- .5 0.97 for tanker with 100,000 DWT or less; and
- .6 1.00 for tanker with more than 100,000 DWT.

2.2.3.5 In cases where the speed-power curve is not available or the sea trial report does not contain the EEDI or design load draught condition, the ship speed  $V_{ref}$  can be approximated by  $V_{ref,app}$  to be obtained from statistical mean of distribution of ship speed and engine power, as defined below:

$$V_{ref,app} = (V_{ref,avg} - m_V) \times \left[\frac{\sum P_{ME}}{0.75 \times MCR_{avg}}\right]^{\frac{1}{3}} \quad [\text{knot}]$$

For LNG carriers having diesel electric propulsion system and cruise passenger ship having non-conventional propulsion,

$$V_{ref,app} = (V_{ref,avg} - m_V) \times \left[\frac{\sum MPP_{Motor}}{MPP_{avg}}\right]^{\frac{1}{3}}$$
 [knot]

where,

 $V_{ref,avg}$  is a statistical mean of distribution of ship speed in given ship type and ship size, to be calculated as follows:

$$V_{ref,avg} = A \times B^C$$

where

A, B and C are the parameters given in the appendix;

 $m_V$  is a performance margin of a ship, which should be 5% of  $V_{ref,avg}$  or one knot, whichever is lower; and

 $MCR_{avg}$  is a statistical mean of distribution of MCRs for main engines and  $MPP_{avg}$  is a statistical mean of distribution of MPPs for motors in given ship type and ship size, to be calculated as follows:

$$MCR_{ava} or MPP_{ava} = D \times E^F$$

where

D, E and F are the parameters given in the appendix;

In cases where the overridable Shaft / Engine Power Limitation is installed, the ship speed  $V_{ref}$  approximated by  $V_{ref,app}$  should be calculated as follows:

$$V_{ref,app} = (V_{ref,avg} - m_V) \times \left[\frac{\sum P_{ME}}{0.75 \times MCR_{avg}}\right]^{\frac{1}{3}} \quad [\text{knot}]$$

For LNG carriers having diesel electric propulsion system and cruise passenger ship having non-conventional propulsion, the ship speed  $V_{ref}$  approximated by  $V_{ref,app}$  should be calculated as follows:

$$V_{ref,app} = \left(V_{ref,avg} - m_V\right) \times \left[\frac{\sum MPP_{lim}}{MPP_{avg}}\right]^{\frac{1}{3}}$$

2.2.3.6 Notwithstanding the above, in cases where the energy saving device<sup>\*</sup> is installed, the effect of the device may be reflected in the ship speed  $V_{ref}$  with the approval of the verifier, based on the following methods in accordance with defined quality and technical standards:

- .1 sea trials after installation of the device; and/or
- .2 dedicated model tests; and/or
- .3 numerical calculations.

<sup>\*</sup> Devices that shift the power curve, which results in the change of P<sub>P</sub> and V<sub>ref</sub>, as specified in MEPC.1/Circ.815 on 2013 Guidance on treatment of innovative energy efficiency technologies for calculation and verification of the attained EEDI.

## 2.2.4 SFC; Certified specific fuel consumption

In cases where overridable Shaft / Engine Power Limitation is installed, the *SFC* corresponding to the  $P_{ME}$  should be interpolated by using *SFC*s listed in an applicable test report included in an approved NO<sub>X</sub> Technical File of the main engine as defined in paragraph 1.3.15 of the NO<sub>X</sub> Technical Code.

Notwithstanding the above, the *SFC* specified by the manufacturer or confirmed by the verifier may be used.

For those engines which do not have a test report included in the NO<sub>X</sub> Technical File and which do not have the *SFC* specified by the manufacturer or confirmed by the verifier, the *SFC* can be approximated by  $SFC_{app}$  defined as follows:

$$SFC_{ME,app} = 190 \left[ g/kWh \right]$$

$$SFC_{AE,app} = 215 \left[ g/kWh \right]$$

## 2.2.5 $C_F$ ; Conversion factor between fuel consumption and CO<sub>2</sub> emission

For those engines which do not have a test report included in the NO<sub>X</sub> Technical File and which do not have the *SFC* specified by the manufacturer, the  $C_F$  corresponding to  $SFC_{app}$  should be defined as follows:

 $C_F = 3.114 [t \cdot CO_2/t \cdot Fuel]$  for diesel ships (incl. HFO use in practice)

Otherwise, paragraph 2.2.1 of the EEDI Calculation Guidelines applies.

## 2.2.6 Correction factor for ro-ro cargo and ro-ro passenger ships (*f*<sub>*jRoRo*</sub>)

For ro-ro cargo and ro-ro passenger ships,  $f_{iRoRo}$  is calculated as follows:

$$f_{jRoRo} = \frac{1}{F_{n_L}^{\alpha} \cdot \left(\frac{Lpp}{B_S}\right)^{\beta} \cdot \left(\frac{B_S}{d_S}\right)^{\gamma} \cdot \left(\frac{Lpp}{p_1^{-1}/3}\right)^{\delta}} \qquad ; \text{ if } f_{jRoRo} > 1 \text{ then } f_j = 1$$

where the Froude number,  $F_{n_i}$ , is defined as:

$$F_{n_L} = \frac{0.5144 \cdot V_{ref,F}}{\sqrt{L_{pp} \cdot g}}$$

where  $V_{ref,F}$  is the ship design speed corresponding to 75% of  $MCR_{ME}$ .

and the exponents  $\alpha$ ,  $\beta$ ,  $\gamma$  and  $\delta$  are defined as follows:

Ship type	Exponent:			
	α	β	γ	δ
Ro-ro cargo ship	2.00	0.50	0.75	1.00
Ro-ro passenger ship	2.50	0.75	0.75	1.00

## 2.2.7 Cubic capacity correction factor for ro-ro cargo ships (vehicle carrier) ( $f_{cVEHICLE}$ )

For ro-ro cargo ships (vehicle carrier) having a DWT/GT ratio of less than 0.35, the following cubic capacity correction factor,  $f_{cVEHICLE}$ , should apply:

$$f_{cVEHICLE} = \left(\frac{\left(\frac{DWT}{GT}\right)}{0.35}\right)^{-0.8}$$

Where DWT is the capacity and GT is the gross tonnage in accordance with the International Convention of Tonnage Measurement of Ships 1969, annex I, regulation 3.

#### APPENDIX

## Parameters to calculate V<sub>ref,avg</sub>

Ship type	А	В	С
Bulk carrier	10.6585	DWT of the ship	0.02706
Gas carrier	7.4462	DWT of the ship	0.07604
Tanker	8.1358	DWT of the ship	0.05383
Containership	3.2395 DWT of the ship where DWT ≤ 80,000 80,000 where DWT > 80,000		0.18294
General cargo ship	2.4538	DWT of the ship	0.18832
Refrigerated cargo carrier	1.0600	DWT of the ship	0.31518
Combination carrier	8.1391	DWT of the ship	0.05378
LNG carrier	11.0536	DWT of the ship	0.05030
Ro-ro cargo ship (vehicle carrier)	16.6773	DWT of the ship	0.01802
Ro-ro cargo ship	8.0793	DWT of the ship	0.09123
Ro-ro passenger ship	4.1140	DWT of the ship	0.19863
Cruise passenger ship having non-conventional propulsion	5.1240	GT of the ship	0.12714

Parameters to calculate  $MCR_{avg}$  or  $MPP_{avg}$  (= D x E<sup>F</sup>)

Ship type	D	E	F
Bulk carrier	23.7510	DWT of the ship	0.54087
Gas carrier	21.4704	DWT of the ship	0.59522
Tanker	22.8415	DWT of the ship	0.55826
Containership	0.5042	DWT of the ship where DWT ≤ 95,000 95,000 where DWT > 95,000	1.03046
General cargo ship	0.8816 DWT of the ship		0.92050
Refrigerated cargo carrier	0.0272	DWT of the ship	1.38634
Combination carrier	22.8536 DWT of the ship		0.55820
LNG carrier	20.7096	DWT of the ship	0.63477
Ro-ro cargo ship (vehicle carrier)	262.7693	DWT of the ship	0.39973
Ro-ro cargo ship	37.7708	DWT of the ship	0.63450
Ro-ro passenger ship	9.1338	DWT of the ship	0.91116
Cruise passenger ship having non- conventional propulsion	1.3550	GT of the ship	0.88664

## Calculation of parameters to calculate V<sub>ref,avg</sub> and MCR<sub>avg</sub>

#### Data sources

## 1 IHS Fairplay (IHSF) database with the following conditions are used.

Ship type	Ship size	Delivered period	Type of propulsion systems	Population
Bulk carrier	≥ 10,000 DWT		Conventional	2,433
Gas carrier	≥ 2,000 DWT		Conventional	292
Tanker	≥ 4,000 DWT		Conventional	3,345
Containership	≥ 10,000 DWT		Conventional	2,185
General cargo ship	≥ 3,000 DWT	From 1 January 1999	Conventional	1,673
Refrigerated cargo carrier	≥ 3,000 DWT	to 1 January 2009	Conventional	53
Combination carrier	≥ 4,000 DWT		Conventional	3,351
LNG carrier	≥ 10,000 DWT		Conventional, Non-conventional	185
Ro-ro cargo ship (vehicle carrier)	≥ 10,000 DWT		Conventional	301
Ro-ro cargo ship	≥ 1,000 DWT	From 1 January 1998	Conventional	188
Ro-ro passenger ship	≥ 250 DWT	to 31 December 2010	Conventional	350
Cruise passenger ship having non-conventional propulsion	≥ 25,000 GT	From 1 January 1999 to 1 January 2009	Non-conventional	93

2 Data sets with blank/zero "Service speed", "Capacity" and/or Total kW of M/E" are removed.

3 Ship type is in accordance with table 1 and table 2 of resolution MEPC.231(65) on 2013 Guidelines for calculation of reference lines for use with the Energy Efficiency Design Index (EEDI). However, "Gas carrier" does not include "LNG carrier". Parameters for "LNG carrier" are given separately.

\*\*\*

#### RESOLUTION MEPC.334(76) (adopted on 17 June 2021)

## 2021 GUIDELINES ON SURVEY AND CERTIFICATION OF THE ATTAINED ENERGY EFFICIENCY EXISTING SHIP INDEX (EEXI)

THE MARINE ENVIRONMENT PROTECTION COMMITTEE,

RECALLING Article 38(a) of the Convention on the International Maritime Organization concerning the functions of the Marine Environment Protection Committee conferred upon it by international conventions for the prevention and control of marine pollution from ships,

NOTING that it adopted, by resolution MEPC.328(76), the 2021 revised MARPOL Annex VI, which is expected to enter into force on 1 November 2022 upon its deemed acceptance on 1 May 2022,

NOTING IN PARTICULAR that the 2021 revised MARPOL Annex VI contains amendments concerning mandatory goal-based technical and operational measures to reduce carbon intensity of international shipping,

NOTING FURTHER that regulation 5 (Surveys) of MARPOL Annex VI, as amended, requires that ships to which chapter 4 applies shall also be subject to survey and certification taking into account guidelines developed by the Organization,

RECOGNIZING that the aforementioned amendments to MARPOL Annex VI require relevant guidelines for uniform and effective implementation of the regulations and to provide sufficient lead time for industry to prepare,

HAVING CONSIDERED, at its seventy-sixth session, draft 2021 Guidelines on survey and certification of the Energy Efficiency Existing Ship Index (EEXI),

1 ADOPTS the 2021 Guidelines on survey and certification of the Energy Efficiency Existing Ship Index (EEXI), as set out in the annex to the present resolution;

2 INVITES Administrations to take the annexed Guidelines into account when developing and enacting national laws which give force to and implement requirements set forth in regulation 5 of MARPOL Annex VI;

3 REQUESTS the Parties to MARPOL Annex VI and other Member Governments to bring the annexed Guidelines to the attention of masters, seafarers, shipowners, ship operators and any other interested parties;

4 AGREES to keep the Guidelines under review in light of experience gained with their implementation and in light of the review of EEXI regulations to be completed by the Organization by 1 January 2026 as identified in regulation 25.3 of MARPOL Annex VI.

## 2021 GUIDELINES ON SURVEY AND CERTIFICATION OF THE ATTAINED ENERGY EFFICIENCY EXISTING SHIP INDEX (EEXI)

## **Table of contents**

- 1 GENERAL
- 2 DEFINITIONS
- 3 APPLICATION
- 4 PROCEDURES FOR SURVEY AND CERTIFICATION
  - 4.1 General
  - 4.2 Verification of the attained EEXI
  - 4.3 Verification of the attained EEXI in case of major conversion
- APPENDIX Sample of EEXI Technical File

## 1 GENERAL

The purpose of these guidelines is to assist verifiers of the Energy Efficiency Existing Ship Index (EEXI) of ships in conducting the survey and certification of the EEXI, in accordance with regulations 5, 6, 7, 8 and 9 of MARPOL Annex VI, and assist shipowners, shipbuilders, manufacturers and other interested parties in understanding the procedures for the survey and certification of the EEXI.

## 2 DEFINITIONS<sup>1</sup>

2.1 *Verifier* means an Administration, or organization duly authorized by it, which conducts the survey and certification of the EEXI in accordance with regulations 5, 6, 7, 8 and 9 of MARPOL Annex VI and these Guidelines.

2.2 *Ship of the same type* means a ship the hull form (expressed in the lines such as sheer plan and body plan), excluding additional hull features such as fins, and principal particulars of which are identical to that of the base ship.

2.3 *Tank test* means model towing tests, model self-propulsion tests and model propeller open water tests. Numerical calculations may be accepted as equivalent to model propeller open water tests or used to complement the tank tests conducted (e.g. to evaluate the effect of additional hull features such as fins, etc. on ships' performance), or as a replacement for model tests provided that the methodology and numerical model used have been validated/calibrated against parent hull sea trials and/or model tests, with the approval of the verifier.

2.4 *MARPOL* means the International Convention for the Prevention of Pollution from Ships, 1973, as modified by the Protocols of 1978 and 1997 relating thereto, as amended.

2.5 For the purpose of these Guidelines, the definitions in MARPOL Annex VI, as amended, apply.

## 3 APPLICATION

These Guidelines should be applied to ships for which an application for a survey for verification of the ship's EEXI specified in regulation 5 of MARPOL Annex VI has been submitted to a verifier.

## 4 PROCEDURES FOR SURVEY AND CERTIFICATION

## 4.1 General

4.1.1 The attained EEXI should be calculated in accordance with regulation 23 of MARPOL Annex VI and the 2021 Guidelines on the method of calculation of the attained Energy Efficiency Existing Ship Index (EEXI) (resolution MEPC.333(76)) (EEXI Calculation Guidelines).

4.1.2 The 2013 Guidance on treatment of innovative energy efficiency technologies for calculation and verification of the attained EEDI (MEPC.1/Circ.815) should be applied for calculation of the attained EEXI, if applicable.

<sup>&</sup>lt;sup>1</sup> Other terms used in these Guidelines have the same meaning as those defined in the 2018 Guidelines on the method of calculation of the attained EEDI for new ships (resolution MEPC.308(73), as amended) and the 2021 Guidelines on the method of calculation of the attained EEXI (resolution MEPC.333(76)).

4.1.3 The information used in the verification process may contain confidential information of submitters, including shipyards, which requires Intellectual Property Rights (IPR) protection. In the case where the submitter wants a non-disclosure agreement with the verifier, the additional information should be provided to the verifier upon mutually agreed terms and conditions.

## 4.2 Verification of the attained EEXI

4.2.1 For verification of the attained EEXI, an application for a survey and an EEXI Technical File containing the necessary information for the verification and other relevant background documents should be submitted to a verifier, unless the attained EEDI of the ship satisfies the required EEXI.

4.2.2 The EEXI Technical File should be written at least in English. The EEXI Technical File should include, but not be limited to:

- .1 deadweight (DWT) or gross tonnage (GT) for ro-ro passenger ship and cruise passenger ship having non-conventional propulsion;
- .2 the rated installed power (*MCR*) of the main and auxiliary engines;
- .3 the limited installed power (*MCR*<sub>lim</sub>) in cases where the overridable Shaft / Engine Power Limitation system is installed;
- .4 the ship speed  $(V_{ref})$ ;
- .5 the approximate ship speed ( $V_{ref,app}$ ) for pre-EEDI ships in cases where the speed-power curve is not available, as specified in paragraph 2.2.3.5 of the EEXI Calculation Guidelines;
- .6 an approved speed-power curve under the EEDI condition as specified in paragraph 2.2 of the EEDI Calculation Guidelines, which is described in the EEDI Technical File, in cases where regulation 22 of MARPOL Annex VI (Attained EEDI) is applied;
- .7 an estimated speed-power curve under the EEDI condition, or under a different load draught to be calibrated to the EEDI condition, obtained from tank test and/or numerical calculations, if available;
- .8 estimation process and methodology of the power curves, as necessary, including documentation on consistency with the defined quality standards (e.g. ITTC 7.5-03-01-02 and ITTC 7.5-03-01-04 in their latest revisions) and the verification of the numerical set-up with parent hull or the reference set of comparable ships in case of using numerical calculations;
- .9 a sea trial report including sea trial results, which may have been calibrated by the tank test, under the sea condition as specified in paragraph 2.2.2 of the EEDI Calculation Guidelines, if available;
- .10 calculation process of  $V_{ref,app}$  for pre-EEDI ships in cases where the speed-power curve is not available, as specified in paragraph 2.2.3.5 of the EEXI Calculation Guidelines;
- .11 type of fuel;

- .12 the specific fuel consumption (*SFC*) of the main and auxiliary engines, as specified in paragraph 2.2.3 of the EEXI Calculation Guidelines;
- .13 the electric power table<sup>2</sup> for certain ship types, as necessary, as defined in the EEDI Calculation Guidelines;
- .14 the documented record of annual average figure of the auxiliary engine load at sea obtained prior to the date of application for a survey for verification of the ship's EEXI, as specified in paragraph 2.2.2.3 of the EEXI Calculation Guidelines, if applicable;
- .15 calculation process of  $P_{AE,app}$ , as specified in paragraph 2.2.2.3 of the EEXI Calculation Guidelines, if applicable;
- .16 principal particulars, ship type and the relevant information to classify the ship as such a ship type, classification notations and an overview of the propulsion system and electricity supply system on board;
- .17 description of energy saving equipment, if available;
- .18 calculated value of the attained EEXI, including the calculation summary, which should contain, at a minimum, each value of the calculation parameters and the calculation process used to determine the attained EEXI; and
- .19 for LNG carriers:
  - .1 type and outline of propulsion systems (such as direct drive diesel, diesel electric, steam turbine);
  - .2 LNG cargo tank capacity in m<sup>3</sup> and BOR as defined in paragraph 2.2.5.6.3 of the EEDI Calculation Guidelines;
  - .3 shaft power of the propeller shaft after transmission gear at 100% of the rated output of motor (*MPP*<sub>Motor</sub>) and  $\eta_{(i)}$  for diesel electric;
  - .4 shaft power of the propeller shaft after transmission gear at the de-rated output of motor (*MPP*<sub>Motor,lim</sub>) in cases where the overridable Shaft / Engine Power Limitation is installed;
  - .5 maximum continuous rated power (*MCR*<sub>SteamTurbine</sub>) for steam turbine;
  - .6 limited maximum continuous rated power (*MCR*<sub>SteamTurbine,lim</sub>) for steam turbine in cases where the overridable Shaft / Engine Power Limitation is installed; and
  - .7 SFC<sub>SteamTurbine</sub> for steam turbine, as specified in paragraph 2.2.7.2 of the EEDI Calculation Guidelines. If the calculation is not available from the manufacturer, SFC<sub>SteamTurbine</sub> may be calculated by the submitter.

A sample of an EEXI Technical File is provided in the appendix.

<sup>&</sup>lt;sup>2</sup> Electric power tables should be validated separately, taking into account the guidelines set out in appendix 2 of the 2014 Guidelines on survey and certification of the Energy Efficiency Design Index (EEDI) (resolution MEPC.254(67), as amended by resolutions MEPC.261(68) and MEPC.309(73)); consolidated text: MEPC.1/Circ.855/Rev.2, as may be further amended).

4.2.3 The *SFC* should be corrected to the value corresponding to the ISO standard reference conditions using the standard lower calorific value of the fuel oil, referring to ISO 15550:2002 and ISO 3046-1:2002. For the confirmation of the *SFC*, a copy of the approved NO<sub>X</sub> Technical File and documented summary of the correction calculations should be submitted to the verifier.

4.2.4 For ships equipped with dual-fuel engine(s) using LNG and fuel oil, the  $C_{F}$ -factor for gas (LNG) and the specific fuel consumption (*SFC*) of gas fuel should be used by applying the criteria specified in paragraph 4.2.3 of the 2014 Guidelines on survey and certification of the Energy Efficiency Design Index (EEDI), as amended,<sup>3</sup> as a basis for the guidance of the Administration.

4.2.5 Notwithstanding paragraphs 4.2.3 and 4.2.4, in cases where overridable Shaft / Engine Power Limitation is installed, or in cases where engines do not have a test report included in the NO<sub>X</sub> Technical File, *SFC* should be calculated in accordance with paragraph 2.2.3 of the EEXI Calculation Guidelines. For this purpose, actual performance records of the engine may be used if satisfactory and acceptable to the verifier.

4.2.6 The verifier may request further information from the submitter, as specified in paragraph 4.2.7 of the EEDI Survey and Certification Guidelines, in addition to that contained in the EEXI Technical File, as necessary, to examine the calculation process of the attained EEXI.

4.2.7 In cases where the sea trial report as specified in paragraph 4.2.2.9 is submitted, the verifier should request further information from the submitter to confirm that:

- .1 the sea trial was conducted in accordance with the conditions specified in paragraphs 4.3.3, 4.3.4 and 4.3.7 of the EEDI Survey and Certification Guidelines, as applicable;
- .2 sea conditions were measured in accordance with ISO 15016:2002 or the equivalent if satisfactory and acceptable to the verifier;
- .3 ship speed was measured in accordance with ISO 15016:2002 or the equivalent if satisfactory and acceptable to the verifier; and
- .4 the measured ship speed was calibrated, if necessary, by taking into account the effects of wind, tide, waves, shallow water and displacement in accordance with ISO 15016:2002 or the equivalent which may be acceptable provided that the concept of the method is transparent for the verifier and publicly available/accessible.

4.2.8 The estimated speed-power curve obtained from the tank test and/or numerical calculations and/or the sea trial results calibrated by the tank test should be reviewed on the basis of the relevant documents in accordance with the EEDI Survey and Certification Guidelines, the defined quality standards (e.g. ITTC 7.5-03-01-02 and ITTC 7.5-03-01-04 in their latest revisions) and the verification of the numerical set-up with parent hull or the reference set of comparable ships.

4.2.9 In cases where the overridable Shaft / Engine Power Limitation system is installed, the verifier should confirm that the system is appropriately installed and sealed in accordance with the *2021 Guidelines on the Shaft / Engine Power Limitation system to comply with the EEXI requirements and use of a power reserve* (resolution MEPC.335(76)) and that a verified Onboard Management Manual (OMM) for overridable Shaft / Engine Power Limitation is on board the ship.

<sup>&</sup>lt;sup>3</sup> Resolution MEPC.254(67), as amended.

## 4.3 Verification of the attained EEXI in case of major conversion

4.3.1 In cases of a major conversion of a ship taking place at or after the completion date of the survey for EEXI verification specified in regulation 5.4.7 of MARPOL Annex VI, the shipowner should submit to a verifier an application for a general or partial survey with the EEXI Technical File duly revised, based on the conversion made and other relevant background documents.

- 4.3.2 The background documents should include as a minimum, but are not limited to:
  - .1 details of the conversion;
  - .2 EEXI parameters changed after the conversion and the technical justifications for each respective parameter;
  - .3 reasons for other changes made in the EEXI Technical File, if any; and
  - .4 calculated value of the attained EEXI with the calculation summary, which should contain, as a minimum, each value of the calculation parameters and the calculation process used to determine the attained EEXI after the conversion.

4.3.3 The verifier should review the revised EEXI Technical File and other documents submitted and verify the calculation process of the attained EEXI to ensure that it is technically sound and reasonable and follows regulation 23 of MARPOL Annex VI and the EEXI Calculation Guidelines.

4.3.4 For verification of the attained EEXI after the major conversion, speed trials of the ship may be conducted, as necessary.

## APPENDIX

## SAMPLE OF EEXI TECHNICAL FILE

## 1 Data

## 1.1 General information

Shipowner	XXX Shipping Line
Shipbuilder	XXX Shipbuilding Company
Hull no.	12345
IMO no.	94112XX
Ship type	Bulk carrier

## 1.2 Principal particulars

Length overall	250.0 m
Length between perpendiculars	240.0 m
Breadth, moulded	40.0 m
Depth, moulded	20.0 m
Summer load line draught, moulded	14.0 m
Deadweight at summer load line draught	150,000 tons

## 1.3 Main engine

Manufacturer	XXX Industries		
Туре	6J70A		
Maximum continuous rating (MCR <sub>ME</sub> )	15,000 kW x 80 rpm		
Limited maximum continuous rating with the Engine Power Limitation installed (MCR <sub>ME,lim</sub> )	9,940 kW x 70 rpm		
SFC at 75% of MCR <sub>ME</sub> or 83% of MCR <sub>ME,lim</sub>	166.5 g/kWh		
Number of sets	1		
Fuel type	Diesel Oil		

## 1.4 Auxiliary engine

Manufacturer	XXX Industries
Туре	5J-200
Maximum continuous rating (MCR <sub>AE</sub> )	600 kW x 900 rpm
SFC at 50% MCR <sub>AE</sub>	220.0 g/kWh
Number of sets	3
Fuel type	Diesel Oil

## 1.5 Ship speed

Ship speed $(V_{ref})$ (with the Engine Power	13.20 knots
Limitation installed)	

## 2 Power curve

(Example 1; case of the EEDI ship)

An approved speed-power curve contained in the EEDI Technical File is shown in figure 2.1.

(Example 2; case of the pre-EEDI ship)

An estimated speed-power curve obtained from the tank test and/or numerical calculations, if available, is also shown in figure 2.1.



Figure 2.1: Power curve

(Example 3; case of the pre-EEDI ship with sea trial result calibrated to a different load draught) An estimated speed-power curve under a ballast draught calibrated to the design load draught, obtained from the tank test and/or numerical calculations, if available, is shown in figure 2.2.



Figure 2.2: Power curve

## 3 Overview of propulsion system and electric power supply system

- 3.1 Propulsion system
- 3.1.1 Main engine Refer to paragraph 1.3 of this appendix.
- 3.1.2 Propeller

Туре	Fixed pitch propeller
Diameter	7.0 m
Number of blades	4
Number of sets	1

- 3.2 Electric power supply system
- 3.2.1 Auxiliary engines Refer to paragraph 1.4 of this appendix.
- 3.2.2 Main generators

Manufacturer	XXX Electric		
Rated output	560 kW (700 kVA) x 900 rpm		
Voltage	AC 450 V		
Number of sets	3		



## Figure 3.1: Schematic figure of propulsion and electric power supply system

## 4 Estimation process of speed-power curve

(Example; case of pre-EEDI ship)

Speed-power curve is estimated based on model test results and/or numerical calculations, if available. The flow of the estimation processes is shown below.



#### Figure 4: Flow chart of process for estimating speed-power curve from tank tests

## 5 Description of energy saving equipment

5.1 Energy saving equipment the effects of which are expressed as  $P_{AEeff(i)}$  and/or  $P_{eff(i)}$  in the EEXI calculation formula

N/A

5.2 Other energy saving equipment

(Example)

- 5.2.1 Rudder fins
- 5.2.2 Rudder bulb

..... (Specifications, schematic figures and/or photos, etc. for each piece of equipment or device should be indicated. Alternatively, attachment of a commercial catalogue may be acceptable.)

#### 6 Calculated value of attained EEXI

#### 6.1 Basic data

Type of ship	Capacity DWT	Speed V <sub>ref</sub> (knots)
Bulk carrier	150,000	13.20

#### 6.2 Main engine

MCR <sub>ME</sub> (kW)	MCR <sub>ME,lim</sub> (kW)	Р <sub>ме</sub> (kW)	Type of fuel	С <sub>FME</sub>	SFC <sub>ME</sub> (g/kWh)
15,000	9,940	8,250	Diesel oil	3.206	166.5

#### 6.3 Auxiliary engines

P <sub>AE</sub> (kW)	Type of fuel	CFAE	SFC <sub>AE</sub> (g/kWh)
625	Diesel oil	3.206	220.0

#### 6.4 Ice class

N/A

6.5 Innovative electrical energy-efficient technology

N/A

6.6 Innovative mechanical energy-efficient technology

N/A

6.7 Cubic capacity correction factor

N/A

6.8 Calculated value of attained EEXI

$$\begin{split} EEXI &= \frac{\left(\prod_{j=1}^{M} f_{j}\right)\left(\sum_{i=1}^{nME} P_{ME(i)} \cdot C_{FME(i)} \cdot SFC_{ME(i)}\right) + \left(P_{AE} \cdot C_{FAE} \cdot SFC_{AE}\right)}{f_{i} \cdot f_{c} \cdot f_{l} \cdot Capacity \cdot f_{w} \cdot V_{ref} \cdot f_{m}} \\ &+ \frac{\left\{\left(\prod_{j=1}^{M} f_{j} \cdot \sum_{i=1}^{nPTI} P_{PTI(i)} - \sum_{i=1}^{neff} f_{eff(i)} \cdot P_{AEeff(i)}\right) \cdot C_{FAE} \cdot SFC_{AE}\right\}}{f_{i} \cdot f_{c} \cdot f_{l} \cdot Capacity \cdot f_{w} \cdot V_{ref} \cdot f_{m}} \\ &- \frac{\left(\sum_{i=1}^{neff} f_{eff(i)} \cdot P_{eff(i)} \cdot C_{FME} \cdot SFC_{ME}\right)}{f_{i} \cdot f_{c} \cdot f_{l} \cdot Capacity \cdot f_{w} \cdot V_{ref} \cdot f_{m}} \\ &= \frac{1 \times (8250 \times 3.206 \times 166.5) + (625 \times 3.206 \times 220.0) + 0 - 0}{1 \times 1 \times 1 \times 150000 \times 1 \times 13.20 \times 1} \\ &= 2.45 \left(g - CO_{2}/ton \cdot mile\right) \end{split}$$

## attained EEXI: 2.45 g-CO<sub>2</sub>/ton mile

\*\*\*

#### RESOLUTION MEPC.335(76) (adopted on 17 June 2021)

## 2021 GUIDELINES ON THE SHAFT / ENGINE POWER LIMITATION SYSTEM TO COMPLY WITH THE EEXI REQUIREMENTS AND USE OF A POWER RESERVE

THE MARINE ENVIRONMENT PROTECTION COMMITTEE,

RECALLING Article 38(a) of the Convention on the International Maritime Organization concerning the functions of the Marine Environment Protection Committee conferred upon it by international conventions for the prevention and control of marine pollution from ships,

NOTING that it adopted, by resolution MEPC.328(76), the 2021 revised MARPOL Annex VI, which is expected to enter into force on 1 November 2022 upon its deemed acceptance on 1 May 2022,

NOTING IN PARTICULAR that the 2021 revised MARPOL Annex VI contains amendments concerning mandatory goal-based technical and operational measures to reduce carbon intensity of international shipping,

NOTING FURTHER that ships may be equipped with a Shaft / Engine Power Limitation system in order to comply with regulation 25 (Required EEXI),

RECOGNIZING that the aforementioned amendments to MARPOL Annex VI require relevant guidelines for uniform and effective implementation of the regulations and to provide sufficient lead time for industry to prepare,

HAVING CONSIDERED, at its seventy-sixth session, draft 2021 Guidelines on the shaft / engine power limitation system to comply with the EEXI requirements and use of a power reserve,

1 ADOPTS the 2021 Guidelines on the shaft / engine power limitation system to comply with the EEXI requirements and use of a power reserve, as set out in the annex to the present resolution;

2 INVITES Administrations to take the annexed Guidelines into account when developing and enacting national laws which give force to and implement requirements set forth in regulations 23 and 25 of MARPOL Annex VI;

3 REQUESTS the Parties to MARPOL Annex VI and other Member Governments to bring the annexed Guidelines to the attention of masters, seafarers, shipowners, ship operators and any other interested parties;

4 AGREES to keep the Guidelines under review in light of experience gained with their implementation and in light of the review of EEXI regulations to be completed by the Organization by 1 January 2026 as identified in regulation 25.3 of MARPOL Annex VI;

5 NOTES that the Guidelines may be consolidated with possible future guidelines on the shaft / engine power limitation system under the EEDI framework as appropriate upon consideration by the Committee, taking into account circumstances and technical limitation of existing ships.

# 2021 GUIDELINES ON THE SHAFT / ENGINE POWER LIMITATION SYSTEM TO COMPLY WITH THE EEXI REQUIREMENTS AND USE OF A POWER RESERVE

## Table of contents

- 0 General
- 1 Definitions
- 2 Technical requirements for the SHaPoLi / EPL system
- 3 Use of a power reserve by unlimiting the shaft / engine power limitation
- 4 Onboard Management Manual (OMM) for SHaPoLi / EPL
- 5 Demonstration of compliance of the SHaPoLi / EPL system

## 0 General

The purpose of these Guidelines is to provide technical and operational conditions that the SHaPoLi / EPL system should satisfy in complying with the EEXI requirements and in using a power reserve for existing ships. However, noting that guidelines on the SHaPoLi / EPL system under EEDI framework on new ships are currently considered at the Committee, these guidelines under EEXI and EEDI may be consolidated into one set of guidelines as appropriate upon consideration by the Committee, taking into account circumstances and technical limitation of existing ships.

## 1 Definitions

1.1 *Shaft power* means the mechanical power transmitted by the propeller shaft to the propeller hub. It is the product of the shaft torque and the shaft rotational speed. In case of multiple propeller shafts, the shaft power means the sum of the power transmitted to all propeller shafts.

1.2 *Engine power* means the mechanical power transmitted from the engine to the propeller shaft. In case of multiple engines, the engine power means the sum of the power transmitted from the engines to the propeller shafts.

1.3 *Overridable Shaft Power Limitation (SHaPoLi) system* means a verified and approved system for the limitation of the maximum shaft power by technical means that can only be overridden by the ship's master or the officer in charge of navigational watch (OICNW) for the purpose of securing the safety of a ship or saving life at sea. (See figure 1 for an illustration of engine load diagram.)

1.4 *Overridable Engine Power Limitation (EPL) system* means a verified and approved system for the limitation of the maximum engine power by technical means that can only be overridden by the ship's master or OICNW for the purpose of securing the safety of a ship or saving life at sea. (See figure 1 for an illustration of engine load diagram.)

1.5 *Power reserve* means shaft / engine power above the limited power which cannot be used in normal operation unless in the case when SHaPoLi / EPL is unlimited for the purpose of securing the ship safety.

1.6 *MARPOL* means the International Convention for the Prevention of Pollution from Ships, 1973, as modified by the Protocols of 1978 and 1997 relating thereto, as amended.

1.7 For the purpose of these Guidelines, the definitions in MARPOL Annex VI, as amended, apply.


# 2 Technical requirements for the SHaPoLi / EPL system

### 2.1 Required main systems

The SHaPoLi / EPL system should consist of the following main arrangements:

- .1 SHaPoLi:
  - .1 sensors for measuring the torque and rotational speed delivered to the propeller(s) of the ship. The system includes the amplifier and the analogue to the digital converter;
  - .2 a data recording and processing device for tracking and calculation of the data as given in paragraph 2.2.5.1 of these Guidelines; and
  - .3 a control unit for calculation and limitation of the power transmitted by the shaft to the propeller(s);
- .2 EPL:
  - .1 for the mechanically controlled engine, a sealing device which can physically lock the fuel index by using a mechanical stop screw sealed by wire or an equivalent device with governor limit setting so that the ship's crew cannot release the EPL without permission from the ship's master or OICNW, as shown in figure 2; or
  - .2 for the electronically controlled engine, fuel index limiter which can electronically lock the fuel index or direct limitation of the power in the engine's control system so that the ship's crew cannot release the EPL without permission from the ship's master or OICNW; and
- .3 where technically possible and feasible, the Sha/PoLi/EPL system should be controlled from the ships' bridge and not require attendance in the machinery space by ship's personnel.



Mechanical stop screw sealed by wire

Engine side control console in the governor

#### Figure 2: Sealing of mechanical stop screw

#### 2.2 General system requirements

2.2.1 The SHaPoLi / EPL system should be non-permanent but should require the deliberate action of the ship's master or OICNW to enable the use of unlimited shaft / engine power (power reserve) of the ship. For systems that use a Password/PIN to control access to the power reserve override, attention should be paid to ensure that the necessary Password/PIN is always available when override is required.

2.2.2 For SHaPoLi / EPL system for the electronically controlled engine, the control unit should inform the ship's master or OICNW clearly and conspicuously when the ship's shaft / engine power exceeds the limited shaft / engine power as stated in the Onboard Management Manual (OMM) for SHaPoLi / EPL or in any case of system malfunction.

- 2.2.3 For EPL for the mechanically controlled engine, the sealing device should either:
  - .1 visibly indicate removal of the sealing when the ship's engine power exceeds the limited engine power as stated in the OMM for EPL or in any case of system malfunction; or
  - .2 be equipped with other systems such as an alert-monitoring system which can indicate when the ship's engine power exceeds the limited engine power as stated in the OMM for EPL or in any case of system malfunction and recording the use of unlimited mode, verified by the Administration or the RO.
- 2.2.4 The SHaPoLi / EPL system (or each subsystem) should be tamper-proof.

2.2.5 The SHaPoLi / EPL system for the electronically controlled engine should indicate the following data during operation:

- .1 for SHaPoLi, shaft rotational speed, shaft torque and shaft power (and total shaft power in case of multiple shaft arrangements) to be recorded constantly in unlimiting mode; or
- .2 for EPL, a fuel index sealing system or power limitation system which can indicate and record the use of unlimited mode.

2.2.6 The procedure for SHaPoLi / EPL depends on the propulsion system and should be described in the OMM for SHaPoLi / EPL in accordance with section 4 of these Guidelines.

# 3 Use of a power reserve by un-limiting the shaft / engine power limitation

3.1 The use of a power reserve is only allowed for the purpose of securing the safety of a ship or saving life at sea, consistent with regulation 3.1 of MARPOL Annex VI (e.g. operating in adverse weather and ice-infested waters, participation in search and rescue operations, avoidance of pirates and engine maintenance). Use of a power reserve should not have adverse impact on the propeller, shaft and related systems. It is important that the ship master and OICNW are not restricted from exercising judgement to override the SHaPoLi / EPL when required for safety purposes. The authority for this should be clearly set out in the OMM and/or the Safety Management System manual, as appropriate.

3.2 Any use of a power reserve should be recorded in the record page of the OMM for SHaPoLi / EPL, signed by the master and should be kept on board. The record should include:

- .1 ship type;
- .2 IMO number;
- .3 ship size in DWT and/or GT, as applicable;
- .4 ship's limited shaft / engine power and ship's maximum unlimited shaft / engine power;
- .5 position of the ship and timestamp when the power reserve was used;
- .6 reason for using the power reserve;
- .7 Beaufort number and wave height or ice condition in case of using the power reserve under adverse weather condition;
- .8 supporting evidence (e.g. expected weather condition) in case of using the power reserve for avoidance action;
- .9 records from the SHaPoLi / EPL system for the electronically controlled engine during the power reserve was used; and
- .10 position of the ship and timestamp when the power limit was reactivated or replaced.

3.3 Where an EPL/ShaPoLi override is activated but the power reserve is not subsequently used, this event should be recorded in the bridge and engine-room logbooks. The engine-room logbook should record power used during the period when the override was activated. The EPL/ShaPoLi should be reset as soon as possible, and details of the reset should also be recorded in the bridge and engine-room logbooks.

3.4 In case of having used a power reserve, the ship should without delay notify its Administration or RO responsible for issuing the relevant certificate and the competent authority of the relevant port of destination with the information recorded in accordance with paragraph 3.2. On an annual basis, the Administration should report uses of a power reserve to IMO with the information recorded in accordance with paragraph 3.2.

3.5 Once the risks have been mitigated, the ship should be operated below the certified level of engine power under the SHaPoLi / EPL. The SHaPoLi / EPL system should be reactivated or replaced by the crew immediately after the risks have been prevented and the ship can be safely operated with the limited shaft / engine power. The reactivation or replacement of the SHaPoLi / EPL system should be confirmed (e.g. validation of mechanical sealing) with supporting evidence (e.g. engine power log, photo taken at the occasion of resetting the mechanical sealing) by the Administration or the RO at the earliest opportunity.

3.6 Any defect of the SHaPoLi / EPL system should be reported to the Administration or RO responsible for issuing the relevant certificate in accordance with regulation 5.6 of MARPOL Annex VI.

3.7 The port State control officers should inspect whether the SHaPoLi / EPL system has been properly installed and used in accordance with the IEE Certificate and the OMM as described in section 4 of these Guidelines. If overriding of the SHaPoLi / EPL without proper notification in accordance with paragraph 3.3 of these Guidelines has been detected, the reactivation or replacement of the SHaPoLi / EPL should be immediately conducted in the presence of the Administration or the RO at the port.

# 4 Onboard Management Manual (OMM) for SHaPoLi / EPL

4.1 The SHaPoLi / EPL system should be accompanied by the OMM for SHaPoLi / EPL that should be permanently on board the ship for inspection.

4.2 The OMM for SHaPoLi / EPL should be verified by the Administration or the RO after a survey verifying the ship's attained EEXI, as required by regulation 5.4 of MARPOL Annex VI.

- 4.3 The OMM for SHaPoLi / EPL should, as a minimum, include:
  - .1 SHaPoLi:
    - .1 a technical description of the main system as specified in section 2 of these guidelines as well as relevant auxiliary systems;
    - .2 identification of key components of the system by manufacturer, model/type, serial number and other details as necessary;
    - .3 description of a verification procedure demonstrating that the system is in compliance with the technical description in accordance with items .1 and .2;
    - .4 the maximum shaft power for which the unit is designed;
    - .5 service, maintenance and calibration requirements of sensors according to sensor manufacturer and a description how to monitor the appropriateness of the calibration intervals, if applicable;
    - .6 the SHaPoLi record book for the recording of service, maintenance and calibration of the system;
    - .7 the description how the shaft power can be limited and unlimited and how this is displayed by the control unit as required by paragraph 2.2.5 of these Guidelines;

- .8 the description of how the controller limits the power delivered to the propeller shaft;
- .9 the identification of responsibilities;
- .10 procedures for notification of the use of power reserve and the detections of malfunctions of the system in accordance with paragraphs 3.4 and 3.5 of these Guidelines;
- .11 time required for un-limiting the SHaPoLi; and
- .12 procedures for survey of the SHaPoLi system by the Administration/RO.
- .2 EPL:
  - .1 rated installed power (MCR) or motor output (MPP) and engine speed (N<sub>MCR</sub>);
  - .2 limited installed power (MCR<sub>lim</sub>) or motor output (MPP<sub>lim</sub>) and engine speed ( $N_{MCR,lim}$ );
  - .3 technical description of the EPL system;
  - .4 method for sealing the EPL (mechanically controlled engine);
  - .5 method for locking and monitoring the EPL (electronically controlled engine);
  - .6 procedures and methods for releasing the EPL;
  - .7 time required for unlimiting the EPL;
  - .8 procedures for survey of the EPL system by the Administration/RO;
  - .9 procedure for the report on release of the EPL; and
  - .10 administrator of the EPL system.

#### 5 Demonstration of compliance of the SHaPoLi / EPL system

5.1 The demonstration of compliance of the SHaPoLi / EPL system should be verified by an appropriate survey in accordance with regulation 5.4 of MARPOL Annex VI for the verification of the ship's EEXI according to regulation 23. The survey should include the verification and validation of the system by addressing the following items:

- .1 the verification of compliance of the system with the OMM for SHaPoLi / EPL;
- .2 the verification of compliance of the system with the specifications set out in section 2 of these Guidelines; and
- .3 the verification of the OMM for SHaPoLi / EPL that the OMM for SHaPoLi / EPL is in compliance with the specifications set out in section 4 of these Guidelines.

5.2 In cases where the SHaPoLi / EPL system is applied and no changes are made to  $NO_X$  critical settings and/or components<sup>\*</sup> outside what is allowed by the engine technical file as defined in the 2008  $NO_X$  Technical Code (NTC 2008), engine re-certification is not needed.

5.3 In cases where the SHaPoLi / EPL system is applied and the NO<sub>X</sub> critical settings and/or components are altered beyond what is allowed by the engine technical file as defined in NTC 2008, the engine needs to be re-certified. In such a case, for an EEDI-certified ship where the SHaPoLi / EPL system is applied at a power below that required by regulation 24.5 of MARPOL Annex VI (minimum power requirement), the certified engine power should be at the power satisfying that requirement.

NO<sub>X</sub> critical parameters and components are listed in NO<sub>X</sub> Technical File under the section "Components, setting and operating values of the engine which may influence its NO<sub>X</sub> emission".

#### RESOLUTION MEPC.336(76) (adopted on 17 June 2021)

# 2021 GUIDELINES ON OPERATIONAL CARBON INTENSITY INDICATORS AND THE CALCULATION METHODS (CII GUIDELINES, G1)

# THE MARINE ENVIRONMENT PROTECTION COMMITTEE,

RECALLING Article 38(a) of the Convention on the International Maritime Organization concerning the functions of the Marine Environment Protection Committee conferred upon it by international conventions for the prevention and control of marine pollution from ships,

NOTING that it adopted, by resolution MEPC.328(76), the 2021 revised MARPOL Annex VI, which is expected to enter into force on 1 November 2022 upon its deemed acceptance on 1 May 2022,

NOTING IN PARTICULAR that the 2021 revised MARPOL Annex VI contains amendments concerning mandatory goal-based technical and operational measures to reduce carbon intensity of international shipping,

NOTING FURTHER that regulation 28.1 of MARPOL Annex VI requires ships to which this regulation apply to calculate the attained annual operational CII taking into account the guidelines developed by the Organization,

RECOGNIZING that the aforementioned amendments to MARPOL Annex VI require relevant guidelines for uniform and effective implementation of the regulations and to provide sufficient lead time for industry to prepare,

HAVING CONSIDERED, at its seventy-sixth session, draft 2021 Guidelines on operational carbon intensity indicators and the calculation methods (CII Guidelines, G1),

1 ADOPTS the 2021 Guidelines on operational carbon intensity indicators and the calculation methods (CII Guidelines, G1), as set out in the annex to the present resolution;

2 INVITES Administrations to take the annexed Guidelines into account when developing and enacting national laws which give force to and implement requirements set forth in regulation 28.1 of MARPOL Annex VI;

3 REQUESTS the Parties to MARPOL Annex VI and other Member Governments to bring the annexed Guidelines to the attention of masters, seafarers, shipowners, ship operators and any other interested parties;

4 AGREES to consider substantiated proposals for CII correction factors for certain ship types, operational profiles and/or voyages with a view to enhancing, as appropriate, the annexed Guidelines before entry into force of the aforementioned amendments to MARPOL Annex VI;

5 AGREES to keep the Guidelines under review in light of experience gained with their implementation and in light of the review of CII regulations to be completed by the Organization by 1 January 2026 as identified in regulation 28.11 of MARPOL Annex VI.

# 2021 GUIDELINES ON OPERATIONAL CARBON INTENSITY INDICATORS AND THE CALCULATION METHODS (CII GUIDELINES, G1)

#### 1 Introduction

1.1 In the *Initial IMO Strategy on Reduction of GHG Emissions from Ships* (Resolution MEPC.304(72)), the level of ambition on carbon intensity of international shipping is quantified by the  $CO_2$  emissions per transport work, as an average across international shipping.

1.2 These Guidelines address the calculation methods and the applicability of the operational carbon intensity indicator (CII) for individual ships to which chapter 4 of MARPOL Annex VI, as amended, applies.

#### 2 Definitions

2.1 *MARPOL* means the International Convention for the Prevention of Pollution from Ships, 1973, as modified by the Protocols of 1978 and 1997 relating thereto, as amended.

2.2 *IMO DCS* means the data collection system for fuel oil consumption of ships referred to in regulation 27 and related provisions of MARPOL Annex VI.

2.3 For the purpose of these Guidelines, the definitions in MARPOL Annex VI, as amended, apply.

2.4 The metrics indicating the average CO<sub>2</sub> emissions per transport work of a ship are generally referred to as operational carbon intensity indicator (CII) in these Guidelines.

- .1 A specific CII calculated based on the actual or estimated mass or volume of the shipment carried on board a ship is generally referred to as *demand-based CII*; and
- .2 A specific CII, in which calculation the capacity of a ship is taken as proxy of the actual mass or volume of the shipment carried on board, is generally referred to as *supply-based CII*.

2.5 The supply-based CII which uses DWT as the capacity is referred to as *AER*, and the supply-based CII which uses GT as the capacity is referred to as *cgDIST*.

#### 3 Application

3.1 For all ships to which regulation 28 of MARPOL Annex VI applies, the operational carbon intensity indicators defined in section 4 should be applied.

3.2 The operational carbon intensity indicators defined in section 5 are encouraged to be additionally used by ships, where applicable, for trial purposes.

# 4 Operational carbon intensity indicator (CII) of individual ships for use in implementing regulation 28 of MARPOL Annex VI

In its most simple form, the attained annual operational CII of individual ships is calculated as the ratio of the total mass of  $CO_2$  (*M*) emitted to the total transport work (*W*) undertaken in a given calendar year, as follows:

attained 
$$CII_{ship} = M / W$$
 (1)

# 4.1 Mass of CO<sub>2</sub> emissions (M)

The total mass of  $CO_2$  is the sum of  $CO_2$  emissions (in grams) from all the fuel oil consumed on board a ship in a given calendar year, as follows:

$$M = FC_j \times C_{F_j} \tag{2}$$

where:

- J is the fuel oil type;
- $FC_j$  is the total mass (in grams) of consumed fuel oil of type  $\dot{J}$  in the calendar year, as reported under IMO DCS; and
- $C_{F_i}$  represents the fuel oil mass to CO<sub>2</sub> mass conversion factor for fuel oil type

 $\dot{J}$ , in line with those specified in the 2018 Guidelines on the method of calculation of the attained Energy Efficiency Design Index (EEDI) for new ships (resolution MEPC.308(73)), as may be further amended. In case the type of the fuel oil is not covered by the guidelines, the conversion factor should be obtained from the fuel oil supplier supported by documentary evidence.

#### 4.2 Transport work (W)

In the absence of the data on actual transport work, the supply-based transport work ( $W_s$ ) can be taken as a proxy, which is defined as the product of a ship's capacity and the distance travelled in a given calendar year, as follows:

$$W_s = C \times D_t \tag{3}$$

where:

- C represents the ship's capacity:
  - For bulk carriers, tankers, container ships, gas carriers, LNG carriers, ro-ro cargo ships, general cargo ships, refrigerated cargo carrier and combination carriers, deadweight tonnage (DWT)<sup>1</sup> should be used as Capacity;
  - For cruise passenger ships, ro-ro cargo ships (vehicle carriers) and ro-ro passenger ships, gross tonnage (GT)<sup>2</sup> should be used as Capacity; and
- $D_t$  represents the total distance travelled (in nautical miles), as reported under IMO DCS.

<sup>2</sup> Gross tonnage (GT) should be calculated in accordance with the International Convention on Tonnage Measurement of Ships, 1969.

<sup>&</sup>lt;sup>1</sup> Deadweight tonnage (DWT) means the difference in tonnes between the displacement of a ship in water of relative density of 1,025 kg/m3 at the summer load draught and the lightweight of the ship. The summer load draught should be taken as the maximum summer draught as certified in the stability booklet approved by the Administration or any organization recognized by it.

# 5 Operational carbon intensity indicator (CII) of individual ships for trial purpose

The following metrics are encouraged to be used for trial purposes, where applicable:

.1 Energy Efficiency Performance Indicator (EEPI)

$$EEPI = \frac{M}{C \times D_l}$$

.2 cbDIST

$$cbDIST = \frac{M}{ALB \times D_t}$$

.3 clDIST

$$clDIST = \frac{M}{Lanemeter \times D_t}$$

.4 EEOI, as defined in MEPC.1/Circ.684 on *Guidelines for voluntary use of the ship energy efficiency operational indicator (EEOI).* 

In the formulas above:

- the mass of CO<sub>2</sub> (*M*), the ship's capacity (*C*) and the total distance travelled (*D*<sub>t</sub>) are identical with those used to calculate the attained CII of individual ships, as specified in section 4.1 and 4.2;
- $D_l$  means the laden distance travelled (in nautical miles) when the ship is loaded;
- *ALB* means the number of available lower berths of a cruise passenger ship; and
- Lanemeter means the length (in metres) of the lanes of a ro-ro ship.

#### RESOLUTION MEPC.337(76) (adopted on 17 June 2021)

#### 2021 GUIDELINES ON THE REFERENCE LINES FOR USE WITH OPERATIONAL CARBON INTENSITY INDICATORS (CII REFERENCE LINES GUIDELINES, G2)

THE MARINE ENVIRONMENT PROTECTION COMMITTEE,

RECALLING Article 38(a) of the Convention on the International Maritime Organization concerning the functions of the Marine Environment Protection Committee conferred upon it by international conventions for the prevention and control of marine pollution from ships,

NOTING that it adopted, by resolution MEPC.328(76), the 2021 revised MARPOL Annex VI, which is expected to enter into force on 1 November 2022 upon its deemed acceptance on 1 May 2022,

NOTING IN PARTICULAR that the 2021 revised MARPOL Annex VI contains amendments concerning mandatory goal-based technical and operational measures to reduce carbon intensity of international shipping,

NOTING FURTHER that regulation 28.4 of MARPOL Annex VI requires reference lines to be established for each ship type to which regulation 28 is applicable,

HAVING CONSIDERED, at its seventy-sixth session, draft 2021 Guidelines on the reference lines for use with operational carbon intensity indicators (CII reference lines guidelines, G2),

1 ADOPTS the 2021 Guidelines on the reference lines for use with operational carbon intensity indicators (CII reference lines guidelines, G2), as set out in the annex to the present resolution;

2 INVITES Administrations to take the annexed Guidelines into account when developing and enacting national laws which give force to and implement requirements set forth in regulation 28.4 of MARPOL Annex VI;

3 REQUESTS the Parties to MARPOL Annex VI and other Member Governments to bring the annexed Guidelines to the attention of masters, seafarers, shipowners, ship operators and any other interested parties;

4 AGREES to keep the Guidelines under review in light of experience gained with their implementation and in light of the review of CII regulations to be completed by the Organization by 1 January 2026 as identified in regulation 28.11 of MARPOL Annex VI.

# 2021 GUIDELINES ON THE REFERENCE LINES FOR USE WITH OPERATIONAL CARBON INTENSITY INDICATORS (CII REFERENCE LINES GUIDELINES, G2)

#### 1 Introduction

1.1 These Guidelines provide the methods to calculate the reference lines for use with operational carbon intensity indicators, and the ship type specific carbon intensity reference lines as referred to in regulation 28 of MARPOL Annex VI.

1.2 One reference line is developed for each ship type to which regulation 28 of MARPOL Annex VI applies, based on the specific indicators stipulated in *2021 Guidelines on operational carbon intensity indicators and the calculation methods* (G1) developed by the Organization, ensuring that only data from comparable ships are included in the calculation of each reference line.

# 2 Definition

2.1 *MARPOL* means the International Convention for the Prevention of Pollution from Ships, 1973, as modified by the Protocols of 1978 and 1997 relating thereto, as amended.

2.2 *IMO DCS* means the data collection system for fuel oil consumption of ships referred to in regulation 27 and related provisions of MARPOL Annex VI.

2.3 For the purpose of these Guidelines, the definitions in MARPOL Annex VI, as amended, apply.

2.4 An operational carbon intensity indicator (CII) reference line is defined as a curve representing the median attained operational carbon intensity performance, as a function of Capacity, of a defined group of ships in year of 2019.

#### 3 Method to develop the CII reference lines

3.1 Given the limited data available for the year of 2008, the operational carbon intensity performance of ship types in year 2019 is taken as the reference.

3.2 For a defined group of ships, the reference line is formulated as follows:

$$CII_{ref} = aCapacity^{-c} \tag{1}$$

where  $_{CII_{ref}}$  is the reference value of year 2019,  $_{Capacity}$  is identical with the one defined in the specific carbon intensity indicator (CII) for a ship type, as shown in Table. 1; *a* and *c* are parameters estimated through median regression fits, taking the attained CII and the Capacity of individual ships collected through IMO DCS in year 2019 as the sample.

# 4 Ship type specific operational carbon intensity reference lines

The parameters for determining the ship type specific reference lines, for use in Eq.(1), are specified as follows:

		Ship type	Capacity a		
Bulk carrier	279,00	) DWT and above	279,000	4745	0.622
	less tha	an 279,000 DWT	DWT	4745	0.622
Gas carrier	65,000	65,000 and above		14405E7	2.071
	less tha	less than 65,000 DWT		8104	0.639
Tanker			DWT	5247	0.610
Container shi	р		DWT 1984 0.4		
		20,000 DWT and above	DWT	31948	0.792
General carg	Jeneral cargo ship less than 20,000 DWT DWT		588	0.3885	
Refrigerated cargo carrier D			DWT	4600	0.557
Combination	carrier		DWT 40853 0.81		
LNG carrier	100,000	100,000 DWT and above		9.827	0.000
	65,000 DWT and above, but less than 100,000 DWT		DWT	14479E10	2.673
	less than 65,000 DWT		65,000	14479E10	2.673
Ro-ro cargo s	o-ro cargo ship (vehicle carrier) GT 5739				0.631
Ro-ro cargo ship			DWT	10952	0.637
Ro-ro passenger ship			GT	7540	0.587
Cruise passenger ship			GT	930	0.383

Table 1: Parameters for determining the 2019 ship type specific reference lines

#### RESOLUTION MEPC.338(76) (adopted on 17 June 2021)

#### 2021 GUIDELINES ON THE OPERATIONAL CARBON INTENSITY REDUCTION FACTORS RELATIVE TO REFERENCE LINES (CII REDUCTION FACTORS GUIDELINES, G3)

THE MARINE ENVIRONMENT PROTECTION COMMITTEE,

RECALLING Article 38(a) of the Convention on the International Maritime Organization concerning the functions of the Marine Environment Protection Committee conferred upon it by international conventions for the prevention and control of marine pollution from ships,

NOTING that it adopted, by resolution MEPC.328(76), the 2021 revised MARPOL Annex VI, which is expected to enter into force on 1 November 2022 upon its deemed acceptance on 1 May 2022,

NOTING IN PARTICULAR that the 2021 revised MARPOL Annex VI contains amendments concerning mandatory goal-based technical and operational measures to reduce carbon intensity of international shipping,

NOTING FURTHER that regulation 28.4 of MARPOL Annex VI requires reduction factors to be established for each ship type to which regulation 28 is applicable,

HAVING CONSIDERED, at its seventy-sixth session, draft 2021 Guidelines on the operational carbon intensity reduction factors relative to reference lines (CII reduction factors guidelines, G3),

1 ADOPTS the 2021 Guidelines on the operational carbon intensity reduction factors relative to reference lines (CII reduction factors guidelines, G3), as set out in the annex to the present resolution;

2 INVITES Administrations to take the annexed Guidelines into account when developing and enacting national laws which give force to and implement requirements set forth in regulation 28.4 of MARPOL Annex VI;

3 REQUESTS the Parties to MARPOL Annex VI and other Member Governments to bring the annexed Guidelines to the attention of masters, seafarers, shipowners, ship operators and any other interested parties;

4 AGREES to keep the Guidelines under review in light of experience gained with their implementation and in light of the review of CII regulations to be completed by the Organization by 1 January 2026 as identified in regulation 28.11 of MARPOL Annex VI, and that annual reduction rates for the period 2027-2030 will be further strengthened and developed taking into account that review.

# 2021 GUIDELINES ON THE OPERATIONAL CARBON INTENSITY REDUCTION FACTORS RELATIVE TO REFERENCE LINES (CII REDUCTION FACTORS GUIDELINES, G3)

#### 1 Introduction

1.1 These Guidelines provide the methods to determine the annual operational carbon intensity reduction factors and their concrete values from year 2023 to 2030, as referred to in regulation 28 of MARPOL Annex VI.

1.2 The annual operational carbon intensity reduction factors apply to each ship type to which regulation 28 of MARPOL Annex VI applies, in a transparent and robust manner, based on the specific carbon intensity indicators stipulated in the *2021 Guidelines on operational carbon intensity indicators and the calculation methods (G1)* (resolution MEPC.336(76)) and the reference lines developed in the *2021 Guidelines on the reference lines for use with operational carbon intensity indicators (G2)*(resolution MEPC.337(76)).

1.3 The reduction factors have been set at the levels to ensure that, in combination with other relevant requirements of MARPOL Annex VI, the reduction in  $CO_2$  emissions per transport work by at least 40% by 2030, compared to 2008, can be achieved as an average across international shipping.

1.4 Section 5 of these Guidelines provides background information on rational ranges of reduction factors of ship types in year 2030 using demand-based measurement and supply-based measurement.

1.5 The Organization should continue to monitor development in annual carbon intensity improvement using both demand-based measurement and supply-based measurement in parallel to the annual analysis of the fuel consumption data reported to the IMO DCS.

#### 2 Definitions

2.1 *MARPOL* means the International Convention for the Prevention of Pollution from Ships, 1973, as modified by the Protocols of 1978 and 1997 relating thereto, as amended.

2.2 *IMO DCS* means the data collection system for fuel oil consumption of ships referred to in regulation 27 and related provisions of MARPOL Annex VI.

2.3 For the purpose of these Guidelines, the definitions in MARPOL Annex VI, as amended, apply.

2.4 The annual operational carbon intensity reduction factor, generally denoted as "Z" in regulation 28 of MARPOL Annex VI, is a positive value, stipulating the percentage points of the required annual operational carbon intensity indicator of a ship for a given year lower than the reference value.

# 3 Method to determine the annual reduction factor of ship types

# 3.1 Operational carbon intensity of international shipping

Given significant heterogeneity across ship types, the attained annual operational CII of international shipping as a whole is calculated as the ratio of the aggregated mass (in grams) of CO<sub>2</sub> (aggregated M) emitted to the aggregated mass (in tonne·nmiles) of transport work (aggregated W) undertaken by all individual ships of representative ship types in a given calendar year, as follows:

attained 
$$CII_{shipping} = aggregated M / aggregated W$$
 (1)

In the absence of the data on actual annual transport work of individual ships, the aggregated transport work obtained from other reliable sources, such as UNCTAD, can be taken as approximation. The representative ship types refer to bulk carriers, gas carriers, tankers, container ships, general cargo ships, refrigerated cargo carrier and LNG carriers, as per the *Fourth IMO GHG Study 2020*.

# 3.2 The achieved carbon intensity reduction in international shipping

For a given year y, the achieved carbon intensity reduction in international shipping relative to the reference year  $y_{ref}$ , denoted as  $R_{shipping,y}$ , can be calculated as follows:

$$R_{shipping,y} = 100\% \times (attained CII_{shipping,y} - attained CII_{shipping,y_{ref}}) / attained CII_{shipping,y_{ref}}$$
(2)

where the *attained*  $CII_{shipping,y}$  and *attained*  $CII_{shipping,y_{ref}}$  represents the attained annual operational carbon intensity of international shipping in year y and in the reference year  $y_{ref}$ , as defined in Eq.(1).

The achieved carbon intensity reduction in international shipping can be alternatively calculated on the carbon intensity performance of ship types. Since CII metrics for different ship types may not be identical, the weighted average of the carbon intensity reduction achieved by ship types can be applied, as follows:

$$R_{shipping,y} = \sum_{type} f_{type,y} R_{type,y}$$
(3)

In Eq(3),

• *type* represents the ship type;

 $f_{type,y}$  is the weight, which is equal to the proportion of CO<sub>2</sub> emitted by the ship type to the total CO<sub>2</sub> emissions of international shipping in year y; and  $R_{type,y}$  represents the carbon intensity reduction achieved by the ship type in year y, calculated as  $R_{type,y}=100\%\times(attained CII_{type,y}-attained CII_{type,y_{ref}})/attained CII_{type,y_{ref}}$ , where the  $attained CII_{type,y}$  and  $attained CII_{type,ref}$  represents the attained annual operational carbon intensity of the ship type in year y and in the reference year  $y_{type}$ , as defined in Eq.(4), as follows:

attained 
$$CII_{type} = \sum_{ship} M_{ship,t} / \sum_{ship} W_{ship,t}$$
 (4)

where:

 $M_{ship,t}$  and  $W_{ship,t}$  represents the total mass of CO<sub>2</sub> emitted from and the total transport work undertaken by a ship of this type in a given calendar year, as stipulated in the *Guidelines on operational carbon intensity indicators and the calculation methods (G1)*.

# 4 The reduction factors for the required annual operational CII of ship types

4.1 In accordance with regulation 28 of MARPOL Annex VI, the required annual operational CII for a ship is calculated as follows:

Required annual operational  $CII = (1 - Z / 100) \times CII_{R}$ 

where  $_{CII_R}$  is the reference value in year 2019 as defined in the *Guidelines on the reference lines for use with operational carbon intensity indicators (G2)*, *Z* is a general reference to the reduction factors for the required annual operational CII of ship types from year 2023 to 2030, as specified in table 1.

Year	Reduction factor relative to 2019
2023	5%*
2024	7%
2025	9%
2026	11%
2027	- **
2028	- **
2029	- **
2030	- **

# Table 1: Reduction factor (Z%) for the CII relative to the 2019 reference line

Note:

- \* Z factors of 1%, 2% and 3% are set for the years of 2020 to 2022, similar as business as usual until entry into force of the measure.
- \*\* Z factors for the years of 2027 to 2030 to be further strengthened and developed taking into account the review of the short-term measure.

# 5 Background information on rational ranges of reduction factors of ship types in year 2030

5.1 In the *Initial IMO Strategy on Reduction of GHG Emissions from Ships* (Resolution MEPC.304(72)), the levels of ambition on carbon intensity of international shipping have been set taking year 2008 as reference. The carbon intensity of international shipping in year 2008, as well as the improvement through 2012 to 2018, has been estimated in the *Fourth IMO GHG Study 2020*. However, since the scope and data collection methods applied in the *Fourth IMO GHG Study 2020* were inconsistent with those under IMO DCS, the results derived from the two sources cannot be compared directly.

5.2 To ensure the comparability of the attained carbon intensity of international shipping through year 2023 to 2030 with the reference line, the following methods are applied to calculate the equivalent carbon intensity target in year 2030 ( $_{eR_{shipping,2030}}$ ), taking year 2019 as reference, i.e. how much additional improvement is needed by 2030 from the 2019 performance level.

5.3 The achieved carbon intensity reduction of international shipping in year 2019 relative to year 2008 ( $_{R_{shipping,2019}}$ ) can be estimated as the sum of the achieved carbon intensity reduction of international shipping in year 2018 relative to year 2008 ( $_{R_{shipping,2018}}$ ) as given by the *Fourth IMO GHG Study 2020* and the estimated average annual improvement during 2012 and 2018 ( $_{\overline{r}_{shipping}}$ ), as follows:

$$R_{shipping,2019} = R_{shipping,2018} + \overline{r}_{shipping}$$
 (5)

5.4 The following provides the calculations using demand-based measurement and supply-based measurement.

# 5.4.1 Demand-based measurement of 2030 target

As estimated by the *Fourth IMO GHG Study 2020*, the attained CII of international shipping (on aggregated demand-based metric) has reduced by **31.8%** ( $_{R_{shipping,2018}=31.8\%}$ ) compared to 2008, with an estimated average annual improvement at **1.5** percentage points ( $_{\overline{r}_{shipping}=1.5\%}$ ). In accordance with Eq.(5), the carbon intensity reduction achieved in year 2019 is estimated as **33.3%** ( $_{R_{shipping,2019}=33.3\%}$ ).

# 5.4.2 Supply-based measurement of 2030 target

As estimated by the *Fourth IMO GHG Study 2020*, the attained CII of international shipping (on aggregated supply-based metric) has reduced by **22.0%** ( $_{R_{shipping,2018}}=22.0\%$ ) compared to 2008, with an estimated average annual improvement at **1.6** percentage points ( $_{\overline{r}_{shipping}}=1.6\%$ ). In accordance with Eq.(5), the carbon intensity reduction achieved in year 2019 relative to 2008 is estimated as **23.6%** ( $_{R_{shipping,2019}}=23.6\%$ ).

5.5 Given the achieved carbon intensity reduction of international shipping in year 2019 relative to year 2008, the carbon intensity reduction target of international shipping in year 2030 can be converted to the equivalent target ( $_{eR_{shipping},2030}$ ) relative to year 2019, as follows:

$$eR_{shipping,2030} = \frac{40\% - R_{shipping,2019}}{1 - R_{shipping,2019}}$$
(6)

#### 5.5.1 Demand-based measurement of 2030 target

In accordance with Eq.(6), the equivalent reduction factor of international shipping in year 2030 relative to year 2019 ( $_{eR_{shipping,2030}}$ ) would be at least **10.0%** measured in aggregated demand-based CII metric, i.e. at least additional **10.0%** improvement from the 2019 level is needed by 2030.

# 5.5.2 Supply-based measurement of 2030 target

In accordance with Eq.(6), the equivalent reduction factor of international shipping in 2030 relative to year 2019 ( $_{eR_{shipping,2030}}$ ) would be at least **21.5%**, measured in aggregated supply-based CII metric, i.e. at least additional **21.5%** improvement from the 2019 level is needed by 2030.

#### RESOLUTION MEPC.339(76) (adopted on 17 June 2021)

#### 2021 GUIDELINES ON THE OPERATIONAL CARBON INTENSITY RATING OF SHIPS (CII RATING GUIDELINES, G4)

THE MARINE ENVIRONMENT PROTECTION COMMITTEE,

RECALLING Article 38(a) of the Convention on the International Maritime Organization concerning the functions of the Marine Environment Protection Committee conferred upon it by international conventions for the prevention and control of marine pollution from ships,

NOTING that it adopted, by resolution MEPC.328(76), the 2021 revised MARPOL Annex VI, which is expected to enter into force on 1 November 2022 upon its deemed acceptance on 1 May 2022,

NOTING IN PARTICULAR that the 2021 revised MARPOL Annex VI contains amendments concerning mandatory goal-based technical and operational measures to reduce carbon intensity of international shipping,

NOTING FURTHER that regulation 28.6 of MARPOL Annex VI requires ships to which this regulation apply to determine operational carbon intensity rating taking into account guidelines developed by the Organization,

RECOGNIZING that the aforementioned amendments to MARPOL Annex VI require relevant guidelines for uniform and effective implementation of the regulations,

HAVING CONSIDERED, at its seventy-sixth session, draft 2021 Guidelines on the operational carbon intensity rating of ships (CII rating guidelines, G4),

1 ADOPTS the 2021 Guidelines on the operational carbon intensity rating of ships (CII rating guidelines, G4), as set out in the annex to the present resolution;

2 INVITES Administrations to take the annexed Guidelines into account when developing and enacting national laws which give force to and implement requirements set forth in regulation 28.6 of MARPOL Annex VI;

3 REQUESTS the Parties to MARPOL Annex VI and other Member Governments to bring the annexed Guidelines to the attention of masters, seafarers, shipowners, ship operators and any other interested parties;

4 AGREES to keep the Guidelines under review in light of experience gained with their implementation, of additional data collected and analysed, and in light of the review of CII regulations to be completed by the Organization by 1 January 2026 as identified in regulation 28.11 of MARPOL Annex VI.

#### 2021 GUIDELINES ON THE OPERATIONAL CARBON INTENSITY RATING OF SHIPS (CII RATING GUIDELINES, G4)

#### 1 Introduction

1.1 These Guidelines provide the methods to assign operational energy efficiency performance ratings to ships, as referred to in regulation 28 of MARPOL Annex VI. On this basis, the boundaries for determining a ship's annual operational carbon intensity performance from year 2023 to 2030 are also provided.

#### 2 Definitions

2.1 *MARPOL* means the International Convention for the Prevention of Pollution from Ships, 1973, as modified by the Protocols of 1978 and 1997 relating thereto, as amended.

2.2 *IMO DCS* means the data collection system for fuel oil consumption of ships referred to in regulation 28 and related provisions of MARPOL Annex VI.

2.3 For the purpose of these Guidelines, the definitions in MARPOL Annex VI, as amended, apply.

2.4 *Operational carbon intensity rating* means to assign a ranking label from among the five grades (A, B, C, D and E) to the ship based on the attained annual operational carbon intensity indicator, indicating a major superior, minor superior, moderate, minor inferior, or inferior performance level.

#### 3 Framework of the operational energy efficiency performance rating

3.1 An operational energy efficiency performance rating should be annually assigned to each ship to which regulation 28 of MARPOL Annex VI applies, in a transparent and robust manner, based on the deviation of the attained annual operational carbon intensity indicator (CII) of a ship from the required value.

3.2 To facilitate the rating assignment, for each year from 2023 to 2030, four boundaries are defined for the five-grade rating mechanism, namely superior boundary, lower boundary, upper boundary, and inferior boundary. Thus, a rating can be assigned through comparing the attained annual operational CII of a ship with the boundary values.

3.3 The boundaries are set based on the distribution of CIIs of individual ships in year 2019. The appropriate rating boundaries are expected to generate the following results: the middle 30% of individual ships across the fleet segment, in terms of the attained annual operational CIIs, are to be assigned rating C, while the upper 20% and further upper 15% of individuals are to be assigned rating D and E respectively, the lower 20% and further lower 15% of the individuals are to be assigned rating B and A respectively, as illustrated in figure 1.



Figure 1: Operational energy efficiency performance rating scale

3.4 Given the incremental operational carbon intensity reduction factors over time, the boundaries for defining performance ratings should be synchronized accordingly, although the relative distance between the boundaries should not change. The rating of a ship would be determined by the attained CII and the predetermined rating boundaries, rather than the attained CII of other ships. Note that the distribution of ship individual ratings in a specific year may not be always identical with the scenario in 2019, where for example 20% may achieve A, 30% may achieve B, 40% may achieve C, 8% may achieve D and 2% may achieve E in a given year.

# 4 Method to determine the rating boundaries

4.1 The boundaries can be determined by the required annual operational CII in conjunction with the vectors, indicating the direction and distance they deviate from the required value (denoted as dd vectors for easy reference), as illustrated in figure 2.



Figure 2: *dd* vectors and rating bands

4.2 Statistically, the *dd* vectors depend on the distribution of the attained annual operational CII of ships of the type concerned, which can be estimated through a quantile regression, taking data collected through DCS in year 2019 as the sample.

4.3 The quantile regression model for a specific ship type can be developed as follows:

$$\ln(\text{attained CII}) = \delta^{(p)} - c \ln(\text{Capacity}) + \varepsilon^{(p)}, \quad p = \{0.15, 0.35, 0.50, 0.65, 0.85\}$$
(5)

where *Capacity* is identical with the one used in the operation carbon intensity indicator as specified in the Guidelines on operational carbon intensity indicators and the calculation

methods (G1); p is the typical quantile, meaning the proportion of observations with a lower value is  $p^{\text{(}p)}$ ;  $\delta^{(p)}$  is the constant term, and  $\varepsilon^{(p)}$  is the error term.



4.4 The quantile regression lines in logarithm form are illustrated in Fig.3.

Figure 3: Quantile regression lines in logarithm form

4.5 Then, the *dd* vectors can be calculated based on the estimates of the intercept ( $\hat{\delta}^{(p)}$ ), in accordance with Eq.(2), as follows:

$$\begin{aligned} d_1 &= \hat{\delta}^{(0.15)} - \hat{\delta}^{(0.50)} \\ d_2 &= \hat{\delta}^{(0.35)} - \hat{\delta}^{(0.50)} \\ d_3 &= \hat{\delta}^{(0.65)} - \hat{\delta}^{(0.50)} \\ d_4 &= \hat{\delta}^{(0.85)} - \hat{\delta}^{(0.50)} \end{aligned}$$
(6)

4.6 Through an exponential transformation of each dd vector, the four boundaries fitted in the original data form can be derived based on the required annual operational carbon intensity indicator (*required CII*), as follows:

superior boundary = 
$$\exp(d_1) \cdot required CII$$
  
lower boundary =  $\exp(d_2) \cdot required CII$   
upper boundary =  $\exp(d_3) \cdot required CII$   
inferior boundary =  $\exp(d_4) \cdot required CII$  (7)

# Rating boundaries of ship types

The estimated dd vectors after exponential transformation for determining the rating boundaries of ship types are as follows:

Ship type		Capacity in CII calculation	<i>dd</i> vectors (after exponential transformation)			
			exp(d1)	exp(d2)	exp(d3)	exp(d4)
Bulk carrier		DWT	0.86	0.94	1.06	1.18
	65,000 DWT and above	DWT	0.81	0.91	1.12	1.44
Gas carrier	less than 65,000 DWT	DWT	0.85	5 0.95 1.06	1.06	1.25
Tanker	DWT 0.82 0.93 1.08		1.08	1.28		
Container ship		DWT	0.83	0.94	1.07	1.19
General cargo ship		DWT	0.83	0.94	1.06	1.19
Refrigerated cargo c	Refrigerated cargo carrier DWT 0.78 0.		0.91	1.07	1.20	
Combination carrier		DWT	0.87 0.96 1.06 1.1		1.14	
INC corrier	100,000 DWT and above		0.89	0.94         1.06           0.91         1.12           0.95         1.06           0.93         1.08           0.94         1.07           0.94         1.07           0.94         1.07           0.94         1.07           0.94         1.06           0.91         1.07           0.96         1.06           0.91         1.07           0.96         1.06           0.98         1.06           0.92         1.10           0.94         1.06           0.92         1.11           0.90         1.12           0.95         1.06	1.06	1.13
LING Carrier	less than 100,000 DWT		0.78		1.37	
Ro-ro cargo ship (ve	hicle carrier)	GT	0.86 0.94 1.06		1.16	
Ro-ro cargo ship		DWT	0.66	0.90	1.11	1.37
Ro-ro passenger ship		GT	0.72	0.90	1.12	1.41
Cruise passenger ship		GT	0.87	0.95	1.06	1.16

Table 1: dd	vectors for	determining	the rating	j boundaries	of ship types
-------------	-------------	-------------	------------	--------------	---------------

By comparing the attained annual operational CII of a specific ship with the four boundaries, a rating can then be assigned. For example, given the required CII of a bulk carrier in a specific year as 10 gCO<sub>2</sub>/(dwt.nmile), then the superior boundary, lower boundary, upper boundary, and inferior boundary is 8.6, 9.4, 10.6 and 11.8 gCO<sub>2</sub>/(dwt.nmile). If the attained CII is 9 gCO<sub>2</sub>/(dwt.nmile), the ship would be rated as "B".